AN ENERGY LANDSCAPING APPROACH TO THE PROTEIN FOLDING PROBLEM

A Thesis Presented to The Academic Faculty

by

Temsiri Sapsaman

In Partial Fulfillment of the Requirements for the Degree Doctor of Philosophy in the The George W. Woodruff School of Mechanical Engineering

> Georgia Institute of Technology December 2009

AN ENERGY LANDSCAPING APPROACH TO THE PROTEIN FOLDING PROBLEM

Approved by:

Professor Harvey Lipkin, Committee Chair The George W. Woodruff School of Mechanical Engineering *Georgia Institute of Technology*

Professor Nader Sadegh The George W. Woodruff School of Mechanical Engineering *Georgia Institute of Technology*

Professor Michael J. Leamy The George W. Woodruff School of Mechanical Engineering *Georgia Institute of Technology* Professor Stephen C. Harvey School of Biology Georgia Institute of Technology

Professor Joel S. Sokol The School of Industrial System and Engineering Georgia Institute of Technology

Date Approved: 13 November 2009

To my grandmother,

Puangtip Tiamtrakul,

and my parents,

Pojana and Wut Sapsaman.

ACKNOWLEDGEMENTS

I would like express my sincere gratitude to everyone supporting me in completing this thesis.

My foremost gratitude goes to my adviser, Dr. Harvey Lipkin, a very knowledgeable individual who likes to see students succeed and from whom I have learned so much. Thank you for always leaving your door open.

I also would like to show my appreciation to Dr. Nader Sadegh, Dr. Michael Leamy, Dr. Steve Harvey, and Dr. Joel Sokol for serving as committee members, to Dr. Jeffrey Donnell for proofreading, and to Jane Chisholm for improving my writing skills.

My heartfelt thanks go to my officemate, my fellow Thai student, "Jiggy" Jomkwun Munnae, for the best advice, constant encouragement, and always being my best friend. My special thanks go to Thom Sokol for editing the thesis and giving constructive criticism, for constant love and support, and for always believing in me.

My deepest appreciation goes to the George W. Woodruff School of Mechanical Engineering for financial support and the opportunities to teach. I also would like to thank Mechanical Engineering Graduate office staff for helping with paperwork and giving emotional support.

Last but certainly not least, I would like to thank my family and Thom's family for encouragement and friends from Woodruff School Graduate Women and Thai Student Organization for great company.

Temsiri Sapsaman

TABLE OF CONTENTS

DED	ICAT	TION .		iii							
ACK	ACKNOWLEDGEMENTS										
LIST	OF	TABLE	S	xi							
LIST	OF	FIGUR	ES	xii							
LIST	OF	SYMBO	DLS OR ABBREVIATIONS	xx							
SUM	MAR	αY		xxiii							
Ι	INT	RODUC	CTION	1							
	1.1	Motiva	tion	1							
	1.2	Proteir	n Structure	2							
	1.3	Proteir	n Folding Process	3							
	1.4	Metho	ds of Protein Conformation Prediction	3							
		1.4.1	Molecular Dynamics	3							
		1.4.2	Robot Model	4							
		1.4.3	Optimization	5							
	1.5	Bottler	neck	6							
	1.6	Contril	bution	7							
	1.7	Organi	zation	8							
II	LITI	ERATU	RE REVIEW	10							
	2.1	Energy	Models	11							
		2.1.1	Force Field Model	11							
	2.2	Reduce	ed Molecular Models	13							
		2.2.1	HP Model	13							
		2.2.2	Residue-Level Models	13							
		2.2.3	Macro-Level Models and Rigidity Analysis	15							
	2.3	Molecu	ılar Dynamics (MD)	15							

		2.3.1	Traditional MD \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots	15
		2.3.2	Internal Coordinates	16
		2.3.3	Normal Mode Analysis	18
	2.4	Optim	lizations	19
		2.4.1	Classical Optimization Methods	19
			2.4.1.1 Line Search	20
		2.4.2	Monte Carlo	21
		2.4.3	De Novo Methods	22
			2.4.3.1 TASSER	22
			2.4.3.2 Rosetta	23
		2.4.4	Non-Traditional Optimization	24
	2.5	Energ	y Landscape Modification	24
		2.5.1	Hypersurface Deformation	24
		2.5.2	Energy Landscape Flattening	26
	2.6	Conclu	usion of the Literature Review	26
III	NEV TO	VTON' PROTI	S METHOD AND QUASI-NEWTON METHODS APPLIED	28
	31	Newto	m's Method	20 28
	0.1	311	Line Search	20 20
	39	Oussi-	Newton (ON) Methods	30
	3.3	Applic	eation to the Protein Folding Problem	33
	0.0	331	Energy and Derivatives Evaluation	34
		332	Newton's Method	35
		333	ON Methods	37
	34	Conclu	usion for Newton's Method and quasi-Newton	01
	0.1	Metho	ds Applied to Protein Folding Problem	37
IV	ENE	ERGY I	LANDSCAPING	41
	4.1	Expon	ential Energy Landscaping (XEL)	41
		4.1.1	Method	41

	4.1.2	Newton?	s Method with XEL (Newton-XEL)	44
	4.1.3	Quasi-N	ewton Algorithm with XEL (QNA-XEL) \ldots .	46
		4.1.3.1	Broyden's method with XEL (Broyden-XEL) $\ . \ .$.	47
	4.1.4	Broyden (BFGS-2	-Fletcher-Goldfarb-Shanno Algorithm with XEL XEL)	49
4.2	Discus	ssion on X	KEL	52
	4.2.1	A Positi	ve Definite Property of Newton-XEL Hessian Matrix	52
	4.2.2	Directio	n of Newton-XEL Step	54
		4.2.2.1	Derivation of the 2D Newton-XEL step	54
		4.2.2.2	N dimensional Newton-XEL step	57
	4.2.3	Properti	es of the Scalar Multiplier λ	59
		4.2.3.1	Eigenvalue λ	59
		4.2.3.2	Other eigenvalues	59
		4.2.3.3	A relationship between λ and a positive definite property	61
	4.2.4	Magnitu	de of Newton-XEL Step	62
	4.2.5	Error re	sidue from Newton-XEL step	62
	4.2.6	Weights	on Different Energy Landscape Information	65
		4.2.6.1	The Newton-XEL and the QNA-XEL (<i>Broyden's method</i>) Hessian matrices	65
		4.2.6.2	The QNA-XEL and the BFGS-XEL Hessian matrix and Hessian matrix inverse estimation	66
	4.2.7	Conclus	ion for Discussion on XEL	67
4.3	Adapt	ive Expo	nential Energy Landscaping (AXEL)	69
	4.3.1	Optimiz	ation with AXEL	69
		4.3.1.1	Adaptive- n XEL algorithms	70
		4.3.1.2	Varying- n XEL algorithms $\ldots \ldots \ldots \ldots \ldots$	75
	4.3.2	AXEL S	chemes	81
4.4	Conclu	usion for	the Energy Landscaping	82

V	SIM	IULATI	ION RESU	ULTS AND DISCUSSIONS	87						
	5.1	XEL .	XEL Algorithms without Probabilistic Search								
		5.1.1	Path Co	mparison	91						
			5.1.1.1	Newton-XEL and Newton-XEL (Hessian only) \cdot .	91						
			5.1.1.2	QNA-XEL and QNA-XEL (Hessian only)	102						
			5.1.1.3	BFGS-XEL and BFGS-XEL (Hessian only)	113						
			5.1.1.4	Conclusion of path comparison	123						
		5.1.2	Compar modified	ison of Energy and Number of Iterations to the Un- l Case	123						
			5.1.2.1	Newton-XEL and Newton-XEL (Hessian only) \cdot .	125						
			5.1.2.2	QNA-XEL and QNA-XEL (Hessian only)	131						
			5.1.2.3	BFGS-XEL and BFGS-XEL (Hessian only)	137						
			5.1.2.4	Conclusion of iterations and energy comparison to the unmodified case	144						
		5.1.3	Global S	Search Performance	148						
		5.1.4	Conclus	ion of XEL Algorithms without Probabilistic Search	154						
	5.2	XEL .	Algorithm	s with Probabilistic Search	157						
		5.2.1	Rosetta		157						
		5.2.2	Score In	provement	161						
			5.2.2.1	QNA-XEL and QNA-XEL (Hessian only) \ldots .	162						
			5.2.2.2	BFGS-XEL and BFGS-XEL (Hessian only) \ldots	169						
			5.2.2.3	Conclusion of score improvement	170						
		5.2.3	Lowest S	Score	171						
			5.2.3.1	QNA-XEL and QNA-XEL (Hessian only) \ldots	172						
			5.2.3.2	BFGS-XEL and BFGS-XEL (Hessian only) \ldots	175						
			5.2.3.3	Conclusion of lowest score	179						
		5.2.4	Average	Iterations	180						
			5.2.4.1	QNA-XEL and QNA-XEL (Hessian only) \ldots	182						
			5.2.4.2	BFGS-XEL and BFGS-XEL (Hessian only)	185						

			5.2.4.3	Conclusion of average iterations	189
		5.2.5	Efficienc	у	191
			5.2.5.1	QNA-XEL and QNA-XEL (Hessian only)	191
			5.2.5.2	BFGS-XEL and BFGS-XEL (Hessian only)	196
			5.2.5.3	Conclusion of efficiency	199
		5.2.6	Similarit	y to the native configuration	200
			5.2.6.1	Conclusion of similarity to the native configuration	205
		5.2.7	Further	analysis: Same-iteration comparison	205
			5.2.7.1	Conclusion of further analysis: same-iteration com- parison	211
		5.2.8	Conclusi	on of XEL Algorithms with Probabilistic Search	211
	5.3	BFGS	with AX	EL	218
		5.3.1	Global s	chemes	218
			5.3.1.1	Scheme A: Random n	219
			5.3.1.2	Schemes B and C: Running average efficiency	221
			5.3.1.3	Scheme D: Stricter convergence criterion	226
		5.3.2	Local scl	hemes	228
			5.3.2.1	Scheme E: Bounded error residue	229
			5.3.2.2	Scheme F: Path and landscape characteristic criteria	ı 230
		5.3.3	Conclusi	on of BFGS with AXEL	234
	5.4	Conclu	usion of S	imulation Results and Discussions	235
VI	CON	ICLUD	ING REN	MARKS	237
	6.1	Contri	butions .		237
	6.2	Future	e Work .		241
APP	END	IX A	SUPPL	EMENTARY DOCUMENTS	243
	A.1	QNA-2	XEL and	BFGS-XEL Combining with Probabilistic Search .	243
		A.1.1	QNA-XI	EL and QNA-XEL (<i>Hessian only</i>)	243
		A.1.2	BFGS-X	EL and BFGS-XEL (<i>Hessian only</i>)	243

REFEI	RE	ΞN	C	ES	5	•				•	•	•			•	•	•		•	•			•	•	•	•	252
VITA																											260

LIST OF TABLES

4.1	Guidelines for adaptively modified energy landscaping	68
4.2	A summary of presented methods: Newton, Newton-XEL, QNA, and QNA-XEL (1 of 4)	83
4.3	A summary of presented methods: BFGS and BFGS-XEL (2 of 4) $$.	84
4.4	A summary of presented methods: Adaptive- n XEL algorithms (3 of 4)	85
4.5	A summary of presented methods: Varying- n XEL algorithms (4 of 4)	86
5.1	A summary of the path characteristics from implemented algorithms.	124
5.2	A summary of the quality and the speed results: XEL algorithms. (1 of 2)	146
5.3	A summary of the quality and the speed results: XEL (<i>Hessian only</i>) algorithms. (2 of 2)	147
5.4	The n case in which the lowest score of each protein is found	181
5.5	The range of the percent difference of the average iteration $\% \Delta AveIt$ yielded by different algorithms with different values of $n. \ldots \ldots$	213
5.6	The range of the percent difference of the average score improvement $\% \Delta AveSI$ yielded by different algorithms with different values of n .	214
5.7	The average TM, GDT, MaxSub scores, and RMSD values for XEL with various values of n and two special runs with more decoys or perturbations.	215
5.8	The percent difference of average TM, GDT, MaxSub scores, and RMSD values compared to $n = 1$ for XEL with various values of n and two special runs with more decoys or perturbations	216
5.9	The values of n for Schemes B and C which give the best efficiency during the different simulation periods for each protein	224
5.10	Two sets of criterion parameters for Scheme F.	232

LIST OF FIGURES

1.1	A series of three peptide planes	2
3.1	A pseudo-code for Newton's method with a line search $\ \ldots \ \ldots \ \ldots$	30
3.2	A pseudo-code for the QN method with a line search and Broyden's method	32
3.3	A pseudo-code for BFGS algorithm with a line search	33
3.4	A simple rotational mass-spring system.	36
3.5	A pseudo-code for Newton's method with a line search applied to the protein folding problem	38
3.6	A pseudo-code for QNA with a line search applied to the protein folding problem	39
3.7	A pseudo-code for BFGS algorithm with a line search applied to the protein folding problem	40
4.1	Comparison of unmodified energy landscape with $n = 1$, XEL with $n = 2$, and XEL with $n = 0.5$	42
4.2	Comparison of unmodified energy landscape with $n = 1$, XEL with $n = 2$, and XEL with $n = 0.5$ in the range of $ E < 1. \ldots \ldots$	43
4.3	A pseudo-code for Newton-XEL with a line search. Note that for Newton-XEL (<i>Hessian only</i>), described in Section 4.1.4, $\tau_k^* = \tau_k$.	46
4.4	A pseudo-code for QNA-XEL with a line search. Note that for QNA-XEL (<i>Hessian only</i>), described in Section 4.1.4, $\tau_k^* = \tau_k \dots \dots \dots$	48
4.5	A pseudo-code for BFGS-XEL with a line search. Note that for BFGS-XEL (<i>Hessian only</i>) $\tau_k^* = \tau_k$.	51
4.6	A pseudo-code for varying- <i>n</i> Newton-XEL with a line search. Note that for varying- <i>n</i> Newton-XEL (<i>Hessian only</i>) $\tau_k^* = \tau_k$	78
4.7	A pseudo-code for varying- <i>n</i> QNA-XEL with a line search. Note that for varying- <i>n</i> QNA-XEL (<i>Hessian only</i>) $\tau_k^* = \tau_k$	79
4.8	A pseudo-code for varying- <i>n</i> BFGS-XEL with a line search. Note that for varying- <i>n</i> BFGS-XEL (<i>Hessian only</i>) $\tau_k^* = \tau_k$	80
5.1	The simple and the complex landscape used to generate optimization paths for all algorithms.	89
5.2	Twenty-five starting points for paths generation	90

5.3	Paths from 1 st through 5 th starting points on a simple energy landscape generated with Newton-XEL.	93
5.4	Paths from 1^{st} through 5^{th} starting points on a simple energy landscape generated with Newton-XEL (<i>Hessian only</i>)	94
5.5	Paths from 1 st through 5 th starting points on a simple energy landscape generated with Newton-XEL.	95
5.6	Paths from 1 st through 5 th starting points on a simple energy landscape generated with Newton-XEL (<i>Hessian only</i>)	96
5.7	Paths from 1 st through 5 th starting points on a complex energy land- scape generated with Newton-XEL.	98
5.8	Paths from 1^{st} through 5^{th} starting points on a complex energy land- scape generated with Newton-XEL (<i>Hessian only</i>)	99
5.9	Paths from 1 st through 5 th starting points on a complex energy land- scape generated with Newton-XEL.	100
5.10	Paths from 1^{st} through 5^{th} starting points on a complex energy land- scape generated with Newton-XEL (<i>Hessian only</i>)	101
5.11	Paths from 1 st through 5 th starting points on a simple energy landscape generated with QNA-XEL	104
5.12	Paths from 1 st through 5 th starting points on a simple energy landscape generated with QNA-XEL (<i>Hessian only</i>).	105
5.13	Paths from 1 st through 5 th starting points on a simple energy landscape generated with QNA-XEL	106
5.14	Paths from 1^{st} through 5^{th} starting points on a simple energy landscape generated with QNA-XEL (<i>Hessian only</i>)	107
5.15	Paths from 1 st through 5 th starting points on a complex energy land- scape generated with QNA-XEL.	109
5.16	Paths from 1^{st} through 5^{th} starting points on a complex energy land- scape generated with QNA-XEL (<i>Hessian only</i>)	110
5.17	Paths from 1 st through 5 th starting points on a complex energy land- scape generated with QNA-XEL.	111
5.18	Paths from 1 st through 5 th starting points on a complex energy land- scape generated with QNA-XEL (<i>Hessian only</i>)	112
5.19	Paths from 1 st through 5 th starting points on a simple energy landscape generated with BFGS-XEL.	114

5.20	Paths from 1 st through 5 th starting points on a simple energy landscape generated with BFGS-XEL (<i>Hessian only</i>)	115
5.21	Paths from 1 st through 5 th starting points on a simple energy landscape generated with BFGS-XEL.	116
5.22	Paths from 1 st through 5 th starting points on a simple energy landscape generated with BFGS-XEL (<i>Hessian only</i>)	117
5.23	Paths from 1 st through 5 th starting points on a complex energy land- scape generated with BFGS-XEL.	119
5.24	Paths from 1 st through 5 th starting points on a complex energy land- scape generated with BFGS-XEL (<i>Hessian only</i>)	120
5.25	Paths from 1 st through 5 th starting points on a complex energy land- scape generated with BFGS-XEL.	121
5.26	Paths from 1^{st} through 5^{th} starting points on a complex energy land- scape generated with BFGS-XEL (<i>Hessian only</i>)	122
5.27	Newton-XEL: The percentage of paths with the lower, the equal, or the higher number of iterations or energy compared to $n = 1$. Paths are generated by a bracketing and Brent line search	126
5.28	Newton-XEL (<i>Hessian only</i>): The percentage of paths with the lower, the equal, or the higher number of iterations or energy compared to $n = 1$. Paths are generated by a bracketing and Brent line search	127
5.29	Newton-XEL: The percentage of paths with the lower, the equal, or the higher number of iterations or energy compared to $n = 1$. Paths are generated by a heuristic exhaustive line search.	128
5.30	Newton-XEL (<i>Hessian only</i>): The percentage of paths with the lower, the equal, or the higher number of iterations or energy compared to $n = 1$. Paths are generated by a heuristic exhaustive line search	129
5.31	QNA-XEL: The percentage of paths with the lower, the equal, or the higher number of iterations or energy compared to $n = 1$. Paths are generated by a bracketing and Brent line search.	132
5.32	QNA-XEL (<i>Hessian only</i>): The percentage of paths with the lower, the equal, or the higher number of iterations or energy compared to $n = 1$. Paths are generated by a bracketing and Brent line search	133
5.33	QNA-XEL: The percentage of paths with the lower, the equal, or the higher number of iterations or energy compared to $n = 1$. Paths are generated by a heuristic exhaustive line search	134

5.34	QNA-XEL (<i>Hessian only</i>): The percentage of paths with the lower, the equal, or the higher number of iterations or energy compared to n = 1. Paths are generated by a heuristic exhaustive line search	135
5.35	BFGS-XEL: The percentage of paths with the lower, the equal, or the higher number of iterations or energy compared to $n = 1$. Paths are generated by a bracketing and Brent line search	138
5.36	BFGS-XEL (<i>Hessian only</i>): The percentage of paths with the lower, the equal, or the higher number of iterations or energy compared to $n = 1$. Paths are generated by a bracketing and Brent line search	139
5.37	BFGS-XEL: The percentage of paths with the lower, the equal, or the higher number of iterations or energy compared to $n = 1$. Paths are generated by a heuristic exhaustive line search	140
5.38	BFGS-XEL (<i>Hessian only</i>): The percentage of paths with the lower, the equal, or the higher number of iterations or energy compared to n = 1. Paths are generated by a heuristic exhaustive line search	141
5.39	The number of paths on a simple or a complex energy landscape gener- ated by Newton-XEL, QNA-XEL, or BFGS-XEL that locate the global minimum. Paths are generated by a bracketing and Brent line search.	149
5.40	The number of paths on a simple or a complex energy landscape gen- erated by Newton-XEL (<i>Hessian only</i>), QNA-XEL (<i>Hessian only</i>), or BFGS-XEL (<i>Hessian only</i>) that locate the global minimum. Paths are generated by a bracketing and Brent line search.	150
5.41	The number of paths on a simple or a complex energy landscape gener- ated by Newton-XEL, QNA-XEL, or BFGS-XEL that locate the global minimum. Paths are generated by a heuristic exhaustive line search.	151
5.42	The number of paths on a simple or a complex energy landscape gen- erated by Newton-XEL (<i>Hessian only</i>), QNA-XEL (<i>Hessian only</i>), or BFGS-XEL (<i>Hessian only</i>) that locate the global minimum. Paths are generated by a heuristic exhaustive line search.	152
5.43	Rosetta centroid refinement protocol	159
5.44	QNA-XEL: The average of the score improvement $(AveSI)$ versus various values of $n. \ldots \ldots$	163
5.45	QNA-XEL (<i>Hessian only</i>): The average of the score improvement $(AveSI)$ versus various values of n .	164
5.46	QNA-XEL: The percent difference between the average score improve- ment of the various values of n and those of the unmodified energy landscape $(n = 1)$ case, $\Delta AveSI_n$	165

5.47	QNA-XEL (<i>Hessian only</i>): The percent difference between the average score improvement of the various values of n and those of the unmodified energy landscape $(n = 1)$ case, $\Delta AveSI_n$.	166
5.48	BFGS-XEL: The average of the score improvement $(AveSI)$ versus various values of n	167
5.49	BFGS-XEL (<i>Hessian only</i>): The average of the score improvement $(AveSI)$ versus various values of n	168
5.50	BFGS-XEL: The percent difference between the average score improve- ment of the various values of n and those of the unmodified energy landscape $(n = 1)$ case, $\Delta AveSI_n$	170
5.51	BFGS-XEL (<i>Hessian only</i>): The percent difference between the average score improvement of the various values of n and those of the unmodified energy landscape $(n = 1)$ case, $\Delta AveSI_n$.	171
5.52	QNA-XEL: The lowest scores (E_{lowest}) versus various values of n	173
5.53	QNA-XEL (<i>Hessian only</i>): The lowest scores (E_{lowest}) versus various values of n	174
5.54	QNA-XEL: The percent difference of the lowest score $(\% \Delta E_{\text{lowest}})$ compared to $n = 1$.	175
5.55	QNA-XEL (<i>Hessian only</i>): The percent difference of the lowest score $(\%\Delta E_{\text{lowest}})$ compared to $n = 1. \ldots \ldots \ldots \ldots$	176
5.56	BFGS-XEL: The lowest scores (E_{lowest}) versus various values of n	177
5.57	BFGS-XEL (<i>Hessian only</i>): The lowest scores (E_{lowest}) versus various values of n	178
5.58	BFGS-XEL: The percent difference of the lowest score $(\% \Delta E_{\text{lowest}})$ compared to $n = 1, \ldots, \ldots, \ldots, \ldots, \ldots, \ldots$	179
5.59	BFGS-XEL (<i>Hessian only</i>): The percent difference of the lowest score $(\%\Delta E_{\text{lowest}})$ compared to $n = 1. \ldots \ldots \ldots \ldots \ldots$	180
5.60	QNA-XEL: Average iterations $(AveIt)$ versus various values of n	183
5.61	QNA-XEL (<i>Hessian only</i>): Average iterations ($AveIt$) versus various values of n	184
5.62	QNA-XEL: The percent difference of average iterations $(\% \Delta AveIt_n)$ compared to $n = 1, \ldots, \ldots, \ldots, \ldots, \ldots, \ldots, \ldots$	185
5.63	QNA-XEL (<i>Hessian only</i>): The percent difference of average iterations $(\% \Delta AveIt_n)$ compared to $n = 1. \ldots \ldots \ldots \ldots \ldots \ldots$	186
5.64	BFGS-XEL: Average iterations $(AveIt)$ versus various values of $n.$	187

5.65	BFGS-XEL (<i>Hessian only</i>): Average iterations ($AveIt$) versus various values of n .	188
5.66	BFGS-XEL: The percent difference of average iterations $(\% \Delta AveIt_n)$ compared to $n = 1. \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	189
5.67	BFGS-XEL (<i>Hessian only</i>): The percent difference of average itera- tions ($\% \Delta AveIt_n$) compared to $n = 1. \ldots \ldots \ldots$	190
5.68	QNA-XEL: The efficiency (e_n) versus various values of n	192
5.69	QNA-XEL (<i>Hessian only</i>): The efficiency (e_n) versus various values of n	.193
5.70	QNA-XEL: The percent difference of the efficiency $(\%\Delta e_n)$ compared to $n = 1$ versus various values of n	194
5.71	QNA-XEL (<i>Hessian only</i>): The percent difference of the efficiency $(\%\Delta e_n)$ compared to $n = 1$ versus various values of $n. \ldots \ldots$	195
5.72	BFGS-XEL: efficiency (e_n) versus various values of n	197
5.73	BFGS-XEL (<i>Hessian only</i>): efficiency (e_n) versus various values of n .	198
5.74	BFGS-XEL: The percent difference of the efficiency $(\% \Delta e_n)$ compared to $n = 1$ versus various values of n	199
5.75	BFGS-XEL (<i>Hessian only</i>): The percent difference of the efficiency $(\%\Delta e_n)$ compared to $n = 1$ versus various values of $n. \ldots \ldots$	200
5.76	BFGS-XEL (<i>Hessian only</i>): The scores versus TM scores of resulting configurations with various values of n for protein 1ubq	201
5.77	BFGS-XEL (<i>Hessian only</i>): The scores versus GTD scores of resulting configurations with various values of n for protein 1ubq	202
5.78	BFGS-XEL (<i>Hessian only</i>): The scores versus MaxSub scores of resulting configurations with various values of n for protein 1ubq	203
5.79	BFGS-XEL (<i>Hessian only</i>): The scores versus RMSD values of result- ing configurations with various values of n for protein 1ubq	204
5.80	Same-Iteration: a) the percent difference of average score improvement $\% \Delta AveSI$, b) the percent difference of lowest score $\% \Delta E_{lowest}$, c) the percent difference of average iteration $\% \Delta AveIt$, and d) the percent difference of the efficiency $\% \Delta e_n$ from special runs with $n = 2^{-9}$ and either more decoys (Decoys+) or more perturbations (Perturb+) and the BFGS-XEL (<i>Hessian only</i>) without additional iterations. All runs use $n = 2^{-9}$ and are performed for proteins 1ubq and dom6	207

5.81	BFGS-XEL (<i>Hessian only</i>): The scores versus the TM, the GTD, the MaxSub scores and the RMSD values of resulting configurations for protein 1ubq. These quantities are from special runs with $n = 2^{-9}$ and either more decoys (Decoys+) or more perturbations (Perturb+). They are compared to those for $n = 1$ without varying the numbers of original decoys or perturbations.	209
5.82	BFGS-XEL (<i>Hessian only</i>): The scores versus the TM, the GTD, the MaxSub scores and the RMSD values of resulting configurations for protein 1ubq. These quantities are from special runs with $n = 2^{-9}$ and either more decoys (Decoys+) or more perturbations (Perturb+). They are compared to those of $n = 2^{-9}$ without varying the numbers of original decoys or perturbations.	210
5.83	Scheme A: a) the percent difference of average score improvement $\% \Delta AveSI$, b) the percent difference of lowest score $\% \Delta E_{lowest}$, c) the percent difference of average iteration $\% \Delta AveIt$, and d) the percent difference of the efficiency $\% \Delta e_n$.	220
5.84	The running average efficiency versus iterations for protein $1d3z$ with various values of n .	222
5.85	The running average efficiency (log scale) versus iterations (log scale) for protein 1d3z with various values of n .	223
5.86	Schemes B and C: a) the percent difference of average score improve- ment $\% \Delta AveSI$, b) the percent difference of lowest score $\% \Delta E_{lowest}$, c) the percent difference of average iteration $\% \Delta AveIt$, and d) the percent difference of the efficiency $\% \Delta e_n$.	225
5.87	Scheme D: a) the percent difference of average score improvement $\% \Delta AveSI$, b) the percent difference of lowest score $\% \Delta E_{lowest}$, c) the percent difference of average iteration $\% \Delta AveIt$, and d) the percent difference of the efficiency $\% \Delta e_n$.	227
5.88	Schemes E: a) the percent difference of average score improvement $\% \Delta AveSI$, b) the percent difference of lowest score $\% \Delta E_{lowest}$, c) the percent difference of average iteration $\% \Delta AveIt$, and d) the percent difference of the efficiency $\% \Delta e_n$.	230
5.89	A pseudo-code for Scheme F with varying n BFGS-XEL	231
5.90	Scheme F: a) the percent difference of average score improvement $\% \Delta AveSI$, b) the percent difference of lowest score $\% \Delta E_{lowest}$, c) the percent difference of average iteration $\% \Delta AveIt$, and d) the percent difference of the efficiency $\% \Delta e_n$ from two sets of criterion paremeters.	233

A.1	QNA-XEL: The percentage of decoys with less, equal, higher scores or number of iterations compared to $n = 1$ for proteins 1ubq (a and d), 1d3z (b and e), and dom6 (c and f).	244
A.2	QNA-XEL: The percentage of decoys with less, equal, higher scores or number of iterations compared to $n = 1$ for proteins hetr (g and i) and mdmi (h and j).	245
A.3	QNA-XEL (<i>Hessian only</i>): The percentage of decoys with less, equal, higher scores or number of iterations compared to $n = 1$ for proteins 1ubq (a and d), 1d3z (b and e), and dom6 (c and f).	246
A.4	QNA-XEL (<i>Hessian only</i>): The percentage of decoys with less, equal, higher scores or number of iterations compared to $n = 1$ for proteins hetr (g and i) and mdmi (h and j).	247
A.5	BFGS-XEL: The percentage of decoys with less, equal, higher scores or number of iterations compared to $n = 1$ for proteins 1ubq (a and d), 1d3z (b and e), and dom6 (c and f).	248
A.6	BFGS-XEL: The percentage of decoys with less, equal, higher scores or number of iterations compared to $n = 1$ for proteins hetr (g and i) and mdmi (h and j).	249
A.7	BFGS-XEL (<i>Hessian only</i>): The percentage of decoys with less, equal, higher scores or number of iterations compared to $n = 1$ for proteins 1ubq (a and d), 1d3z (b and e), and dom6 (c and f)	250
A.8	BFGS-XEL (<i>Hessian only</i>): The percentage of decoys with less, equal, higher scores or number of iterations compared to $n = 1$ for proteins hetr (g and i) and mdmi (h and j).	251

LIST OF SYMBOLS OR ABBREVIATIONS

- **AXEL** Adaptive Exponential Energy Landscaping.
- **BFGS** Broyden-Fletcher-Goldfarb-Shanno algorithm.
- **BFGS-XEL** Broyden-Fletcher-Goldfarb-Shanno algorithm with Exponential Energy Landscaping.

Broyden-XEL Broyden's Method with Exponential Energy Landscaping.

Newton-XEL Newton's method with Exponential Energy Landscaping.

QNA Quasi-Newton Algorithm.

QNA-XEL Quasi-Newton Algorithm with Exponential Energy Landscaping.

XEL Exponential Energy Landscaping.

 α Line search gain.

 α^* Line search gain that implicitly calculates β .

$$\beta \qquad \qquad \beta = n \left| E \right|^{n-1}.$$

 β^k Redefined β or β with n_k : $\beta_k^k = n_k |E_k|^{n_k-1}$ and $\beta_{k-1}^k = n_k |E_{k-1}|^{n_k-1}$.

 δ Bounding constant.

 κ^* Hessian matrix of the adaptively modified energy landscape.

- $\hat{\kappa}^*$ Estimated Hessian matrix of the adaptively modified energy landscape.
- λ^{Θ} Scalar multiplier for varying-*n* Newton-XEL step: $\hat{h}_{\theta}^* = \lambda^{\Theta} \hat{h}$.

 ρ Error residue of Newton's method.

- ρ^* Error residue of Newton-XEL.
- au Torque vector.

 τ^* Torque vector of the modified energy landscape: $\tau^* = \beta \tau$.

$$\Delta \tau \qquad \Delta \tau_{k-1} = \tau_k - \tau_{k-1}.$$

$$\Delta \tau^* \qquad \quad \Delta \tau_{k-1}^* = \tau_k^* - \tau_{k-1}^*.$$

$$\Delta \tau^{*k}$$
 Redefined $\Delta \tau^{*}$ or $\Delta \tau^{*}$ with n_k : $\Delta \tau^{*k}_{k-1} = \beta_k^k \tau_k - \beta_{k-1}^k \tau_{k-1}$.

 θ A vector variable of a cost function. For protein conformation prediction, the dihedral angle vector.

Θ	A composite vector variable of the dihedral angle vector and the exponent of the XEL.
AveSI	Average score improvement.
$\%\Delta AveSI$	The percent difference of the average score improvement.
AveIt	Average iteration.
$\%\Delta AveIt$	The percent difference of the average iteration.
e	Efficiency.
$\%\Delta e$	The percent difference of the efficiency.
E	Energy function.
E^*	Modified energy function.
E_{lowest}	Lowest energy.
$\% \Delta E_{\rm lowest}$	The percent difference of the lowest energy.
f	Cost function.
h	A step for the dihedral angles.
\hat{h}	An optimization direction. For protein conformation prediction, a change in the dihedral angle vector.
h^*	A step for the dihedral angles of the modified energy function.
\hat{h}^*	An optimization direction of the XEL algorithm. For protein conformation prediction, a change in the dihedral angle vector.
$\hat{h}_{ heta}^{*}$	A step for θ .
\hat{h}_{Θ}^{*}	A step for Θ .
\hat{h}_n^*	A step for n .
\hat{H}	The estimation of the Hessian matrix inverse K^{-1} .
\hat{H}^*	The estimation of the modified Hessian matrix inverse K^{*-1} .
K	The second derivative or a Hessian matrix. For protein conformation pre- diction, the stiffness matrix.
Ŕ	The estimation of the second derivative or a Hessian matrix. Hessian matrix of the QNA method. For protein conformation prediction, the estimated stiffness matrix.

- K^* The second derivative or the Hessian matrix of the modified energy land-scape.
- \hat{K}^* The estimated second derivative or the estimated Hessian matrix of the modified energy landscape.
- n The exponent of the XEL.
- *SI* Score improvement.

SUMMARY

The function of a protein is largely dictated by its natural shape called the "native conformation." Since the native conformation and the global minimum energy configuration highly correlate, predicting this conformation is a global optimization known as the "protein folding problem." It is computationally intensive due to the high-dimensional and complex energy landscape. Typical conformation algorithms combine a probabilistic search with analytical optimization [56]. The analytical portion typically takes longer than the probabilistic part since more function evaluations are required, which are algorithm bottlenecks.

To reduce the computational cost, this research studies the effects of exponential energy landscaping (XEL) on three analytical optimization algorithms: Newton's method, a quasi-Newton algorithm (QNA), and the Broyden-Fletcher-Goldfarb-Shanno (BFGS) algorithm. The XEL changes the heights and the depths of the extrema but keeps their location the same, which eliminates the troublesome process of remapping minima onto the original landscape. The Newton-XEL is found to have a similar convergence property as Newton's method by showing that their error residues are of the same order. Found by observation, stability and convergence are improved when the error residue is bounded. While XEL is found to have no effect on the similarity of resulting configurations to the native conformation, results show that the XEL can improve the speed in terms of average iterations in the QNA by 47% and in the BFGS by 41%. In terms of the average score improvement, which indicates how the energy of the resulting configuration is compared to that of the initial configuration, the XEL can improve the quality of resulting configurations in the QNA by 12% and in the BFGS by 10%. Since both results were not achieved simultaneously, the adaptive exponential energy landscaping (AXEL) is developed. The results lead to the conclusion that a trade-off between quality and speed must be considered when XEL is implemented. To improve speed by 15% to 47% and efficiency by 13% to 75%, XEL with n within $2^{-9}-2^{-5}$ should be used and to improve quality by 4% to 7%, AXEL with Scheme E that keeps the error residue bounded should be used.

CHAPTER I

INTRODUCTION

1.1 Motivation

The chemicals that play the primary role in performing, catalyzing, or supporting the biological functions of organisms are proteins [35]. As researchers have known for many years, the functionality of proteins is largely determined from their structural conformations. For example, in the human immune system, an antibody has a shape that only a certain antigen can fit. In drug design, the functions of new proteins are traditionally determined by conducting numerous experimental trials. However, if the protein conformation is known in advance the number of experimental trials can be reduced by using computer simulation to determine substances whose shapes may potentially react to the protein. This process could increase the success rate of pharmaceutical trials.

Since the native conformation and the global minimum energy configuration highly correlate, predicting the protein conformation is a global minimization problem. It is a computationally intensive task due to the high-dimensional and the complex energy landscape. Because on average a protein molecule has several thousands atoms, it can have several hundreds degrees of freedom even with a reduced model. This creates the large energy landscape and an extremely intensive search is generally required. Additionally, a very rough energy landscape also makes the minimization difficult because of more local minima. With atomic energy associated to each atom pair, the energy of the protein drastically varies as the protein configuration changes. In [11] the protein conformation prediction process takes about 150 CPU days for each small protein.

1.2 Protein Structure

A protein molecule is a chain of amino acid units known as residues [36]. The sequence of amino acids is called a protein sequence. Twenty types of amino acids are identified by their side chains, indicated by R_i in Figure 1.1, which illustrates a series of four residues. These side chains connect to alpha-carbon atoms of the amino acids and have zero to four degrees of rotational freedom. The configurations of the side chains depend on the values of their rotation angles, known as the Rotamer angles.



Figure 1.1: A series of three peptide planes.

Between the pairs of residues strong chemical bonds are formed between nitrogen and carbon atoms called "peptide bonds." Each of these bonds creates a relatively rigid plane between two residues, called a "peptide plane." The peptide plane consists of alpha-carbon (C_{α}), carbon (C), and oxygen (O) atoms from one residue and nitrogen (N), hydrogen (H), and alpha-carbon atoms from another. Between each pair of peptide planes, the molecule can rotate along the N–C_{α} and C_{α}–C bonds. The values of these rotation angles, ϕ and ψ , called "dihedral angles," largely determine the shape of a protein.

1.3 Protein Folding Process

Protein folding is the process by which a protein chain transforms into a stable conformation [34, 35, 36]. The process often starts when a protein chain is stretched out and non-functional, in a "de-natured" state. The protein is unstable and will naturally fold to obtain the stable "native state" due to atomic forces. In the native state the protein chain is naturally clustered and has minimum potential energy. Some proteins can have more than one native conformation in different environments or solvents. However, when the environmental properties are constant the protein always folds into one native conformation. Interestingly, the folding path by which this conformation is reached does not have a monotonically-decreasing potential energy. Instead, the protein often encounters energetic barriers before it reaches the conformation that has minimum potential energy. In overcoming these energy barriers the protein has to fold into a higher energy configuration and move against atomic forces. This motion is a result of non-zero kinetic energy occurring as a consequence of a decrease in potential energy. Researchers believe that these energy barriers separate the native state from the de-natured state and prevent the protein from unfolding in response to small disturbances [28].

1.4 Methods of Protein Conformation Prediction

Also known as the protein folding problem, the protein conformation prediction problem is the challenge of determining the native conformation of a new protein. This problem is approached with several methods of conformation prediction, including molecular dynamics simulation and optimization.

1.4.1 Molecular Dynamics

The effort to determine the conformation of the native state has involved the use of molecular dynamics (MD) simulation [14] which simulates the motion of a molecule.

The time step of an accurate simulation must be very small, typically on the order of femtoseconds (10^{-15} s) . In contrast, the folding process takes a much longer time, typically on the order of milliseconds or seconds. Therefore, the MD simulations are computationally expensive [60]. This technique can locate the final conformation for a very small molecule but it is inefficient for larger proteins [7]. Since the smallest amino acid has 10 atoms and roughly 25 degrees of freedom that allow bond stretching, angle bending, and torsional motions, an atomic-level model of a small protein with 100 residues approximately contains at least 2,500 degrees of freedom. As a result, dynamic simulation is impractical because of time constraints. Many studies address this problem through reducing the degrees of freedom of the protein molecule model [53, 54]. Since the motions of atoms on a peptide plane relative to one another are very small, one way to significantly reduce the degrees of freedom is to treat the peptide planes as rigid bodies. This model only allows motions through changes in the dihedral and the Rotamer angles. Because each residue has two dihedral angles, the main chain of an *n*-residue protein model has 2n degrees of freedom. Although this reduction of computational costs could be an order of magnitude, the MD simulation is still not practical for solving the folding problem of large proteins.

Another inaccuracy arises from the empirical multidimensional energy function, also called a force field model, which is used to calculate atomic energies and forces. Expressing the potential energies of a protein molecule as a function of the molecular structure, the force field model yields a reduction of the computational costs compared to a quantum mechanics approach but is less accurate [44].

1.4.2 Robot Model

In addition to computational cost reduction, another advantage of modeling a molecule as a chain of rigid bodies is that robotic kinematics, dynamics, and controls methods can be applied if the protein molecule is considered as a nanomanipulator [3, 34, 36, 61]. A serial manipulator has revolute-joint angles and rigid links corresponding to dihedral angles and peptide planes respectively. An example of robotic studies is the work by Sharma *et al.* [61] who identify the protein reachable workspace using protein kinematics. However, an application of this workspace analysis may not be available in the near future because independently manipulating each individual joint of the molecule is not yet possible. Other studies [29, 30, 71, 75] present more sophisticated models that reduce the degrees of freedom further by identifying the rigid parts of the known protein chain due to constraints such as covalent and hydrogen bonds and only allow motions at the flexible parts of the chain. These techniques are helpful for the study of the operational protein motion such as binding, but they do not help the study of the folding process from the de-natured state because the constraints are unknown.

1.4.3 Optimization

Since at the native state proteins are in a minimum energy configuration, most research on protein conformation prediction uses optimization algorithms that search the conformational space to locate the conformation with the minimum energy. The current statistics on the PDB data show that on average a protein has 587 residues and the maximum protein size is 18,540 residues [1]. In this high-dimensional space, an extremely intensive search is therefore required. Additionally, the difficulties of the search arise due to the imprecise and highly complex force field model.

Although there are numerous algorithms for the protein-folding problem, all are variations of an analytical or statistical optimization algorithm, or a combination of both. The most popular statistical optimization algorithm used in the proteinfolding problem is likely the Monte Carlo (MC) method [44], which is the basis of most probabilistic searches [56]. MC simulation generates a collection of statistical configurations (or decoys) by adding random changes to the current configuration and accepting or rejecting them under the Metropolis criterion [44]. For a large protein, the MC algorithm can be inefficient because most generated configurations will be rejected. Therefore, a combination of MC and other optimization techniques is generally used in practice. For example, the Rosetta software [56], which is discussed in more detail in Chapter 5, randomly chooses pre-generated changes that are more energetically suitable for parts of the protein sequence to increase the acceptance rate.

One example of an analytical optimization algorithm, the method of steepest descent (SD), that is used in conjunction with the robotic model of the protein molecule is the work by Kazerounian *et al.* [34]. Noting that the minimum-energy conformation has zero torques about the dihedral angle axes, they developed the successive kineto-static compliance (SKSC) method which is a simple control law that alters the protein configuration until zero torques are reached. The algorithm is capable of finding a configuration with joint torques close to zero; however, this configuration does not guarantee the native state but rather a local minimum at which the torques are also zero. Due to the complexity of the energy landscape, this criterion is not sufficient for finding the global minimum energy. Their extended work [9] modifies the energy landscape by using segmental manipulation and progressive side-chain scaling. Although the new algorithm yields better results, the final configurations are still not in the native configuration.

1.5 Bottleneck

Analytical optimization algorithms can locate a local solution faster than probabilistic optimization algorithms, but the analytical optimization algorithms by themselves are unsuccessful in a global search because of the high-dimensional and complex energy landscape. Therefore, most protein conformation prediction algorithms combine a probabilistic search and analytical optimization algorithms [56]. The combined algorithms probabilistically generate several initial configurations (decoys) for local searching so to cover more areas in the energy landscape. Each local search progresses by slightly changing the shape of each decoy to yield an initial guess and finds a local minimum near the perturbed configuration. Then the resulting configuration is perturbed again to yield another initial guess. This cycle of perturbation and minimization is usually repeated hundreds of times and is called "refinement". There are usually several rounds of refinement and only most promising decoys are refined in further rounds.

Because more function evaluations are required, the analytical optimization algorithms usually take longer than the probabilistic portion and they become bottlenecks in the protein conformation prediction process. The efficiency of the analytical minimization becomes more problematic due to the following reasons: 1) high degrees of freedom in the protein molecule yields a high-dimensional minimization problem which is difficult and time consuming, 2) the refinement process involves several hundreds cycles of minimizations, and 3) to cover a large search area several thousands of decoys are needed even for a small protein and these decoys go through refinement. As the protein gets larger this problem gets magnified. To put it into perspective, in a high-resolution structure prediction [11] 20,000 to 30,000 decoys are generated for each small protein with 85 or fewer residues and the entire process for each protein takes 150 CPU days. To reduce the computational cost, the efficiency of the analytical algorithms needs to be addressed.

1.6 Contribution

This work investigates the effects of the method of energy landscaping on three analytical algorithms: Newton's method, a quasi-Newton algorithm (QNA) and the Broyden-Fletcher-Goldfarb-Shanno (BFGS) algorithm. The investigated landscaping, called the method of Exponential Energy Landscaping (XEL) [58], changes the heights and the depths of the extrema but keeps their location the same, which eliminates the troublesome process of remapping minima onto the original landscape. The simulation results show that in general the method of XEL uniformly affects the performance of all investigated algorithms. Since different degrees of the energy landscape modification yield different performances, an adaptively modified energy landscape (AXEL) scheme has been developed to dynamically modify the energy landscape based on the current performance of the algorithm. For minimization on some of the AXEL, varying-n XEL algorithms have been developed, taking into account the effect of the changing energy landscape. The effects of AXEL schemes on the XEL and the varying-n XEL variations of the three above algorithms are investigated.

The goal of this work is to improve the speed and quality of the analytical optimization algorithms using an exponential energy landscaping. With this energy landscaping, the computational cost can be reduced by up to 47% and a protein conformation with up to 30% lower energy can be located. Although both results were not achieved simultaneously, the exponential energy landscaping can speed up the prediction process in balancing quality of resulting configurations and computational cost.

1.7 Organization

This thesis is organized into six chapters. Chapter I discusses the motivation of the research, introduces the protein structure and the protein folding problem, and briefly reviews the methods of protein conformation prediction. Chapter II presents the literature review of several aspects in the protein conformation prediction problem, such as problem formulation and algorithms. Chapter III discusses the theoretical background for two analytical optimization algorithms, Newton's method and quasi-Newton methods, and their applications to the protein folding problem. Chapter IV presents and discusses the method of exponential energy landscaping and the adaptively modified energy landscape. Chapter V presents simulation results, observations, and discussions. Lastly, Chapter VI presents a conclusion and discusses future work.

CHAPTER II

LITERATURE REVIEW

Protein analysis can be basically categorized into two problems. The first is to identify the sequence of amino acids in a protein when the conformation is known [82], and the second is to identify the conformation of a protein when the sequence is known. Both conformation and sequence predictions are studied in single proteins as well as in protein-protein interactions.

The methods of predicting protein conformation are classified into three groups: comparative modeling, fold recognition, and *ab initio* methods. Both comparative modeling and fold recognition methods assume that the conformation of an unknown protein is comparable to that of a known protein with a similar sequence. Utilizing the protein database (PDB), which contains comprehensive data of studied proteins, both methods compare the sequence of the unknown to that of known proteins. Because the comparative modeling applies homology to identify sequence similarity, the predicted conformation is more accurate if the homologous relationship is established. Rating the compatibility of the unknown protein to known structures, the fold recognition methods are more applicable if the sequence similarity can not be established.

Unlike the other two methods, the *ab initio* method does not use conformational information from the PDB but uses a force field model which has parameters empirically developed from the PDB. The *ab initio* method usually determines conformation using a molecular dynamic simulation or an optimization search. To improve the performance of dynamic simulations and optimization approaches, researchers have focused on improving both the molecular and energy models and the dynamics and optimization algorithms. This literature review is organized as follows: Sections 2.1 and 2.2 discuss work on models, namely energy and reduced molecular models respectively. Sections 2.3 and 2.4 discuss work on conformation prediction methods, namely molecular dynamics and optimizations respectively. Then Section 2.5 discusses work on energy landscaping that improves optimization, and lastly Section 2.6 concludes the literature review.

2.1 Energy Models

In addition to the molecular models, a protein simulation requires an energy model that describes potential energies between atoms. These energies are important for evaluating a protein configuration in the protein prediction process. Moreover, atomic forces can be determined from the negative of energy gradients.

Two basic energy modeling approaches are physics based and knowledge based [65]. While the physics-based approaches are governed by the laws of physics, the knowledge-based approaches use potential energies with sequence-independent and sequence-specific terms that are empirically derived from the protein structures in the PDB library. In protein conformation prediction the knowledge-based approaches are more successful than the physics-based approaches [65] as they can obtain a configuration closer to the native state; however, the resulting configurations are not necessary the native conformation. Research on energy-landscape sculpting and an improved structure selection has produced encouraging results, but further improvements are still needed [65].

2.1.1 Force Field Model

Molecular chemists have modeled intra-molecular reactions with force field models, which govern the dynamic motion of the folding process in order to calculate the potential energy and internal forces of the protein. Expressing the potential energies of a protein molecule as functions of the molecular structure, the force field model reduces the computational costs over a quantum mechanics approach but is less accurate [44]. In attempts to describe the quantum mechanics of a molecule, parameterized functions in the force field model are empirically found. Although the force field model can sometimes yield results that are as accurate as those obtained from quantum mechanics, it cannot produce accurate values for properties affected by electronic distribution in the molecule, such as electrostatic energy.

Typically force field models fall into two types, all-atom and coarse-grained models. The all-atom force field model describes potential energies that include bond, bending (bond angle), torsion, van der Waals, and electrostatic energies. Since at least one of these energies occurs between every pair of atoms, this model is used with a molecular model that explicitly describes the positions of most atoms. In contrast, when the molecular model has several atoms lumped into larger elements and when atom positions are not explicit, a coarse-grained force field model is used. This model describes the potential energies between the larger elements and these terms are different from the energy terms in the all-atom force field model. Moreover, parameters in individual functions describing each potential energy in one all-atom force field model are not transferable to another force field model because the parameterization of all potential energies in a model are dependent [60].

The most widely accepted all-atom force field models are those of Charmm [12] and Amber [16] which are frequently used as bases for other advanced models. Improving the energy calculation for large molecules, a modified force field [20] reduces the number of atoms by replacing aliphatic and aromatic CH3, CH2, and CH groups with several types of single, united atoms. Similarly, the new-generation Amber unitedatom force field model [77] reduces the force calculations in dynamic simulations by assuming that the positions of hydrogen atoms bonded to an aliphatic carbon atom are the same as the position of the carbon atom. Further improvement can be obtained using the coarse-grained molecular model, which reduces the degrees of freedom of a protein more than the united-atom model. However, the coarse-grained force field
model for one type of element is not transferable to another. While all-atom force field models can be used with any molecular models in which explicit atom positions can be found, the coarse-grained force field models do not have the same flexibility.

2.2 Reduced Molecular Models

2.2.1 HP Model

To reduce the complexity of the protein-analysis problem, researchers generally use a reduced protein model. For example, sequence identification performed by searching in a sequence space is a hard combinatorial optimization problem because its sequence space has 20^N possible sequences for a protein with N amino acids. Therefore, most studies of this problem have been done with smaller search spaces using HP and lattice models. The HP model reduces the sequential space by classifying amino acids into only two groups, hydrophobic (H) and polar (P); a residue-level, self-avoiding, compact lattice model reduces the conformation space by allowing a residue to have at most three interactions with other residues for a 2D model and five interactions for a 3D model [43]. The force model also becomes much simpler. However, even with these models the combinatorial optimization search is difficult. Koh *et al.* [41, 42]avoided this difficulty by defining a continuous state energy function of the simplified protein so that a continuous optimization algorithm could be used. Besides sequence identification, HP and lattice models are widely used in a conformation-identification problem, such as protein folding [78]. However, they are not applicable to tertiary structure identification because of their coarse resolution.

2.2.2 Residue-Level Models

The residue-level model is based on an assumption that the peptide planes are rigid. Modeling a protein as a series of rigid bodies connected by revolute joints makes it possible to use the kinematic model of a serial manipulator that is used in robotics [36, 81]. For example, Kazerounian *et al.* explicitly described the kinematics of the protein, including both a main chain and side chains [36], using the zero-reference position notation [25, 26] which defines joint angles from a user-defined zero position. This notation supposedly requires less computation than other notations. They have proposed improved intrinsic molecular parameters, such as bond angles and lengths, which are more suitable for the residue level rigid-chained protein model [69]. They have also developed an inverse kinematics algorithm to calculate dihedral angles [68]. Their algorithm is compared with other algorithms by deriving atom coordinates from the forward kinematics of their rigid-chain molecular model. However, since other methods do not use the same molecular model to derive dihedral angles, this comparison may be biased. Using one form of Denavit-Hartenberg parameter notation in robotics and modeling larger atom clusters as rigid bodies, Zhang and Kavraki [81] developed a method to derive a molecular conformation. Although this method is supposedly fast, its advantage over the others is not clear because the study did not directly compare the results using the same large clustered atom model.

Besides a reduction in the degrees of freedom, the residue-level model also reduces computational cost by simplifying the force field models [36]. Because of the rigidity assumption, bond lengths, bending angles, and torsion angles are assumed to be unchanged. The Amber force field terms are reduced to van der Waals and electrostatic energies because other energies are constant and they do not affect atomic forces. The mathematical formulation of the Amber force field [34] is

$$E_{ij} = \sum_{vdW} \left(\frac{A_{ij}}{R_{ij}^{12}} - \frac{B_{ij}}{R_{ij}^{6}} \right) + \sum_{elect} \left(\frac{q_i q_j}{\epsilon R_{ij}} \right)$$
(2.1)

where E_{ij} is the potential energy between atoms *i* and *j*, *A* and *B* are experimentally determined constants, R_{ij} is the distance between the two atoms, q_i and q_j are the charges on each atom, and ϵ is the dielectric constant.

2.2.3 Macro-Level Models and Rigidity Analysis

While residue-level models greatly reduce the degrees of freedom of a protein chain, macro-level models have been developed to further reduce the degrees of freedom of a known protein. This reduction is done by identifying the rigid parts of the protein chain using a rigidity or flexibility analysis. A simple method for determining rigid clusters is to compare the different conformations of the same protein in the PDB library. A rigid cluster is a region where the dihedral angles are the same in all conformations. Using this method and a rigid-body-based model, Kim *et al.* [39] generated a feasible path between two conformations. A more sophisticated rigidity analysis uses a graph theoretical technique. Analyzing a molecule modeled as a bar-joint network, Thorpe et al. [30, 71] used the "pebble game" to identify the over- or under-constrained regions of a network that correspond to rigid and flexible regions of the molecule. They have also developed the computer program FIRST (Floppy Inclusion and Rigid Substructure Topography) [29], which identifies rigid clusters and flexible substructures in macromolecules. After the rigid regions (called "ghost templates") are defined their new computational method FRODA (Framework Rigidity Optimized Dynamic Algorithm) [75] can be used to dynamically determine the protein motions. Although the rigidity analysis is a great tool to reduce the computational time in simulations it requires information from the PDB.

2.3 Molecular Dynamics (MD)

2.3.1 Traditional Molecular Dynamics

Two common *ab initio* techniques for the protein-folding problem are the molecular dynamics (MD) simulation and the optimization search. Although in theory MD simulations can be used to locate the native state, they are typically used to study detailed kinetics or dynamics of the protein folding mechanism [14]. The main limitation of MD simulations for protein folding is the required computational power [7, 14]. Since a typical MD simulation is done iteratively by computing the atomic forces on each atom, the acceleration, and then the displacement numerous calculations are required for each iteration, even for a small protein. Moreover, another computational problem is caused by the requirements of a small time step and a lengthy simulation. Since the force functions are highly non-linear the time step of MD simulation must be small, typically in fs for the constant-acceleration approximation between increments to be accurate [60]. Because the protein-folding process usually takes a relatively long time, typically in ms or s, the simulation of protein folding is nearly impossible. Because of these limitations, most MD simulations of the folding process are implemented in small fragments of protein molecules [14] and the MD simulations of full-size proteins are generally used in studies of protein requires significant computational time, it can not be done in a reasonable amount of time with current computational power.

2.3.2 Internal Coordinates

By introducing internal coordinates, or configuration variables, many robotic dynamics algorithms can be applied in order to reduce the computational cost in molecular dynamics computations. As bond angles and lengths are fixed, the cost reduction is due to the decreasing number of the degrees of freedom of a protein chain. Also efficient forward dynamics algorithms [23, 24, 31, 53, 55] are already developed in robotics. For example, a traditional inertia-matrix algorithm (IMA) solves an equation of motion, $H\ddot{q} = \tau - C$, where H is a joint-space inertia matrix (JSIM), \ddot{q} is a joint acceleration vector, τ is a joint force vector, and C is a bias vector containing velocity related terms. While an efficient method to calculate H, such as a composite-rigidbody algorithm (CRBA), is available the IMA still has a computational complexity of $O(N^3)$, of which N is the degrees of freedom of the protein molecule. A complexity of only O(N) can be achieved with an articulated-body algorithm (ABA) which describes an applied force as a linear function of an acceleration for an unconstrained or floating chain of bodies. As a result, the CRBA becomes less efficient than the ABA, when N > 9 [21, 24]. Another JSIM calculation in the IMA that is more efficient than CRBA is a sparse factorization algorithm [23]. This exploits the sparsity of the JSIM of a branched kinematic tree such as a protein structure and the fact that sparsity and symmetry are preserved. Although the reduced complexity is between O(N) and $O(N^3)$, the IMA is still less efficient than the ABA because the ABA can be applied to a branched chain structure without increasing complexity. Another drawback of the IMA is that the JSIM becomes ill-conditioned as N increases [22].

Jain et al. [31] suggested a fast recursive IMA for molecular dynamics simulations called "Newton-Euler Inverse Mass Operator (NEIMO) dynamics." Assuming that the base of the molecule has six degrees of freedom, which is equivalent to floating, NEIMO can be proved equivalent to the ABA using spatial operators [24, 31], which has only O(N) complexity. In [47], NEIMO is implemented in several polypeptides along with the cell multipole method (CMM). The CMM improves the computational time of non-bonded interactions. By dividing the molecule into small cells, the algorithm evaluates the non-bonded interactions only if the atoms are in the same or adjacent cells; otherwise the interactions are estimated. The CMM is more accurate than the cut-off method, in which the non-bonded interactions of distant atoms are ignored. Both NEIMO and CMM reduce the computational time and increase the simulation time step, but not significantly enough to solve the protein-folding problem in large proteins. Since proteins and their solvents in typical chemistry experiments are in contact with a constant-temperature heat bath, [72] presents constanttemperature constrained molecular dynamics simulated by solving NEIMO-Nosé and NEIMO-Hoover equations. In [8], NEIMO is used to study the α -helix formation, which provides some useful insight in protein folding. With a rigid-link method identical to NEIMO, Rapaport [53] also folds some α -helices. However, in addition to a high-energy initial configuration, the simulated annealing technique [40] is used as the temperature of the simulated system is reduced by velocity scaling. The simulated annealing technique is explained in Section 2.4.2. Although this algorithm results in a higher success rate of 85% compared to NEIMO, the torsional potential derived from the native configuration of an α -helix may be a main contribution.

2.3.3 Normal Mode Analysis

Since the force field model requires a large number of computations, a normal mode analysis (NMA) has been widely used to approximate interactions between atoms or residues as the dynamics of a molecule can be represented by the sum of vibration modes. The NMA assumes that point masses are restrained to C_{α} atoms at which the side chains are located, and it models a protein molecule as a cluster of rigid bodies connected by springs but not dampers. Although the real motion of the protein is highly damped, the NMA has been widely used. Alexandrov *et al.* [2] have statistically analyzed the applicability of the NMA to protein-conformation prediction and found little correlation between a single normal-mode or lowest-frequency vibration and protein motion.

One type of the NMA is the elastic-network model in which all springs have equal stiffness and only bodies within a cutoff distance are considered connected. The work of Kim *et al.* [38] on C-alpha coarse-grained elastic network interpolation shows a feasible path between two conformations. Changes between intermediate conformations are chosen such that potential energy is minimized. An extension of this work, rigid-cluster coarse-grained elastic network interpolation (ENI) [39], uses a macro-level protein model whose clusters are formed by rigidly-attached point masses. Since the number of rigid bodies in the protein chain decreases and because only six linear springs on each cluster are necessary to represent rigid-body motions, the number of interactions significantly decreases. The hybrid ENI has rigid clusters for the rigid parts of the chain and point masses on C_{α} atoms for fairly flexible parts. Although various NMA models may prove useful in the simplification of the atomic interactions they are not applicable to protein folding [2].

2.4 Optimizations

2.4.1 Classical Optimization Methods

Since the MD simulation cannot solve the protein-folding problem with currently available computation resources, an optimization algorithm that searches for a minimum energy conformation is frequently used [60]. However, the performance of an optimization algorithm greatly depends on the formulation details of the problem, such as potential functions, the protein model, and the type of a protein, so a performance comparison between algorithms is sometimes difficult. Although numerous algorithms were identified for the protein-folding problem, all are variations of an analytical or statistical optimization algorithm, or a combination of both.

Examples of analytical optimization algorithms are a gradient method called "the method of steepest descent (SD)," Newton's method, and Newton-based methods. While the SD is simple to implement and not computationally expensive its convergence is slow. Newton's method provides a quadratic convergence rate which is superior to the linear rate found in the SD method [18, 49]. However, the second derivative of the potential energy can be computationally expensive for the all-atom model and in many cases it may not be available [60]. Both SD and Newton's method are more robust when a one-dimensional optimization called a "line search" (described below) is implemented, but the computation time can significantly increase. A line search is not necessary in some methods such as the quasi-Newton (QN), the non-linear conjugate gradient (CG), the truncated Newton (TN), and the traditional SD

methods because they are designed to provide a proper step size. QN methods such as the Davidon-Fletcher-Powell (DFP) and the Broyden-Fletcher-Goldfarb-Shanno (BFGS) methods, except for the limited-memory type, require extensive memory storage. This problem does not occur in the CG and TN methods; however, the CG method converges more slowly than the QN method and the implementation of the TN method is more complex. Compared with the other methods mentioned above Newton's method is the most reliable with the fastest convergence despite its expensive computation.

2.4.1.1 Line Search

After a descent direction in which the cost function locally decreases is determined by the SD, the Newton's, or the QN methods, a line search along this direction determines a scalar multiplier here called a "line search gain" that locally minimizes the cost function [6, 60]. An example of a very simple line search is a bracketing algorithm [6] that advances the search step with a varying or a constant interval in monotonically decreasing direction until the function no longer decreases. Unlike the bracketing algorithm that expands the search interval, interval reduction methods, such as the Fibonacci and the golden section methods, reduce the search interval until a stopping criterion is met. Requiring a minimum inside an initial search interval, these algorithms ensure that reduced intervals contain a minimum as well. The Fibonacci and the golden section methods reduce the interval such that a ratio between consecutive intervals equal to the ratio between two consecutive Fibonacci numbers or the golden ratio, respectively. A more robust algorithm is an integration of a bracketing algorithm and the golden section method. With a bracketing technique this integrated algorithm identifies an initial interval by using the golden ratio to determine an advancing step and the interval is reduced by the golden section method. Because of a slow convergence this algorithm requires a large number of function evaluations and can be computationally expensive. To improve efficiency it is combined with a polynomial fit-sectioning algorithm. For example, Brent's method performs quadratic fitting on a bracketed interval and determines a minimum from the obtained quadratic equation when applicable and it uses golden section method otherwise. Less frequently used than a quadratic fit, a cubic and a higher polynomial fits require more points or derivative information so they are inefficient if not inapplicable when the derivative along the descent direction is unavailable.

2.4.2 Monte Carlo

One of the most popular statistical optimization algorithms used in the proteinfolding problem is the Monte Carlo (MC) algorithm [44]. MC simulation generates statistical ensembles (or configurations) by adding random changes to the current position. These changes are between the maximum displacements $\pm \delta r_{max}$ and the acceptance of the new configuration is determined by the Metropolis criterion [48]. If the new configuration has a lower potential energy it will be accepted as a new current position. If it has a higher potential energy it will be rejected except when the Boltzmann factor is greater than or equal to a random number between 0 and 1. The Boltzmann factor is defined as $exp(-\Delta E/k_BT)$, where ΔE is the energy difference, k_B is the Boltzmann constant, and T is the temperature. This exception allows the higher-energy ensembles to be accepted at some probability. The smaller the value of ΔE , the higher the probability of acceptance. As the Boltzmann factor determines the probability of an ensemble, the δr_{max} controls the rate of change in the ensemble generation. When δr_{max} is small, the simulation will generate many acceptable ensembles but they may not be far away from the initial position. When δr_{max} is large, the simulation will generate very different ensembles but most of them may be rejected. Therefore, the difficulty of the MC algorithm is finding an efficient value of δr_{max} for each protein.

As most generated ensembles will be rejected, especially for a large protein, the MC algorithm can be inefficient. Therefore, a combination of MC and other techniques is generally used in practice. For example, a hybrid MC algorithm [60] uses MD to bias the ensemble generation and the MC criterion to accept or reject the new ensemble. Another example [44] is MC algorithm with simulated-annealing technique [40], inspired by manufacturing process that strengthens materials. The simulated-annealing process for conformation prediction starts at a high temperature and ends at absolute zero. While also applicable to MD and MC with other algorithms, this technique uses MC algorithm to find a thermal equilibrium configuration at each temperature. Once the temperature drops to the absolute zero, the molecule should be at the global minimum energy conformation. However, this conformation is not guaranteed because at each temperature step thermal equilibrium must be achieved and an infinite number of steps are required. Therefore, in practice several different runs are done to increase the probability of finding the global minimum energy conformation.

2.4.3 De Novo Methods

Compared to *ab initio* techniques for solving the protein-folding problem *de Novo*, or knowledge-based methods, that use information from the PDB have better performance. Here two representative examples are discussed.

2.4.3.1 TASSER

Zhang and Skolnick [84, 83, 85] reduce the degrees of freedom of a protein by structure alignment that utilizes the PDB to identify templates or secondary structures of a protein. A part of an unknown protein with a 35% sequence identity to a secondary structure of a known protein in the PDB yields a template which assumes the shape of that structure. Then several of these templates construct an initial template alignment that was used as an initial structure of a conformation search done by MC sampling. The internal configurations of the templates are kept constant while the configuration of the residues in the gapped region (between templates) is generated using the *ab initio* lattice modeling approach. A representative benchmark study [85] on 1,489 single-domain, medium-size proteins in the PDB with 41 to 200 amino acids shows that a template with the root-mean-square deviation (RMSD) from the native state of 2.5 Å can always be found. While the average RMSD value of the full-length models is 2.25 Å, 97% of them have RMSD values below 4 Å. Another benchmark study [84] is on 745 medium-to-large-size proteins with 201 to 300 residues. The results show that the templates of 365 cases have an RMSD value of 6.5 Å or less with 70% alignment coverage. Furthermore, 408 of the full-length models have the RMSD values lower than 6.5 Å.

2.4.3.2 Rosetta

Rosetta [11, 56] predicts the protein conformation using a *de Novo* method that constructs several configurations called "decoys" from pre-generated fragments and then a refinement protocol optimizes the predicted conformation. As an energy function, Rosetta uses a potential energy surface (PES) called "scoring function" which is derived based on a Bayesian separation of the total energy [63, 64]. Using the Monte Carlo-minimization approach [46], also called "basin-hopping" [73], the refinement protocol applies a series of random changes and minimization to each decoy. More details about this software and its method are discussed in Chapter 5. Bradley *et al.* [11] achieves high-resolution structure prediction for sixteen small proteins which have less than 85 residues. Five proteins have C_{α} -RMSD less than 1.5 Å and eight proteins have 1.5 to 5.0 Å but no native side-chain packing. Since 20,000 to 30,000 generated decoys are necessary for each protein, this process is very computationally expensive, as it takes about 150 CPU days per molecule. The improvement in the efficiency of the refinement protocol can significantly impact the performance of the method and reduce the computational time.

2.4.4 Non-Traditional Optimization

Besides the family of Newton's methods and MC algorithms, many studies have used non-traditional approaches. For example, Yoon [78] applied integer programming and constraint programming models to protein folding using HP and lattice models. Although his models are not fast enough to be used in practice, they showed superiority over comparison algorithms. Another exciting approach is a control approach even though it has not been widely explored. Most control approaches applied to protein folding are generic or evolutionary algorithms, as in [19], but a study of protein folding using other control techniques, such as classical and adaptive controls, is rare. Canutescu and Dunbrack [15] use an interesting control approach in "protein loop closure" which determines if a structure can geometrically close a protein loop. Fixing one end of the structure to one end of the protein and changing the dihedral angles of the structure one at a time, their cyclic coordinate descent (CCD) algorithm verified if another end of the structure could possibly meet another end of the protein. Since CCD algorithms check only the geometric feasibility of the structure but not the energetic compatibility between the combined compounds, it is not applicable to the protein-folding problem.

2.5 Energy Landscape Modification

The complexity of the energy function results in numerous local minima, at which the optimization schemes get trapped. Addressing this challenge, some studies work on the modification of the energy landscape so that it assists the minimization algorithm.

2.5.1 Hypersurface Deformation

An approach of energy landscape modification, called "hypersurface deformation", has been investigated for at least three decades [4, 5, 13, 50, 51, 59, 66, 74]. In general, these algorithms smoothen the landscape by reducing the local minima, which aids the global minimum search. However, the obtained global minimum has to be mapped back to the original surface, which can be troublesome and unreliable [73].

In a global optimization scheme, Azmi *et al.* [4, 5] use a non-integration based spatial averaging to smooth the non-bonded interactions of the potential energy function removing local minima while maintaining the basic structure of the energy landscape. However, the smoothing method alters the locations of the minima, so several desmoothing steps that perform a local minimization on the original energy landscape are required. Although this optimization scheme successfully predicts the conformation of a 70-residue helical protein, the smoothing method is not applicable to other energy functions but only to Lennard-Jones and electrostatic energy functions.

Based on an idea that the energy landscape is a funnel with bumps, the Convex Global Underestimator (CGU) global optimization algorithm [50] iteratively forms a least-underestimated separable-quadratic surface under minima found by sampling the energy landscape. After finding the global minimum on the obtained surface, it locates the corresponding local minimum on the energy landscape. Then more samplings are focused to an area near the minimum of the surface and the surface is regenerated. The global minimum is found if it is equal to its corresponding local minimum. The algorithm appears to work well on simple homopolymers and simplified Sun energy functions that are funnel-like. However, the method of determining a successful run is questionable. Since the global minima for most proteins are unknown, a run is considered successful if the predicted global minimum is within 1% of the lowest energy value sampled from the energy landscape. Moreover, the CGU algorithm becomes much less effective on real proteins and more complex energy functions because the underestimated surface does not represent the energy landscape very well [50].

Since most methods, such as in [50], require intensive samplings in order to find the global solution, Burke and Yaliraki [13] use sum of squares (SOS) decomposition to transform a nonconvex function in Euclidean space to a nonnegative convex SOS function in the vector space of monomials. Without sampling the global minimum, or a solution close to the global minimum, can be found from the SOS function which is a much easier minimization problem. Global minimization on several potentials with similar shapes to a potential energy surface is presented. However, since these potentials are represented only as polynomial equations, this work may not be applicable to the actual energy functions unless their polynomial forms can be approximated.

2.5.2 Energy Landscape Flattening

Encouraging the advancing of the Monte Carlo optimization step, parallel hyperbolic sampling (PHS) flattens the energy landscape by shortening high energy barriers with little effect on lower energy barriers [87]. With the same computation time, the PHS results in lower energy configurations than the replica sampling algorithm [27, 70], which changes the state of the low-energy configuration so that the Monte Carlo step can pass over high-energy barriers. This work, however, does not study a comparison between the flattened and unmodified energy landscape. Also, since the PHS can be used with only the Monte Carlo method, other methods of energy landscape modification are required for other optimization algorithms.

2.6 Conclusion of the Literature Review

Due to the inapplicability of the molecular dynamic simulation, several optimization algorithms including both probabilistic and analytical methods are used in the protein folding problem instead. While analytical methods can locate local solutions very well, by themselves they are not successful in a very complex and nonlinear optimization problem. Therefore, the ability to search a global space with probabilistic methods is a very useful addition to the analytical methods. Hence, the most successful algorithms, *de Novo* algorithms, perform exceptionally well because they use the knowledge obtained from the protein database and combine both probabilistic and analytical methods.

Although the *de Novo* methods can be enhanced by improving the efficiency of either the probabilistic methods or the analytical methods, improvements in the latter methods are more significant because of two following reasons: 1) they are the bottleneck of the *de Novo* methods as they take a longer time, 2) they involve in the majority of the process as they are intensively used in refinement protocols which repeatedly perform local minimization. Reducing the computational time required for the analytical algorithms will significantly improve the overall processing time of the *de Novo* methods.

The literature review shows that the energy landscape modification is a promising method that can be used to improve the efficiency of the analytical algorithm. Yet the method has never been thoroughly investigated. The hypersurface deformation method suggests that the energy landscape modification is beneficial but altering the minima's locations requires other troublesome tasks. Without the alteration, the energy landscape flattening provides an improvement to the Monte Carlo method but not analytical algorithms. In most research on the energy landscape modification a comprehensive study is needed to fully understand its effect.

In conclusion, the objective of this research is to develop a method of energy landscaping that enhances the efficiency of the analytical algorithms for the protein prediction problem. Since Newton's method provides the basis for several powerful analytical methods that are commonly used in conformation prediction, it is chosen to be investigated along with two quasi-Newton methods. A comparative study of the effect of an energy landscaping method on several variations of these algorithms is performed.

CHAPTER III

NEWTON'S METHOD AND QUASI-NEWTON METHODS APPLIED TO PROTEIN FOLDING PROBLEM

This chapter discusses basic concepts of Newtonian optimization and its application to the protein folding problem. Sections 3.1 and 3.2 discuss the derivation of Newton's method and two quasi-Newton methods for an optimization problem. Section 3.3 discusses how these methods are applied to the protein folding problem. Lastly, Section 3.4 summarizes and concludes the chapter.

3.1 Newton's Method

Newton's method is often used to locate the root of a function, but for minimization problems it is used to locate the roots of the derivatives of the cost function f giving a stationary value of the cost function. Newton's method can be derived from the Taylor series expansion of the cost function derivative,

$$\frac{\partial f(\theta_k + h)}{\partial \theta} = \frac{\partial f(\theta_k)}{\partial \theta} + \frac{\partial^2 f(\theta_k)}{\partial \theta^2} h + \frac{1}{2} h^T \frac{\partial^3 f(\theta_k)}{\partial \theta^3} h + \dots$$

where $h \equiv \Delta \theta$ is a small change in the vector variable θ . Dropping the third- and higher-order terms and setting $\frac{\partial f(\theta_k + h)}{\partial \theta} = 0$ gives

$$0 = \frac{\partial f(\theta_k)}{\partial \theta} + \frac{\partial^2 f(\theta_k)}{\partial \theta^2} \hat{h}$$

where \hat{h} is an approximation of h. Rearranging gives

$$\hat{h}_k = -K_k^{-1} \frac{\partial f(\theta_k)}{\partial \theta} \tag{3.1}$$

where \hat{h}_k is the k^{th} descent direction of the Newton's method, here called "Newton step," and $K_k \equiv \frac{\partial^2 f(\theta_k)}{\partial \theta^2}$ is the Hessian matrix. The $(k+1)^{\text{th}}$ approximated solution becomes

$$\theta_{k+1} = \theta_k + \hat{h}_k$$

Newton's method fundamentally gives a solution to the quadratic approximation of a function [18]. Therefore, it can achieve a quadratic convergence rate superior to the linear rate found in the gradient methods, such as the steepest descent method [18, 49]. Generally, the Newton's method can locate a solution only if it is close to the starting point. Also if the Hessian matrix is not positive definite or if it is singular the Newton step may worsen the optimization or the Newton step may not exist [6]. To avoid divergence, Newton's method is commonly implemented with a line search [6, 49].

3.1.1 Line Search

Usually implemented with Newton-based algorithms, a line search determines a line search gain or a scalar multiplier of the step.

$$\alpha_k = \texttt{line_search}(\hat{h}_k)$$

where α_k is the gain that minimizes $f(\theta_k + \alpha \hat{h}_k)$. The revised $(k+1)^{\text{th}}$ approximated solution becomes

$$\theta_{k+1} = \theta_k + \alpha_k \dot{h}_k \tag{3.2}$$

where $\alpha_k \hat{h}_k$ is the total step.

It is impractical to globally minimize $f(\theta_k + \alpha \hat{h}_k)$ so the gain only gives a solution on a limited interval. Many robust line search algorithms define an initial value of α equal to 1 so the magnitude of the Newton step $|\hat{h}_k|$ sets the size of the initial search interval which means different values of $|\hat{h}_k|$ yield different searched locations. Since the gain only locally minimizes f in the direction of the Newton step, different Initialize ϵ , θ_0 , and θ_1 Calculate $f(\theta_0)$ for $k = 1, ..., k_{max}$ do Calculate $f(\theta_k)$, $\frac{\partial f(\theta_k)}{\partial \theta}$, and $\frac{\partial^2 f(\theta_k)}{\partial \theta^2}$ if $\left| \frac{f(\theta_k) - f(\theta_{k-1})}{f(\theta_{k-1})} \right| < \epsilon$ then break // Convergence criterion met. end if $\hat{h}_k = -\left(\frac{\partial^2 f(\theta_k)}{\partial \theta^2}\right)^{-1} \frac{\partial f(\theta_k)}{\partial \theta}$ // Determine the Newton step $\alpha_k = \texttt{line_search}(\hat{h}_k)$ // Determine the line search gain $\theta_{k+1} = \theta_k + \alpha_k \hat{h}_k$ // $(k+1)^{\text{th}}$ solution end for

Figure 3.1: A pseudo-code for Newton's method with a line search

searched locations can result in different gains as they correspond to different local minima. Figure 3.1 shows a pseudo-code for the Newton's method with a line search.

3.2 Quasi-Newton (QN) Methods

Since the calculation of the Hessian matrix K can be computationally expensive or is often unavailable, quasi-Newton (QN) methods use only the first derivative to estimate K by satisfying the secant equation,

$$\frac{\partial f(\theta_{k+1})}{\partial \theta} - \frac{\partial f(\theta_k)}{\partial \theta} = \hat{K}\hat{h}$$
(3.3)

where \hat{K} is the estimated Hessian matrix. Equation (3.3) is underdetermined so many methods apply other criteria to find the solution. Minimizing the Frobenius norm $||K_{k+1} - K_k||_{\mathbf{F}}$, Broyden's method [17] recursively estimates K,

$$\hat{K}_{k} = \hat{K}_{k-1} - \left(\hat{h}_{k-1}^{T}\hat{h}_{k-1}\right)^{-1} \left(\gamma_{k-1} + \hat{K}_{k-1}\hat{h}_{k-1}\right)\hat{h}_{k-1}^{T}$$
(3.4)

where

$$\gamma_{k-1} \equiv \frac{\partial f(\theta_k)}{\partial \theta} - \frac{\partial f(\theta_{k-1})}{\partial \theta}$$

is the change in the gradient of the cost function f. The QN step becomes

$$\hat{h}_k = -\hat{K}_k^{-1} \frac{\partial f(\theta_k)}{\partial \theta}$$
$$\theta_{k+1} = \theta_k + \alpha_k \hat{h}_k$$

Figure 3.2 displays a pseudo-code for a QN method with a line search that uses Broyden's method to estimate K.

While Broyden's method works fairly well, it does not guarantee a symmetric and positive definite \hat{K} [17]. Since the Hessian matrix is alway symmetric and often positive definite, the Broyden-Fletcher-Goldfarb-Shanno (BFGS) algorithm ensures these properties as it estimates K^{-1} by satisfying the quasi-Newton condition [6]

$$\hat{H}_k \gamma_k = \hat{h}$$

where \hat{H} is an estimation of K^{-1} . Widely used in an nonlinear optimization, the BFGS algorithm is one of the most powerful analytical optimization algorithms when used with a line search. The BFGS algorithm recursively calculates \hat{H} as,

$$\hat{H}_{k} = \hat{H}_{k-1} - \left(\frac{\hat{H}_{k-1}\gamma_{k-1}\hat{h}_{k-1}^{T} + \hat{h}_{k-1}\gamma_{k-1}^{T}\hat{H}_{k-1}}{\hat{h}_{k-1}^{T}\gamma_{k-1}}\right) + \left(1 + \frac{\gamma_{k-1}^{T}\hat{H}_{k-1}\gamma_{k-1}}{\hat{h}_{k-1}^{T}\gamma_{k-1}}\right)\frac{\hat{h}_{k-1}\hat{h}_{k-1}^{T}}{\hat{h}_{k-1}^{T}\gamma_{k-1}}$$

The k^{th} BFGS step \hat{h}_k is defined as

$$\hat{h}_k = -\hat{H}_k \frac{\partial f(\theta_k)}{\partial \theta}$$
$$\theta_{k+1} = \theta_k + \alpha_k \hat{h}_k$$

Initializing \hat{H}_0 to be positive definite, BFGS algorithm guarantees positive definite \hat{H} 's. However, its direction can not guarantee to lower a function value because the

actual Hessian matrix may not be positive definite. Another drawback is that the QN methods may be slow to converge. The Hessian matrix is usually not constant and if the magnitude of the QN step \hat{h}_k is too large \hat{K} may fail to converge. Because of these drawbacks QN methods are usually implemented with a line search for best results.

Figure 3.3 gives a pseudo-code for the QN method and uses the BFGS algorithm

Pseudo-code: QN method with a line search and Broyden's method

Initialize ϵ , θ_0 , θ_1 , and \hat{K}_0 **Calculate** $f(\theta_0)$ and $\frac{\partial f(\theta_0)}{\partial \theta}$ **for** $k = 1, ..., k_{max}$ **do Calculate** $f(\theta_k)$ and $\frac{\partial f(\theta_k)}{\partial \theta}$ **if** $\left| \frac{f(\theta_k) - f(\theta_{k-1})}{f(\theta_{k-1})} \right| < \epsilon$ **then**

break // Convergence criterion met.

end if

Calculate \hat{K}_k // Broyden's method

$$\begin{split} \hat{h}_{k-1} &= \theta_k - \theta_{k-1} \\ \gamma_{k-1} &= \frac{\partial f(\theta_k)}{\partial \theta} - \frac{\partial f(\theta_{k-1})}{\partial \theta} \\ \hat{K}_k &= \hat{K}_{k-1} - \left(\hat{h}_{k-1}^T \hat{h}_{k-1}\right)^{-1} \left(\gamma_{k-1} + \hat{K}_{k-1} \hat{h}_{k-1}\right) \hat{h}_{k-1}^T \\ \hat{h}_k &= -\hat{K}_k^{-1} \frac{\partial f(\theta_k)}{\partial \theta} / / \text{ Determine the QN step} \\ \alpha_k &= \texttt{line_search}(\hat{h}_k) / / \text{ Determine the line search gain} \end{split}$$

 $\theta_{k+1} = \theta_k + \alpha_k \hat{h}_k / / (k+1)^{\text{th}}$ solution

end for

Figure 3.2: A pseudo-code for the QN method with a line search and Broyden's method

Pseudo-code: BFGS algorithm with a line search

Initialize ϵ , n, θ_0 , θ_1 , and \hat{H}_0 Calculate $f(\theta_0)$ and $\frac{\partial f(\theta_0)}{\partial \theta}$ for $k = 1, ..., k_{max}$ do Calculate $f(\theta_k)$, and $\frac{\partial f(\theta_k)}{\partial \theta}$ if $\left| \frac{f(\theta_k) - f(\theta_{k-1})}{f(\theta_{k-1})} \right| < \epsilon$ then

break // Convergence criterion met.

end if

Calculate \hat{H}_k

$$\begin{aligned} h_{k-1} &= \theta_k - \theta_{k-1} \\ \gamma_{k-1} &= \frac{\partial f(\theta_k)}{\partial \theta} - \frac{\partial f(\theta_{k-1})}{\partial \theta} \\ \hat{H}_k &= \hat{H}_{k-1} - \left(\frac{\hat{H}_{k-1}\gamma_{k-1}\hat{h}_{k-1}^T + \hat{h}_{k-1}\gamma_{k-1}^T \hat{H}_{k-1}}{\hat{h}_{k-1}^T \gamma_{k-1}} \right) \\ &+ \left(1 + \frac{\gamma_{k-1}^T \hat{H}_{k-1} \gamma_{k-1}}{\hat{h}_{k-1}^T \gamma_{k-1}} \right) \frac{\hat{h}_{k-1}\hat{h}_{k-1}^T}{\hat{h}_{k-1}^T \gamma_{k-1}} \end{aligned}$$

 $\hat{h}_k = -\hat{H}_k \frac{\partial f(\theta_k)}{\partial \theta} // \text{ Determine the BFGS step}$ $\alpha_k = \texttt{line_search}(\hat{h}_k) // \text{ Determine the line search gain}$ $\theta_{k+1} = \theta_k + \alpha_k \hat{h}_k // (k+1)^{\text{th}} \text{ solution}$

end for

Figure 3.3: A pseudo-code for BFGS algorithm with a line search

to estimate K^{-1} with a line search.

3.3 Application to the Protein Folding Problem

When optimization algorithms are applied to the protein folding problem, the cost function is the molecular energy function or force field, sometimes called the "score" or "score function."

3.3.1 Energy and Derivatives Evaluation

In addition to van der Waals and electrostatic energies described in Equation (2.1), a force field for protein conformation prediction could include parameterized potentials that encourage a formation of secondary structures. These energy terms are of functions of different variables from molecular geometry such as the distance between atoms and molecular characteristics such as the amino acid type. Optimization with respect to the molecular geometry variables is a difficult constrained minimization problem so it usually done with respect to the dihedral angles and/or the Rotamer angles also called internal coordinates which is a non-constrained minimization problem. A closed form of the energy function in terms of these angles is however unavailable so transforming the angle information into molecular geometry information is needed before the function can be evaluated.

The expression of the first and second derivatives of the energy function with respect to internal coordinates are difficult to derive. Because energy functions are of functions of the distance between atoms, the chain rule is used. The first derivative calculation requires the derivatives of the distance between each atom pairs with respect to internal coordinates $\partial R_{ij}/\partial \theta_k$, where R_{ij} is the distance between i^{th} and j^{th} atoms and θ_k is the k^{th} angle in internal coordinates. These terms are difficult and inefficient to calculate because of two following reasons [45]: the number of atom pairs in a molecule are almost the square of the number of atoms and $\partial R_{ij}/\partial \theta_k$ is not zero if angle k is on the part of the chain joining i^{th} and j^{th} atoms. Also, since an average protein chain has several hundreds degrees of freedom, the calculation of the transformation becomes very inefficient. In practice only the first derivative is evaluated with various methods available [60] and the second derivative is estimated. For energies described by molecular geometry such as the van der Waals and electrostatic energies, the first derivative of the energy (2.1) with respect to the distance R_{ij} between two atoms gives the attractive force F_{ij} acting on each atom [44]. The derivative of the energy with respect to each dihedral angle or Rotamer angle is the resulting torque with the same direction as the rotation vector of the angles contributed by these atomic forces. This is the sum of the dot product of the force and the resulting vector of a cross product between the rotational unit vector of the angle and the vector from the rotational vector to the force vector. The expression of the derivative of other energy terms are also available [44]. In this work the energy functions are from known simulated energy surfaces with known derivatives and from the Rosetta software package that also provides the first derivatives but not the second derivative.

The minimization variables of the energy function can be the dihedral angles, the Rotamer angles, or both. For coarse resolution minimization, which is usually done at the beginning of the prediction, the Rotamer angles are fixed at their most stable values and only the dihedral angles are subject to minimization. For full atom refinement, the Rotamer angles and/or the dihedral angles are subject to minimization. From this point forward this distinction will not be made as the variables of the energy function will be the dihedral angles only.

3.3.2 Newton's Method

Applying Newton's method to the molecular potential-energy function $f(\theta) = E$ gives the dihedral angle update vector \hat{h} for energy minimization of a protein molecule. The first and the second order derivatives of the cost function are

$$\frac{\partial f(\theta)}{\partial \theta} \equiv \frac{\partial E(\theta)}{\partial \theta} = -\tau$$
$$\frac{\partial^2 f(\theta)}{\partial \theta^2} \equiv \frac{\partial^2 E(\theta)}{\partial \theta^2} = -\frac{\partial \tau(\theta)}{\partial \theta} = K$$

where θ is the dihedral angle vector, E is the potential energy, the negative gradient of E is the torque vector τ (τ_i is positive in the same direction as θ_i), and the Hessian of E is the stiffness matrix K. The Newton step (3.1) becomes

$$\hat{h}_k = K_k^{-1} \tau_k \tag{3.5}$$

For an analogous one dimensional rotational mass-spring system, this stiffness matrix becomes the stiffness K of a torsion spring at a joint connecting a body to the base, Figure 3.4. Away from the equilibrium state $\theta = 0$, the potential energy Estored in the spring is $E = \frac{1}{2}K\theta^2$ and the torque on the body is $\tau = -K\theta$. If the body rotates in the direction of the torque, opposite to the direction of the displacement angle, the potential energy in the spring decreases. This is analogous to Newton step (3.5) where the joint torques are scaled by the inverse of the stiffness matrix.



Figure 3.4: A simple rotational mass-spring system.

Some molecular energy functions contain non-energy terms that are empirically derived from statistical data in the protein database or from experimental data. Although the first and the second derivatives of these potentials are not torques or stiffnesses and may not have physical meanings, they are torque-like vectors and stiffness-like matrices, respectively. Therefore, the Newton update can still be applied to the score functions by expanding the definition of the derivative terms, τ and K, to include torque- and stiffness-like quantities.

3.3.3 QN Methods

Applied to the energy function E the QN step becomes

$$\hat{h}_k = \hat{K}_k^{-1} \tau_k \tag{3.6}$$

and γ is changed to

$$\gamma_{k-1} = \frac{\partial E(\theta_k)}{\partial \theta} - \frac{\partial E(\theta_{k-1})}{\partial \theta} = -\Delta \tau_{k-1}$$

where $\Delta \tau_{k-1} \equiv \tau_k - \tau_{k-1}$ is the change in the torque vector. so Broyden's method is rewritten as

$$\hat{K}_{k} = \hat{K}_{k-1} + \left(\hat{h}_{k-1}^{T}\hat{h}_{k-1}\right)^{-1} \left(\Delta\tau_{k-1} - \hat{K}_{k-1}\hat{h}_{k-1}\right)\hat{h}_{k-1}^{T}$$
(3.7)

From this point forward the "quasi-Newton algorithm (QNA)" implies QN step for the potential-energy function (3.6) that uses Broyden's method (3.7) to estimate the stiffness matrix.

Applying the BFGS to the energy function E yields

$$\hat{H}_{k} = \hat{H}_{k-1} - \left(\frac{\hat{H}_{k-1}\Delta\tau_{k-1}\hat{h}_{k-1}^{T} + \hat{h}_{k-1}\Delta\tau_{k-1}^{T}\hat{H}_{k-1}}{\hat{h}_{k-1}^{T}\Delta\tau_{k-1}}\right) + \left(\frac{\Delta\tau_{k-1}^{T}\hat{H}_{k-1}\Delta\tau_{k-1}}{\hat{h}_{k-1}^{T}\Delta\tau_{k-1}} - 1\right)\frac{\hat{h}_{k-1}\hat{h}_{k-1}^{T}}{\hat{h}_{k-1}^{T}\Delta\tau_{k-1}}$$
(3.8)

The BFGS step \hat{h}_k is defined as

$$\hat{h}_k = \hat{H}_k \tau_k \tag{3.9}$$

Equation (3.8) updates \hat{H} using the current information of the energy function and the initial \hat{H}_0 is usually defined as an identity matrix.

3.4 Conclusion for Newton's Method and quasi-Newton Methods Applied to Protein Folding Problem

The derivation and the application to the protein folding problem of Newton's method, the QNA, and the BFGS algorithm have been discussed. In protein folding

Pseudo-code: Newton's method with a line search applied to the protein folding problem

Initialize ϵ , θ_0 , and θ_1 Calculate E_0 for $k = 1, ..., k_{max}$ do Calculate E_k , τ_k , and K_k if $\left|\frac{E_k - E_{k-1}}{E_{k-1}}\right| < \epsilon$ then break // Convergence criterion met. end if $\hat{h}_k = K_k^{-1} \tau_k$ // Determine the Newton step $\alpha_k = \texttt{line_search}(\hat{h}_k)$ // Determine the line search gain $\theta_{k+1} = \theta_k + \alpha_k \hat{h}_k$ // $(k + 1)^{\text{th}}$ solution end for

Figure 3.5: A pseudo-code for Newton's method with a line search applied to the protein folding problem

the estimation of the Hessian matrix \hat{K} or its inverse \hat{H} is most likely inaccurate and even the matrix K itself is not always positive definite due to the complex energy landscape. Even though a line search is used to ensure that the optimization moves toward a lower energy, the convergence can still be slow. For faster convergence a method of energy landscaping is presented in Chapter 4. Figures 3.5 thru 3.7 give pseudo-codes summarizing a Newton's method, the QNA, and the BFGS algorithm with a line search applied to the protein folding problem.

Pseudo-code: QNA with a line search applied to the protein folding problem

Initialize ϵ , θ_0 , θ_1 , and \hat{K}_0 Calculate E_0 and τ_0 for $k = 1, ..., k_{max}$ do Calculate E_k and τ_k if $\left|\frac{E_k - E_{k-1}}{E_{k-1}}\right| < \epsilon$ then break // Convergence criterion met. end if Calculate \hat{K}_k $\hat{h}_{k-1} = \theta_k - \theta_{k-1}$ $\Delta \tau_{k-1} = \tau_k - \tau_{k-1}$ $\hat{K}_k = \hat{K}_{k-1} + (\hat{h}_{k-1}^T \hat{h}_{k-1})^{-1} (\Delta \tau_{k-1} - \hat{K}_{k-1} \hat{h}_{k-1}) \hat{h}_{k-1}^T$ $\hat{h}_k = \hat{K}_k^{-1} \tau_k //$ Determine the QNA step

 $\alpha_k = \texttt{line_search}(\hat{h}_k) \; / / \; \text{Determine the line search gain}$

$$\theta_{k+1} = \theta_k + \alpha_k \hat{h}_k / (k+1)^{\text{th}}$$
 solution

end for

Figure 3.6: A pseudo-code for QNA with a line search applied to the protein folding problem

Pseudo-code: BFGS algorithm with a line search applied to the protein folding problem

Initialize ϵ , n, θ_0 , θ_1 , and \hat{H}_0 Calculate E_0 and τ_0 for $k = 1, \dots, k_{max}$ do Calculate E_k , and τ_k if $\left|\frac{E_k - E_{k-1}}{E_{k-1}}\right| < \epsilon$ then break // Convergence criterion met. end if Calculate \hat{H}_k $\hat{h}_{k-1} = \theta_k - \theta_{k-1}$ $\Delta \tau_{k-1} = \tau_k - \tau_{k-1}$ $\hat{H}_k = \hat{H}_{k-1} - \left(\frac{\hat{H}_{k-1}\Delta \tau_{k-1}\hat{h}_{k-1}^T + \hat{h}_{k-1}\Delta \tau_{k-1}^T \hat{H}_{k-1}}{\hat{h}_{k-1}^T \Delta \tau_{k-1}}\right)$ $+ \left(\frac{\Delta \tau_{k-1}^T \hat{H}_{k-1}\Delta \tau_{k-1}}{\hat{h}_{k-1}^T \Delta \tau_{k-1}} - 1\right) \frac{\hat{h}_{k-1}\hat{h}_{k-1}^T}{\hat{h}_{k-1}^T \Delta \tau_{k-1}}$ $\hat{h}_k = \hat{H}_k \tau_k //$ Determine the BFGS step $\alpha_k = 1$ ine_search $(\hat{h}_k) //$ Determine the line search gain $\theta_{k+1} = \theta_k + \alpha_k \hat{h}_k // (k+1)$ th solution

end for

Figure 3.7: A pseudo-code for BFGS algorithm with a line search applied to the protein folding problem

CHAPTER IV

ENERGY LANDSCAPING

This chapter presents a method of energy landscaping called the "exponential energy landscaping (XEL)" method. Unlike the hypersurface deformation discussed in the literature review this method only changes the heights and the depths of the extrema but not their locations. Section 4.1 presents the XEL method and its application to the Newton, the QNA, and the BFGS methods. Section 4.2 details characteristics of XEL algorithms and presents application guidelines. Section 4.3 derives two types of optimization algorithms for an adaptively modified energy landscape (AXEL). Lastly, Section 4.4 summarizes and concludes the chapter.

4.1 Exponential Energy Landscaping (XEL)

4.1.1 Method

With XEL the goal is to transform the energy landscape to facilitate the optimization search without changing the location of its minima and maxima. While these extrema in XEL are at the same locations they have different values. This avoids the troublesome remapping steps seen in hypersurface deformation algorithms. The modified energy function E^* is described by a nonlinear n^{th} -ordered exponential of the energy function E,

$$E^* = sign\left(E\right) \left|E\right|^n \tag{4.1}$$

where n, the exponential order of the energy function, is a positive real number, n > 0. The $|E|^n$ term modifies the magnitude of the energy value and the sign(E) keeps the sign of the modified function the same as the energy function. This definition of the modified energy function is superior to E^n for two reasons. First, when n is noninteger E^* is well defined for all energy values, unlike E^n . For example $E^{0.5}$ is undefined for negative E's. Secondly, E^* always preserves the locations of the extrema because it is always an odd function that preserves the sign of E. Unlike E^* , when n is an even integer E^n becomes an even function and a negative E gives a positive E^n value. This means negative minima and maxima become positive maxima and minima, respectively, and minima appear where sign(E) changes.

Figure 4.1 illustrates XEL with the different values of n. When n < 1 the minima and the maxima are less defined as the energy landscape is flattened. This results in a smoother energy landscape and a larger step, which is shown in Section 4.2.4. A larger step can speed optimization at the beginning when the starting location is far from a minimum. The optimization path is also less likely to get stuck at shallow minima. On the contrary, when n > 1 the minima and the maxima are more defined



Figure 4.1: Comparison of unmodified energy landscape with n = 1, XEL with n = 2, and XEL with n = 0.5. The n = 2 XEL accentuates the extrema while the n = 0.5 XEL flattens the landscape.

as energy landscape is sharpened. This results in the more prominent minimum and the smaller step. This may be preferable toward the end of the optimization to ensure that the deeper minimum can be found.

Note that in the small range of |E| < 1 the effects of n on extrema are opposite to what described above. Figure 4.2 illustrates these effects. For n < 1, $|E|^n > |E|$ which means the energy landscape is sharpened. On the contrary, for n > 1, $|E|^n < |E|$ which means the energy landscape is flattened. Fortunately, because the value of the energy can be changed up to only ± 1 these opposite effects are not pronounced as they are unnoticeable in Figure 4.1. Also since most energy values on the energy landscape are less than 0, these opposite effects are less likely to happen during the optimization process.

The next three sections discuss the derivation of optimization algorithms with



Figure 4.2: Comparison of unmodified energy landscape with n = 1, XEL with n = 2, and XEL with n = 0.5 in the range of |E| < 1. The n = 0.5 XEL accentuates the extrema while the n = 2 XEL flattens the landscape.

the XEL. These algorithms are implemented in simulations and are compared to unmodified algorithms.

4.1.2Newton's Method with Exponential Energy Landscaping (Newton-XEL)

The Newton-XEL is derived from the Taylor's series expansion of the first derivative of the modified energy landscape E^* . Let h^* be a step that gives a minimum energy or zero gradient of the modified energy landscape. The Taylor series expansion of the derivative of the modified energy function becomes

$$0 = \frac{\partial E^*(\theta_k + h^*)}{\partial \theta} = \frac{\partial E^*(\theta_k)}{\partial \theta} + \frac{\partial^2 E^*(\theta_k)}{\partial^2 \theta} h^* + \frac{1}{2} h^{*T} \frac{\partial^3 E^*(\theta_k)}{\partial^3 \theta} h^* + \cdots$$

Dropping the high-order terms and rearranging gives the Newton-XEL step

$$\hat{h}^* = -\left(\frac{\partial^2 E^*(\theta_k)}{\partial^2 \theta}\right)^{-1} \frac{\partial E^*(\theta_k)}{\partial \theta}$$
(4.2)

where \hat{h}^* is an estimation for h^* .

Next the first and the second derivatives of E^* are determined. Since

$$f(\theta) = E^* = sign(E) |E|^n = E |E|^{n-1} = E \sqrt{E^2}^{n-1}$$

the first derivative becomes

$$\frac{\partial f(\theta)}{\partial \theta} = \frac{\partial E^*}{\partial \theta} = \frac{\partial E}{\partial \theta} \sqrt{E^2}^{n-1} + E \frac{\partial \left(\sqrt{E^2}^{n-1}\right)}{\partial \theta}
= -\tau \sqrt{E^2}^{n-1} + E(n-1) \sqrt{E^2}^{n-2} \left(\frac{1}{2\sqrt{E^2}}\right) 2E(-\tau)
= -\left(\sqrt{E^2}^{n-1} + (n-1) \sqrt{E^2}^{n-1}\right) \tau
= -n \sqrt{E^2}^{n-1} \tau
= -n |E|^{n-1} \tau
= -\beta \tau
= -\tau^*$$
(4.3)

,

where $\beta \equiv n |E|^{n-1}$ and $\tau^* \equiv \beta \tau$ is a torque vector of E^* . The second derivative becomes

$$\frac{\partial^2 f(\theta)}{\partial \theta^2} = \frac{\partial^2 E^*}{\partial \theta^2} = -n \left(\frac{\partial \left(\sqrt{E^2}^{n-1} \right)}{\partial \theta} \tau + \sqrt{E^2}^{n-1} \frac{\partial \tau}{\partial \theta} \right)$$
$$= -n \left(-(n-1) \sqrt{E^2}^{n-3} E(\tau \tau^T) + \sqrt{E^2}^{n-1} \left(-K \right) \right)$$
$$= n \sqrt{E^2}^{n-1} \left((n-1) \frac{E}{\sqrt{E^2}^2} \tau \tau^T + K \right)$$
$$= n \left| E \right|^{n-1} \left((n-1) \frac{E}{E^2} \tau \tau^T + K \right)$$
$$= \beta \left((n-1) E^{-1} \tau \tau^T + K \right)$$
$$\equiv K^*$$
(4.4)

where K^* is the Hessian matrix of E^* .

Substituting the first and the second derivative terms into Equation (4.2) the Newton-XEL step becomes

$$\hat{h}^* = (K^*)^{-1} \tau^* \tag{4.5}$$

Thus K^{-1} in Newton's method (3.5) is generalized to $(K^*)^{-1}$ which changes the direction and the magnitude of τ^* to give \hat{h}^* . Note that for n = 1, τ^* becomes τ , $(K^*)^{-1}$ becomes K^{-1} , and \hat{h}^* becomes \hat{h} . While K^{-1} and $(K^*)^{-1}$ yield different magnitudes of \hat{h} and \hat{h}^* , they unexpectedly yield the same direction. This characteristic and its proof are discussed in more detail in Section 4.2.2 in which it is also shown that the directions are distinct for the QNA and the QNA-XEL cases. The QNA-XEL is discusses below. Figure 4.3 gives a pseudo-code for the Newton-XEL with a line search. Initialize ϵ , n, θ_0 , and θ_1 Calculate E_0 for $k = 1, ..., k_{max}$ do Calculate E_k , τ_k , and K_k if $\left| \frac{E_k - E_{k-1}}{E_{k-1}} \right| < \epsilon$ then break // Convergence criterion met. end if $\beta_k = n |E_k|^{n-1}$ $K_k^* = \beta_k ((n-1)E_k^{-1}\tau_k\tau_k^T + K_k)$ $\tau_k^* = \beta_k \tau_k$ $\hat{h}_k^* = (K_k^*)^{-1} \tau_k^* //$ Determine the Newton-XEL step $\alpha_k = \texttt{line_search}(\hat{h}_k^*) //$ Determine the line search gain $\theta_{k+1} = \theta_k + \alpha_k \hat{h}_k^* // (k+1)^{\text{th}}$ solution

end for

Figure 4.3: A pseudo-code for Newton-XEL with a line search. Note that for Newton-XEL (*Hessian only*), described in Section 4.1.4, $\tau_k^* = \tau_k$.

4.1.3 Quasi-Newton Algorithm with Exponential Energy Landscaping (QNA-XEL)

The QNA-XEL is similar to the Newton-XEL update except that the Hessian matrix K is estimated. The QNA-XEL step is

$$\hat{h}^{*} = \beta^{-1} \left((n-1)E^{-1}\tau\tau^{T} + \hat{K} \right)^{-1} \beta\tau$$

$$= \left(\hat{K}^{*} \right)^{-1} \tau^{*}$$
(4.6)

where

$$\hat{K}^* \equiv \beta \left((n-1)E^{-1}\tau\tau^T + \hat{K} \right) \tag{4.7}$$

is the estimated Hessian matrix of the modified energy landscape. Note that when n = 1 the QNA-XEL step (4.6) becomes the QNA step (3.6). Broyden's method (3.7) can be used to estimate \hat{K} in the \hat{K}^* but it reduces the effect of XEL on the QNA because it does not allow the direction of the QNA-XEL step to differ from the QNA step which is discussed in Section 4.2.2. The Broyden-XEL described below is used to estimate \hat{K}^* instead.

4.1.3.1 Broyden's method with XEL (Broyden-XEL)

The Broyden-XEL has a different $\Delta \tau$ term defined by replacing E in Equation (3.7) with E^* ,

$$\Delta \tau_{k-1}^* = -\frac{\partial E^*(\theta_k)}{\partial \theta} - \left(-\frac{\partial E^*(\theta_{k-1})}{\partial \theta}\right)$$
$$= \tau_k^* - \tau_{k-1}^*$$
(4.8)

where $\Delta \tau^*$ is a change in the torque vector of E^* . Since $\tau^* = \beta \tau$, terms in $\Delta \tau^*$ are similar to $\Delta \tau$ but weighted by β 's of the current and previous configurations, which contain the information about the energy landscape. In addition to $\Delta \tau^*$, the other two modified terms are \hat{K}^* and \hat{h}^* which are substituted into (3.7) to yield the Broyden-XEL update for \hat{K}^*

$$\hat{K}_{k}^{*} = \hat{K}_{k-1}^{*} + \left(\hat{h}_{k-1}^{*}{}^{T}\hat{h}_{k-1}^{*}\right)^{-1} \left(\Delta\tau_{k-1}^{*} - \hat{K}_{k-1}^{*}\hat{h}_{k-1}^{*}\right) \hat{h}_{k-1}^{*}$$

$$(4.9)$$

Note when n = 1 (4.9) reduces to (3.7).

As discussed in Section 4.2.2, the direction of the QNA-XEL step is different from that of the QNA step. From this point forward QNA-XEL is (4.6) that uses Broyden-XEL (4.9) to estimated \hat{K}^* and QNA-XEL (*Broyden's method*) is (4.6) that uses Broyden's method (3.7) to estimate \hat{K} . Figure 4.4 gives a pseudo-code for the QNA-XEL with a line search.

Pseudo-code: QNA-XEL with a line search

Initialize ϵ , n, θ_0 , θ_1 , and \hat{K}_0^* Calculate E_0 , and τ_0 for $k = 1, \ldots, k_{max}$ do Calculate E_k , and τ_k if $\left|\frac{E_k - E_{k-1}}{E_{k-1}}\right| < \epsilon$ then break // Convergence criterion met. end if Calculate \hat{K}_k^* $\hat{h}_{k-1}^* = \theta_k - \theta_{k-1}$ $\beta_k = n \cdot |E_k|^{n-1}$ $\beta_{k-1} = n \cdot |E_{k-1}|^{n-1}$ $\Delta \tau_{k-1}^* = \beta_k \tau_k - \beta_{k-1} \tau_{k-1}$ $\hat{K}_{k}^{*} = \hat{K}_{k-1}^{*} + \left(\hat{h}_{k-1}^{*} \hat{h}_{k-1}^{*}\right)^{-1} \left(\Delta \tau_{k-1}^{*} - \hat{K}_{k-1}^{*} \hat{h}_{k-1}^{*}\right) \hat{h}_{k-1}^{*}$ $\tau_k^* = \beta_k \tau_k$ $\hat{h}_{k}^{*} = \left(\hat{K}_{k}^{*}\right)^{-1} \tau_{k}^{*}$ // Determine the QNA-XEL step $\alpha_k = \texttt{line_search}(\hat{h}_k^*) \; / / \; \text{Determine the line search gain}$

$$\theta_{k+1} = \theta_k + \alpha_k \hat{h}_k^* / (k+1)^{\text{th}}$$
 solution

end for

Figure 4.4: A pseudo-code for QNA-XEL with a line search. Note that for QNA-XEL (*Hessian only*), described in Section 4.1.4, $\tau_k^* = \tau_k$.
4.1.4 Broyden-Fletcher-Goldfarb-Shanno Algorithm with Exponential Energy Landscaping(BFGS-XEL)

Implementing XEL in BFGS changes the objective function from E to E^* and therefore τ becomes τ^* . The BFGS-XEL step becomes

$$\hat{h}_{k}^{*} = \hat{H}_{k}^{*} \tau_{k}^{*} \tag{4.10}$$

where \hat{H}^* is the estimated inverse of the Hessian matrix of E^* . Substituting the terms for E^* , $\Delta \tau^*$ (4.8) and \hat{h}^* into (3.8) yields the BFGS-XEL update for \hat{H}^*

$$\hat{H}_{k}^{*} = \hat{H}_{k-1}^{*} - \left(\frac{\hat{H}_{k-1}^{*} \Delta \tau_{k-1}^{*} \hat{h}_{k-1}^{*}^{T} + \hat{h}_{k-1}^{*} \Delta \tau_{k-1}^{*}^{T} \hat{H}_{k-1}^{*}}{\hat{h}_{k-1}^{*} \Delta \tau_{k-1}^{*}} - 1\right) + \left(\frac{\Delta \tau_{k-1}^{*} \hat{H}_{k-1}^{*} \Delta \tau_{k-1}^{*}}{\hat{h}_{k-1}^{*} \Delta \tau_{k-1}^{*}} - 1\right) \frac{\hat{h}_{k-1}^{*} \hat{h}_{k-1}^{*}}{\hat{h}_{k-1}^{*} \Delta \tau_{k-1}^{*}}$$
(4.11)

Used with a line search the BFGS-XEL update becomes

$$\theta_{k+1} = \theta_k + \alpha_k \hat{H}_k^* \tau_k^*$$
$$= \theta_k + \alpha_k \beta_k \hat{H}_k^* \tau_k$$

since $\tau^* = \beta \tau$. Define $\alpha_k^* \equiv \alpha_k \beta_k$ is the new gain found by a line search without an explicit value of β_k . In theory α_k and α_k^* would globally minimize the energy function and yield the same total steps since $\hat{H}_k^* \tau_k^*$ and $\hat{H}_k^* \tau_k$ have the same direction. However, in practice α_k and α_k^* only locally minimizes the energy function so they could yield different total steps since $\hat{H}_k^* \tau_k^*$ and $\hat{H}_k^* \tau_k$ have different magnitudes.

To study an impact of the different magnitudes of the step on the quality of resulted configurations and speed, a variation of the BFGS-XEL of which the step equals $\hat{H}_k^* \tau_k$ is investigated. Letting $\tau^* = \tau$ keeps the same notation and the variation of the BFGS-XEL step (4.10) becomes

$$\begin{aligned} \tau_k^* &= \tau_k \\ \hat{h}_k^* &= \hat{H}_k^* \tau_k^* \\ \alpha_k^* &= \texttt{line_search}(\hat{h}_k^*) \\ \theta_{k+1} &= \theta_k + \alpha_k^* \hat{h}_k^* \end{aligned}$$

Letting $\tau^* = \tau$ is basically substituting only the estimated inverse of the Hessian matrix for E^* . From this point forward the variation of the BFGS-XEL is referred to as BFGS-XEL (*Hessian only*). Similarly, by letting $\tau_k^* = \tau_k$, the Newton-XEL and the QNA-XEL become Newton-XEL (*Hessian only*) and QNA-XEL (*Hessian only*). Note that at the end of this chapter Tables 4.2 through 4.5 on pages 83–86 summarize all presented methods.

Comparing the Newton-XEL,

$$\hat{h}^{*} = (K_{k}^{*})^{-1} \tau_{k}^{*}$$

$$= \beta^{-1} \left((n-1)E^{-1}\tau_{k}\tau_{k}^{T} + K \right)^{-1} \beta \tau_{k}$$

$$= \left((n-1)E^{-1}\tau_{k}\tau_{k}^{T} + K \right)^{-1} \tau_{k}$$
(4.12)

to the Newton-XEL (*Hessian only*),

$$\hat{h}^* = (K_k^*)^{-1} \tau_k^*$$
$$= \beta^{-1} \left((n-1)E^{-1} \tau_k \tau_k^T + K \right)^{-1} \tau_k$$

The Newton-XEL (*Hessian only*) contains an extra term β^{-1} which increases the influence of n on the magnitudes of the step. Since $\beta = n |E|^{n-1}$, $\beta^{-1} = \frac{1}{n} |E|^{1-n}$ which is inversely dependent on n so the magnitudes of the Newton-XEL step (*Hessian only*) are expected to be larger when n < 1 and smaller when n > 1. This is consistent with the influence of n on the magnitudes of the XEL step discussed in Section 4.2.4 and the simulations discussed in Section 5.

Figure 4.5 gives a pseudo-code for the BFGS-XEL with a line search.

Pseudo-code: BFGS-XEL with a line search

Initialize ϵ , n, θ_0 , θ_1 , and \hat{H}_0^*

Calculate E_0 and τ_0

for $k = 1, \ldots, k_{max}$ do

Calculate E_k , and τ_k

if
$$\left|\frac{E_k - E_{k-1}}{E_{k-1}}\right| < \epsilon$$
 then

break // Convergence criterion met.

end if

Calculate \hat{H}_k^*

$$\begin{aligned} \hat{h}_{k-1}^{*} &= \theta_{k} - \theta_{k-1} \\ \beta_{k} &= n \cdot |E_{k}|^{n-1} \\ \beta_{k-1} &= n \cdot |E_{k-1}|^{n-1} \\ \Delta \tau_{k-1}^{*} &= \beta_{k} \tau_{k} - \beta_{k-1} \tau_{k-1} \\ \hat{H}_{k}^{*} &= \hat{H}_{k-1}^{*} - \left(\frac{\hat{H}_{k-1}^{*} \Delta \tau_{k-1}^{*} \hat{h}_{k-1}^{*} T + \hat{h}_{k-1}^{*} \Delta \tau_{k-1}^{*} T \hat{H}_{k-1}^{*}}{\hat{h}_{k-1}^{*} T \Delta \tau_{k-1}^{*}} - 1 \right) \frac{\hat{h}_{k-1}^{*} \hat{h}_{k-1}^{*} T - 1}{\hat{h}_{k-1}^{*} T \Delta \tau_{k-1}^{*}} \\ \tau_{k}^{*} &= \beta_{k} \tau_{k} \end{aligned}$$

 $\hat{h}_{k}^{*} = \hat{H}_{k}^{*} \tau_{k}^{*} // \text{ Determine the BFGS-XEL step}$ $\alpha_{k} = \texttt{line_search}(\hat{h}_{k}^{*}) // \text{ Determine the line search gain}$ $\theta_{k+1} = \theta_{k} + \alpha_{k} \hat{h}_{k}^{*} // (k+1)^{\text{th}} \text{ solution}$

end for

Figure 4.5: A pseudo-code for BFGS-XEL with a line search. Note that for BFGS-XEL (*Hessian only*) $\tau_k^* = \tau_k$.

4.2 Discussion on XEL

This section compares the Newton-XEL to the Newton steps with a basic assumption that these steps exist, i.e., K and K^* are invertible. Section 4.2.1 presents a sufficient condition for the positive definite Hessian matrix of the Newton-XEL step which is important for minimization. Section 4.2.2 shows that the Newton-XEL and the Newton steps are parallel and only the magnitude of Newton-XEL step is dependent on n. Section 4.2.4 shows that the magnitude of the Newton-XEL step is inversely related to n. Section 4.2.5 finds that the error residues from both algorithms are the same order. Section 4.2.6 discusses how the XEL method weights terms in the Hessian matrix and its estimation. Lastly, Section 4.2.7 concludes the discussions and summarizes the list of guidelines for changing n.

4.2.1 A Positive Definite Property of the Newton-XEL Hessian Matrix

Derived from the quadratic approximation of a function, Newton's method assumes that the function is convex with a global minimum if the Hessian matrix is positive definite and the Newton step gives the solution of the quadratic approximation. Although the actual function may not be convex and the Newton step may not give a local solution to the actual function, the positive definite property of the Newton Hessian matrix K implies that the Newton step is in descending direction.

This section shows that if the Hessian matrix of the Newton's method K is positive definite then so is K^* (the Hessian matrix of the Newton-XEL method) when an additional condition is satisfied. A positive definite K satisfies

$$x^T K x > 0; \quad \forall x \neq 0, x \in \Re^N$$

where N is the dimension of the energy landscape. Since $K^T = K$, it follows from

$$K^* = \beta \left((n-1)E^{-1}\tau\tau^T + K \right)$$
[4.4]

that $K^* = K^{*T}$ and is thus symmetric. K^* is positive definite if

$$x^{T}(K^{*}) x > 0; \quad \forall x \neq 0, x \in \Re^{N}$$

$$(4.13)$$

Substituting in (4.4) and expanding gives

$$\beta(n-1)E^{-1}\left(\tau^T x\right)^T \tau^T x + \beta x^T K x > 0$$

Noting that $\beta = n |E|^{n-1} > 0$ because n > 0, that $\tau^T x$ is a scalar so

$$\left(\tau^T x\right)^T \tau^T x = \left|\tau^T x\right|^2 \ge 0$$

and that $x^T K x > 0$ then Equation (4.13) is satisfied when

$$(n-1)E^{-1} \ge 0$$

Note that this is only a sufficient condition for K^* to be positive definite. Further, since $n \neq 1$ corresponds to the modified energy landscape, $(n-1)E^{-1} \neq 0$ and the equality is dropped. Also, E^{-1} is undefined when E = 0 so the sufficient condition is stated as,

$$(n-1)E^{-1} > 0, \quad E \neq 0$$
 (4.14)

This condition can be satisfied in either of two ways:

$$E > 0$$
 and $n > 1$ or $(4.15a)$

$$E < 0 \quad \text{and} \quad n < 1 \tag{4.15b}$$

For an adaptively modified energy landscape algorithm, Equation (4.15) gives a guideline for determining n so that K^* is positive definite. During the beginning of the conformation prediction process the energy is high and often positive, but as it progresses the energy decreases and often becomes negative. This suggests that n > 1 should be used at the beginning and n < 1 should be used toward the end of the process. However, enforcing the rule may not be very beneficial for the Newton-XEL method because it is derived under an assumption that K is positive definite so if K is not positive definite, K^* may not be.

Similarly it can be shown that the same condition applies to the estimated Hessian matrices of the QNA and the QNA-XEL (*Broyden's method*). With the same above reasoning, the Hessian matrix of the QNA-XEL (*Broyden's method*) \hat{K}^* is positive definite when the Hessian matrix \hat{K} is positive definite and Equation (4.15) is satisfied. However, enforcing the rule may not be very beneficial for the QNA-XEL (*Broyden's method*) because the actual K may not be positive definite and thus the QNA-XEL (*Broyden's method*) step may not be in descending direction. A more effective way to ensure the descending direction is using a line search which has been discussed in Section 3.1.1.

4.2.2 Direction of Newton-XEL Step

This section discusses the effect of XEL on the direction of the Newton-XEL step. Expanding the Newton-XEL step gives

$$\hat{h}^* = \left((n-1)E^{-1}\tau\tau^T + K \right)^{-1}\tau$$
[4.12]

With the term $(n-1)E^{-1}\tau\tau^{T}$ added to K, the Newton-XEL step would be expected to be in different direction from the Newton step. However, only the magnitudes differ so they are actually parallel. Section 4.2.2.1 gives a derivation of the Newton-XEL step for a two dimensional (2D) energy landscape which is shown equal to the Newton step multiplied by a constant. Section 4.2.2.2 gives a proof for the N dimensional case.

4.2.2.1 Derivation of the 2D Newton-XEL step

Consider the K^* term in the Newton-XEL step. Since

$$K^* = \beta \left((n-1)E^{-1}\tau\tau^T + K \right)$$
$$= \begin{bmatrix} \beta \left(\frac{n-1}{E} \left(\frac{\partial E}{\partial \theta_1} \right)^2 + \frac{\partial^2 E}{\partial \theta_1^2} \right) & \beta \left(\frac{n-1}{E} \frac{\partial E}{\partial \theta_1} \frac{\partial E}{\partial \theta_2} + \frac{\partial^2 E}{\partial \theta_1 \partial \theta_2} \right) \\ \beta \left(\frac{n-1}{E} \frac{\partial E}{\partial \theta_1} \frac{\partial E}{\partial \theta_2} + \frac{\partial^2 E}{\partial \theta_1 \partial \theta_2} \right) & \beta \left(\frac{n-1}{E} \left(\frac{\partial E}{\partial \theta_2} \right)^2 + \frac{\partial^2 E}{\partial \theta_2^2} \right) \end{bmatrix}$$

the inverse is,

$$(K^*)^{-1} = \frac{1}{\det(K^*)}$$

$$\cdot \begin{bmatrix} \beta \left(\frac{n-1}{E} \left(\frac{\partial E}{\partial \theta_2} \right)^2 + \frac{\partial^2 E}{\partial \theta_2^2} \right) & \beta \left(-\frac{n-1}{E} \frac{\partial E}{\partial \theta_1} \frac{\partial E}{\partial \theta_2} - \frac{\partial^2 E}{\partial \theta_1 \partial \theta_2} \right) \\ \beta \left(-\frac{n-1}{E} \frac{\partial E}{\partial \theta_1} \frac{\partial E}{\partial \theta_2} - \frac{\partial^2 E}{\partial \theta_1 \partial \theta_2} \right) & \beta \left(\frac{n-1}{E} \left(\frac{\partial E}{\partial \theta_1} \right)^2 + \frac{\partial^2 E}{\partial \theta_1^2} \right) \end{bmatrix}$$

The Newton-XEL step is

$$\hat{h}^* = (K^*)^{-1} \tau^*$$

$$= \frac{1}{\det(K^*)} \cdot \begin{bmatrix} \beta^2 \left(\frac{\partial^2 E}{\partial \theta_2^2} \left(-\frac{\partial E}{\partial \theta_1} \right) - \frac{\partial^2 E}{\partial \theta_1 \partial \theta_2} \left(-\frac{\partial E}{\partial \theta_2} \right) \right) \\ \beta^2 \left(-\frac{\partial^2 E}{\partial \theta_1 \partial \theta_2} \left(-\frac{\partial E}{\partial \theta_1} \right) + \frac{\partial^2 E}{\partial \theta_1^2} \left(-\frac{\partial E}{\partial \theta_2} \right) \right) \end{bmatrix}$$

and factorizing gives

$$\hat{h}^* = \frac{\beta^2}{\det(K^*)} \cdot \begin{bmatrix} \frac{\partial^2 E}{\partial \theta_2^2} & -\frac{\partial^2 E}{\partial \theta_1 \partial \theta_2} \\ -\frac{\partial^2 E}{\partial \theta_1 \partial \theta_2} & \frac{\partial^2 E}{\partial \theta_1^2} \end{bmatrix} \begin{bmatrix} -\frac{\partial E}{\partial \theta_1} \\ -\frac{\partial E}{\partial \theta_2} \end{bmatrix}$$

Since

$$\det(K) \cdot K^{-1} = \begin{bmatrix} \frac{\partial^2 E}{\partial \theta_2^2} & -\frac{\partial^2 E}{\partial \theta_1 \partial \theta_2} \\ -\frac{\partial^2 E}{\partial \theta_1 \partial \theta_2} & \frac{\partial^2 E}{\partial \theta_1^2} \end{bmatrix}$$

and

$$\tau = \begin{bmatrix} -\frac{\partial E}{\partial \theta_1} \\ -\frac{\partial E}{\partial \theta_2} \end{bmatrix}$$

then

$$\hat{h}^* = \frac{\beta^2 \det(K)}{\det(K^*)} K^{-1} \tau$$
$$= \frac{\beta^2 \det(K)}{\det(K^*)} \hat{h}$$
(4.16)

Since the constant term $\frac{\beta^2 \det(K)}{\det(K^*)}$ only affects the length of the vector $K^{-1}\tau$, the direction of \hat{h}^* only depends on K^{-1} not n and the Newton-XEL step is parallel to the Newton step. This means that XEL can have less effect on Newton's method than on other algorithms and that changing the n value may not significantly impact the Newton-XEL optimization path. Results presented in Chapter 5 show that on a special energy landscape case the Newton-XEL paths are not varied by modifying n at all.

Affecting only the magnitude of the Newton step, XEL acts similarly to a line search which finds a gain that is a scalar multiplier to the Newton step. One difference is that the line search goes through multiple iterations to find an "optimal" gain that minimizes the energy along the Newton step while the Newton-XEL only goes through one iteration and may not give an optimal gain. Thus, XEL can not replace a line search but can be used to improve performance of the Newton's method and a line search. The Newton-XEL with a line search may give the same results as the Newton's method with a line search, but not always, especially on a rough energy landscape. This is because these methods give different magnitudes of the step and therefore a line search can result in different gains corresponding to different local minima.

Similar to the Newton-XEL step, the QNA-XEL (*Broyden's method*) step, (4.6) and (3.7), on the 2D energy landscape can be derived to show that its direction depends on only the estimated Hessian matrix \hat{K} , i.e., $\hat{h}^* = \frac{\beta^2 \det(\hat{K})}{\det(\hat{K}^*)} \hat{K}^{-1} \tau$. Therefore, it has the same direction as the QNA step, (3.6) and (3.7), and it is independent of n.

The QNA-XEL uses the Broyden-XEL (4.9) which is dependent on n to estimate \hat{K}^* . To show that the direction of the QNA-XEL step is dependent on n, the 2D case

of the $k^{\rm th}$ QNA-XEL step is derived. Consider \hat{K}_k^*

$$\hat{K}_{k}^{*} = \hat{K}_{k-1}^{*} + \left(\hat{h}_{k-1}^{*} \hat{h}_{k-1}^{*}\right)^{-1} \left(\Delta \tau_{k-1}^{*} - \hat{K}_{k-1}^{*} \hat{h}_{k-1}^{*}\right) \hat{h}_{k-1}^{*}^{T}$$
$$= \tilde{K} + \Delta \tau_{k-1}^{*} \tilde{h}$$

where $\tilde{K} \equiv \hat{K}_{k-1}^* - \left(\hat{h}_{k-1}^* \hat{h}_{k-1}^*\right)^{-1} \hat{K}_{k-1}^* \left(\hat{h}_{k-1}^* \hat{h}_{k-1}^*\right)^T$ with elements $\tilde{K}_i, i = 1, \cdots, 4$, and $\tilde{h} \equiv \left(\hat{h}_{k-1}^* \hat{h}_{k-1}^*\right)^{-1} \hat{h}_{k-1}^*^T$ with elements $\tilde{h}_i, i = 1, 2, \Delta \tilde{\tau}_i$ be elements of $\Delta \tau_{k-1}^*$. $\hat{K}_k^*^{-1}$ can be written as

$$\hat{K}_{k}^{*}{}^{-1} = \frac{1}{\det\left(\hat{K}_{k}^{*}\right)} \begin{bmatrix} \tilde{K}_{4} + \Delta \tilde{\tau}_{2}\tilde{h}_{2} & -\tilde{K}_{2} - \Delta \tilde{\tau}_{1}\tilde{h}_{2} \\ -\tilde{K}_{3} - \Delta \tilde{\tau}_{2}\tilde{h}_{1} & \tilde{K}_{1} + \Delta \tilde{\tau}_{1}\tilde{h}_{1} \end{bmatrix}$$

Let $\tilde{\tau}_i$ be elements of τ_k^* the QNA-XEL step becomes

$$\hat{K}_{k}^{*}{}^{-1}\tau_{k}^{*} = \frac{1}{\det\left(\hat{K}_{k}^{*}\right)} \begin{bmatrix} \tilde{K}_{4}\tilde{\tau}_{1} + \Delta\tilde{\tau}_{2}\tilde{h}_{2}\tilde{\tau}_{1} - \tilde{K}_{2}\tilde{\tau}_{2} - \Delta\tilde{\tau}_{1}\tilde{h}_{2}\tilde{\tau}_{2} \\ -\tilde{K}_{3}\tilde{\tau}_{1} - \Delta\tilde{\tau}_{2}\tilde{h}_{1}\tilde{\tau}_{1} + \tilde{K}_{1}\tilde{\tau}_{2} + \Delta\tilde{\tau}_{1}\tilde{h}_{1}\tilde{\tau}_{2} \end{bmatrix}$$
$$= \frac{1}{\det\left(\hat{K}_{k}^{*}\right)} \begin{bmatrix} \tilde{K}_{4}\tilde{\tau}_{1} - \tilde{K}_{2}\tilde{\tau}_{2} + (\Delta\tilde{\tau}_{2}\tilde{\tau}_{1} - \Delta\tilde{\tau}_{1}\tilde{\tau}_{2})\tilde{h}_{2} \\ -\tilde{K}_{3}\tilde{\tau}_{1} + \tilde{K}_{1}\tilde{\tau}_{2} - (\Delta\tilde{\tau}_{2}\tilde{\tau}_{1} - \Delta\tilde{\tau}_{1}\tilde{\tau}_{2})\tilde{h}_{1} \end{bmatrix}$$

Since the scalar $\Delta \tilde{\tau}_2 \tilde{\tau}_1 - \Delta \tilde{\tau}_1 \tilde{\tau}_2$ which depends on n can not be factored out and it does not equal zero, the direction of the QNA-XEL step is dependent of n. This is expected to be true for the N dimensional case.

4.2.2.2 N dimensional Newton-XEL step

For an N dimensional energy landscape the Newton-XEL step is shown to always be parallel to the Newton step so the value of n does not affect the direction of the Newton-XEL step.

Proof that the Newton-XEL step is parallel to the Newton step:

Let \hat{h} and \hat{h}^* be the Newton and the Newton-XEL steps of the N dimensional energy landscape respectively,

$$\hat{h} = K^{-1}\tau \tag{3.5}$$

$$\hat{h}^* = (K^*)^{-1} \tau^*$$

$$= \beta (K^*)^{-1} \tau$$
[4.5]

Eliminating τ gives

$$\hat{h}^* = \beta \left(K^* \right)^{-1} K \hat{h} \tag{4.17}$$

Using K^* in (4.4) and then (3.5) gives

$$\hat{h}^* = \left((n-1)E^{-1}\tau\tau^T + K \right)^{-1}K\hat{h}$$
$$= \left((n-1)E^{-1} \left(K^{-1}\tau \right)\tau^T + I \right)^{-1}\hat{h}$$
$$= \left((n-1)E^{-1}\hat{h}\tau^T + I \right)^{-1}\hat{h}$$

However, \hat{h} is an eigenvector of $\left((n-1)E^{-1}\hat{h}\tau^T + I\right)^{-1}$, and thus also its inverse so,

$$\left((n-1)E^{-1}\hat{h}\tau^T + I\right)\hat{h} = (n-1)E^{-1}\hat{h}\tau^T\hat{h} + \hat{h}$$
$$= \lambda\hat{h}$$

where

$$\lambda = (n-1)E^{-1}(\tau^T \hat{h}) + 1 \tag{4.18}$$

and where τ and \hat{h} are column vectors and $\tau^T \hat{h}$ is a scalar. Thus

$$\hat{h}^* = \lambda \hat{h} \tag{4.19}$$

and \hat{h}^* is a scalar multiple of \hat{h} so the Newton-XEL step is parallel to the Newton step. $\hfill \square$

Clearly when n = 1, so the energy landscape is not modified, then $\lambda = 1$ and the Newton-XEL step becomes the Newton step. The scalar multiplier λ has several properties discussed next in Section 4.2.3.

4.2.3 Properties of the Scalar Multiplier λ

This section discusses several properties of λ , which are used to derive the relationship between n and the magnitude of Newton-XEL step in Section 4.2.4 and the error residue in the Newton-XEL method in Section 4.2.5. As Section 4.2.2.2 previously defined λ as an eigenvalue of a matrix, here Section 4.2.3.1 shows that λ is also an eigenvalue of another matrix $\beta K (K^*)^{-1}$. Section 4.2.3.2 derives the other eigenvalue of $\beta K (K^*)^{-1}$. Lastly, Section 4.2.3.3 discusses the relationship between λ and a positive definite property of K and K^* .

4.2.3.1 Eigenvalue λ

Here λ is found to be an eigenvalue of $\beta K(K^*)^{-1}$. Since

$$\hat{h}^* = \lambda \hat{h}$$

substituting \hat{h}^* and \hat{h} from (4.5) and (3.5) and multiplying by K yield

$$\beta K \left(K^* \right)^{-1} \tau = \lambda \tau$$

and thus τ is an eigenvector of $\beta K(K^*)^{-1}$ corresponding to an eigenvalue λ .

4.2.3.2 Other eigenvalues

This section finds other eigenvalues of $\beta K (K^*)^{-1}$ first for the 2D case and then a proof is given for the *N*-dimensional case. From the derivation of the 2D Newton-XEL step (4.16) the λ for this case is

$$\lambda = \frac{\beta^2 \det(K)}{\det(K^*)}$$

By the property of the determinant of a matrix product [67] this is equal to

$$\lambda = \beta^2 \det \left(K \left(K^* \right)^{-1} \right) = \det \left(\beta K \left(K^* \right)^{-1} \right)$$

Since the product of the eigenvalues equals the determinant of the matrix [67], the other eigenvalue of the 2D matrix $\beta K (K^*)^{-1}$ is 1. The same can be shown for

 $\beta (K^*)^{-1} K$ since

$$\lambda = \det\left(\beta \left(K^*\right)^{-1} K\right)$$

The other eigenvalue of the 2D matrix $\beta (K^*)^{-1} K$ is also 1.

For the N-dimensional $\beta K (K^*)^{-1}$, the only additional eigenvalue is 1 with N-1 corresponding eigenvectors. This is shown in the following proof.

Proof that in N dimensions an eigenvalue 1 has N - 1 corresponding eigenvectors: To show that 1 is an eigenvalue of $\beta K (K^*)^{-1}$ matrix, it is equivalent to show that 1 is an eigenvalue of the inverse matrix $\beta (K^*) K^{-1}$ with N-1 corresponding eigenvectors. For 1 to be an eigenvalue the following characteristic equation must be satisfied,

$$\det \left(\beta \left(K^{*}\right) K^{-1} - I\right) = \det \left(\beta \left((n-1)E^{-1}\tau\tau^{T} + K\right) K^{-1} - I\right) = 0$$

or

$$\det\left(\beta(n-1)E^{-1}\tau\tau^{T}K^{-1}\right) = 0$$

Determinant properties for the product of a matrix and a scalar [67] give

$$\left(\beta(n-1)E^{-1}\right)^N \det\left(\tau\tau^T\right) \det\left(K^{-1}\right) = 0$$

This is satisfied identically since $\tau \tau^T$ is a rank one matrix so det $(\tau \tau^T) = 0$. Therefore, 1 is an eigenvalue of $(K^*) K^{-1}$.

To find a corresponding eigenvector to 1 it must satisfy

$$(\beta (K^*) K^{-1} - I) x = 0$$
$$(\beta ((n-1)E^{-1}\tau\tau^T + K) K^{-1} - I) x = 0$$
$$\beta (n-1)E^{-1}\tau\tau^T K^{-1} x = 0$$

Let $y = K^{-1}x$ then

$$\left(\tau\tau^{T}\right)y=0$$

so y is in the null space of $(\tau \tau^T)$ and has N-1 unique solutions. Since K has full rank x also has N-1 unique solutions which means the $\beta(K^*) K^{-1}$ matrix has N-1eigenvectors corresponding to 1.

Hence, the $\beta K(K^*)^{-1}$ matrix also has 1 as the other eigenvalue with N-1 corresponding eigenvectors.

A similar proof that matrix $\beta (K^*)^{-1} K$ has 1 as the eigenvalue with N-1 corresponding eigenvectors can be given as well.

4.2.3.3 A relationship between λ and a positive definite property

This section shows how λ is related to a positive definite property. Since the product of all eigenvalues is the determinant of the matrix,

$$\lambda = \det \left(\beta K \left(K^*\right)^{-1}\right)$$

$$= \frac{\beta^N \det(K)}{\det(K^*)}$$

$$= \beta^N \prod_{i=1}^N \mu_i / \prod_{i=1}^N \mu_i^*$$
(4.20)

where μ 's are eigenvalues of K and μ^* 's are eigenvalues of K^* . Since a positive definite matrix has all positive eigenvalues, their product is positive. With β^N always positive, for both K and K^* to be positive definite λ must be positive. This is a necessary but not a sufficient condition because both det(K) and det (K^*) can be negative or either matrix can have an even number of negative eigenvalues. Note that Equation (4.20) gives a closed form of λ consistent with the 2D case.

4.2.4 Magnitude of Newton-XEL Step

This section derives another form of λ that reveals a relationship between the magnitude of the Newton-XEL step and n. From the property of eigenvectors and eigenvalues the inverse matrix $\beta^{-1}(K^*) K^{-1}$ has τ and $\frac{1}{\lambda}$ as an eigenvector and an eigenvalue,

$$\beta^{-1}\left(K^*\right)K^{-1}\tau = \frac{1}{\lambda}\tau$$

which reduces to

$$\left[(n-1)E^{-1}(\tau^T K^{-1}\tau) - \left(\frac{1}{\lambda} - 1\right) \right]\tau = 0$$

since $\tau^T K^{-1} \tau$ is a scalar. Vanishing of the scalar coefficient yields

$$\lambda = \frac{1}{1 + (n-1)E^{-1}(\tau^T K^{-1}\tau)}$$
(4.21)

Compared to Equation (4.18), this gives a clearer relationship between λ and n. While λ depends on the current properties of the energy landscape $(E, \tau, \text{ and } K)$ it also has an inverse relationship with n. Because λ is undefined when the denominator is zero, the inverse relationship would be true on the continuous intervals of λ . Since the Newton-XEL step is directly related to λ (4.19), it has an inverse relationship with n. As n gets smaller the magnitude of the Newton-XEL step becomes larger and vice versa. Although it is not proven here, this relationship is expected to carry over to other Newtonian XEL methods (QNA-XEL and BFGS-XEL) and has been supported by simulation results.

Equation (4.21) can also confirm that for both K and K^* to be positive definite λ must be positive. When K is positive definite so is K^{-1} and thus $\tau^T K^{-1} \tau > 0$. Also, when both K and K^* are positive definite, then from (4.14) $(n-1)E^{-1} > 0$ and therefore $\lambda > 0$.

4.2.5 Error residue from Newton-XEL step

This section compares the error residue of the Newton-XEL step to that of the Newton step. They are found to be of the same order.

First, the error residue from Newton step is determined. Let h be a step from θ_k that gives a minimum energy or zero gradient of the unmodified energy landscape at $\theta_k + h$. The Taylor series expansion of the derivative of the unmodified energy function becomes

$$0 = \frac{\partial E(\theta_k + h)}{\partial \theta}$$

= $\frac{\partial E(\theta_k)}{\partial \theta} + \frac{\partial^2 E(\theta_k)}{\partial^2 \theta} h + \frac{1}{2} h^T \frac{\partial^3 E(\theta_k)}{\partial^3 \theta} h + O(h^3)$
= $-\tau + Kh + \frac{1}{2} h^T \frac{\partial^3 E(\theta_k)}{\partial^3 \theta} h + O(h^3)$ (4.22)

The error residue ρ from using Newton step $\hat{h} = K^{-1}\tau$, which approximates h, is

$$\rho = \frac{1}{2}\hat{h}^{T} \frac{\partial^{3} E(\theta_{k})}{\partial^{3} \theta} \hat{h} + O(\hat{h}^{3})$$
$$= O(\hat{h}^{2})$$
(4.23)

Second, the error residue from Newton-XEL step is found. Let h^* be a step from θ_k that gives a minimum energy or zero gradient of the modified energy landscape at $\theta_k + h^*$. The Taylor series expansion of the derivatives of the modified energy function becomes

$$0 = \frac{\partial E^*(\theta_k + h^*)}{\partial \theta}$$
$$= \frac{\partial E^*(\theta_k)}{\partial \theta} + \frac{\partial^2 E^*(\theta_k)}{\partial^2 \theta} h^* + \frac{1}{2} h^{*T} \frac{\partial^3 E^*(\theta_k)}{\partial^3 \theta} h^* + O(h^{*3})$$
(4.24)

Since the minima on the modified and the unmodified energy landscape are at the same locations, $\theta_k + h = \theta_k + h^*$ and therefore

$$h = h^*$$

Substituting h^* for h in Equation (4.22) yields

$$0 = -\tau + Kh^* + \frac{1}{2} h^{*T} \frac{\partial^3 E(\theta_k)}{\partial^3 \theta} h^* + O(h^{*3})$$
(4.25)

In the energy landscaping approach, the Newton-XEL step is used to minimize the energy on the unmodified energy landscape. The error residue from using Newton-XEL step can be found by substituting the Newton-XEL step \hat{h}^* for h^* in Equation (4.25). Since $\hat{h}^* = \lambda \hat{h}$ from Equation (4.19), the Newton-XEL error residue ρ^* becomes

$$\rho^* = -\tau + K\lambda\hat{h} + \frac{1}{2}\left(\lambda\hat{h}\right)^T \frac{\partial^3 E(\theta_k)}{\partial^3 \theta}\lambda\hat{h} + O(\hat{h}^3)$$
$$\rho^* = -\tau + \lambda K\hat{h} + \frac{\lambda^2}{2}\hat{h}^T \frac{\partial^3 E(\theta_k)}{\partial^3 \theta}\hat{h} + O(\hat{h}^3)$$

Substituting $\hat{h} = K^{-1}\tau$ in the second term gives

$$\rho^* = -(1-\lambda)\tau + \frac{\lambda^2}{2}\hat{h}^T \frac{\partial^3 E(\theta_k)}{\partial^3 \theta}\hat{h} + O(\hat{h}^3)$$
(4.26)

Rearranging Equation (4.21) gives

$$1 - \lambda = \lambda(n-1)E^{-1}\left(\tau^T K^{-1}\tau\right)$$

which is substituted into Equation (4.26) to yield

$$\rho^* = -\lambda(n-1)E^{-1}\left(\tau^T K^{-1}\tau\right)\tau + \frac{\lambda^2}{2}\hat{h}^T \frac{\partial^3 E(\theta_k)}{\partial^3 \theta}\hat{h} + O(\hat{h}^3)$$

Using $I = K^{-T}K^{T}$, the error residue becomes

$$\rho^* = -\lambda(n-1)E^{-1}\left(\tau^T K^{-T} K^T K^{-1} \tau\right)\tau + \frac{\lambda^2}{2}\hat{h}^T \frac{\partial^3 E(\theta_k)}{\partial^3 \theta}\hat{h} + O(\hat{h}^3)$$

Since $\hat{h} = K^{-1}\tau$ and K is symmetric,

$$\rho^* = -\lambda(n-1)E^{-1}\left(\hat{h}^T K \hat{h}\right)\tau + \frac{\lambda^2}{2}\hat{h}^T \frac{\partial^3 E(\theta_k)}{\partial^3 \theta}\hat{h} + O(\hat{h}^3)$$
$$= O(\hat{h}^2)$$
(4.27)

The Newton-XEL error residue in Equation (4.27) is of the same order as the Newton error residue in Equation (4.23). Since the Newton-XEL method is Newton's method applied to the modified energy landscape, the convergence property of the Newton-XEL should be the similar to that of Newton's method which is confirmed by the equal order of the error residues of both methods.

Because the error residue of Newton-XEL is directly dependent on λ , a bounded λ would encourage stable optimization and convergence. Therefore, a guideline for choosing n in an adaptively modified energy landscape is to satisfy

$$|\lambda| < \delta \tag{4.28}$$

where δ is a bounding constant. This guideline is expected to be applicable to the QNA-XEL and the BFGS-XEL as well.

4.2.6 Weights on Different Energy Landscape Information

This section discusses how the XEL method gives different weights to the different information on the energy landscape in the XEL Hessian matrices and their estimation.

4.2.6.1 The Newton-XEL and the QNA-XEL (Broyden's method) Hessian matrices

Consider the Newton step in Equation (3.5) and the Newton-XEL step in Equation (4.5).

$$\hat{h} = K^{-1}\tau \tag{3.5}$$

$$\hat{h}^* = (K^*)^{-1} \tau^*$$

$$= \left((n-1)E^{-1}\tau\tau^T + K \right)^{-1} \tau$$
[4.5]

The term $(n-1)E^{-1}\tau\tau^{T}$ added to the Hessian matrix K contains the current energy value and gradient information of the energy landscape and the n value basically determines how much weight is given on this term. The higher the n value, the higher the weight.

Instead of K, the QNA-XEL (*Broyden's method*) method uses the estimated Hessian matrix \hat{K} , which is recursively estimated from the previous Hessian matrix and the change of the energy gradients. Giving more weight to the $(n-1)E^{-1}\tau\tau^{T}$ term gives more weight to the current information. This helps when the estimation of Kis not very accurate, which occurs at the beginning of the minimization or during the optimization of a bumpy energy landscape. Another good aspect of raising the weight or increasing n is that the step size is reduced. Forcing the optimization to move slower improves the accuracy of Hessian matrix estimation because the Hessian matrix does not vary as much from step to step. This is supported by simulation results from the Newton-XEL, the QNA-XEL, and the BFGS-XEL as they show that when n > 1 the quality of resulted configurations is better than when n < 1.

4.2.6.2 The QNA-XEL and the BFGS-XEL Hessian matrix and Hessian matrix inverse estimation

The value of n affects the calculation of the estimated Hessian matrix \hat{K}^* and the estimated inverse of the Hessian matrix \hat{H}^* in the QNA-XEL and the BFGS-XEL because $\Delta \tau_{k-1}^* = \beta_k \tau_k - \beta_{k-1} \tau_{k-1}$ (4.8) and $\beta = n |E|^{n-1}$. The terms β_k and β_{k-1} are weights between the current (k^{th} iteration) and the previous ((k-1)th iteration) information of the energy landscape. To give more weight to the current landscape information, n must be chosen so that $\beta_k > \beta_{k-1}$.

When n > 1 the exponent n - 1 is positive and therefore $\beta_k > \beta_{k-1}$ if $|E_k| > |E_{k-1}| > 1$. This condition is satisfied when $E_k > E_{k-1} > 1$ or $E_k < E_{k-1} < -1$. Thus, at an increasing positive E or a decreasing negative E, n > 1 should be used. The opposite happens when n < 1; the exponent n - 1 is negative and therefore $\beta_k > \beta_{k-1}$ if $|E_{k-1}| > |E_k| > 1$. This condition is satisfied when $E_{k-1} > E_k > 1$ or $E_{k-1} < E_k < -1$. Thus, at a decreasing positive E or an increasing negative E, n < 1 should be used.

The described weighting however has possible undesirable outcomes. First, it requires constant monitoring of the landscape condition which perhaps increases computational cost. Second, in a rugged energy landscape, it results in changing n back and forth which may cause instability. Lastly, giving more weight to the current information with an increasing energy it rewards the optimization path to the nondescending direction.

To avoid these outcomes, another weighting gives more weight to a point with lower energy. This is, $\beta_k > \beta_{k-1}$ when $E_k < E_{k-1}$ or $\beta_k < \beta_{k-1}$ when $E_k > E_{k-1}$.

At the beginning of protein prediction simulation the energy values are higher and usually positive. If $1 < E_k < E_{k-1}$, $|E_k| < |E_{k-1}|$ and for $\beta_k > \beta_{k-1}$ the exponent n-1 must be negative so n < 1 should be used. This applies to the case of $1 < E_{k-1} < E_k$ as well. On the contrary, at the end of the simulation the energy values are lower and usually negative. If $E_k < E_{k-1} < -1$, $|E_k| > |E_{k-1}|$ and for $\beta_k > \beta_{k-1}$ the exponent n-1 must be positive so n > 1 should be used. This applies to the case of $E_{k-1} < E_k < -1$ as well.

To adaptively change the value of n, the effects of XEL on the Hessian matrix estimation suggest to use n < 1 at the beginning of the simulation (E > 0) and n > 1near the end of the simulation (E < 0). This agrees with a previous assumption in Section 4.1 that n < 1 should be implemented during the beginning of the simulation because it flattens the landscape and n > 1 should be implemented near the end of the simulation because it makes the global minimum more prominent.

4.2.7 Conclusion for Discussion on XEL

Section 4.2 has discussed the effects of XEL on several quantities and has suggested guidelines for changing n. First, the positive definite property of the Newton-XEL Hessian matrix is discussed. The conditions of n and energy values that make the matrix positive definite are given. Then the Newton-XEL step is shown to be the scalar multiple of the Newton step ($\hat{h}^* = \lambda \hat{h}$). That is, they are parallel. Moreover, the scalar λ is an eigenvalue of a matrix $\beta K (K^*)^{-1}$ which leads to a proof that λ is inversely related to n. Next, the error residue of the Newton-XEL is found to be of the same order as that of the Newton's method which shows that the Newton-XEL can have a similar convergence property as the Newton's method. The error also suggests that $|\lambda|$ should be bounded. Lastly, the XEL method is found to give different weights on terms in the Hessian matrix and its estimation.

Guidelines for adaptively modified energy landscaping suggested in Section 4.2 are summarized in Table 4.1. For Newton-XEL and QNA-XEL, Guideline *i*. gives conditions for obtaining a positive definite XEL Hessian matrix in different energy values (Section 4.2.1) and Guideline *ii*. gives conditions for increasing or decreasing the magnitude of the Newton-XEL step in any situation (Section 4.2.4). Guideline *ii*. is expected to be applicable to BFGS-XEL methods as well. Guideline *iii*. gives a condition for improving stability and convergence in any situation and it is expected to be applicable to all methods (Section 4.2.5). Guideline *iv*. gives a condition for improving quality when \hat{K} is inaccurate (Section 4.2.6.1). Lastly, Guideline *v*. gives conditions for increasing weight on the lower energy point in different energy values (Section 4.2.6.2).

Guideline	For	Situation	Condition	Applied to
i.	Positive definite XEL Hessian matrix	$\begin{aligned} E > 0 \\ E < 0 \end{aligned}$	n > 1 $n < 1$	Newton-XEL QNA-XEL
ii.	Smaller $ \hat{h}^* $ Larger $ \hat{h}^* $	Any	Larger n Smaller n	All
iii.	Improving stability and convergence	Any	Bounded $ \lambda $	All
iv.	Improving quality of configurations	Inaccurate \hat{K}^* or \hat{H}^*	Larger n	QNA-XEL BFGS-XEL
v.	More weight on lower energy	E < 0 E > 0	n > 1 $n < 1$	QNA-XEL BFGS-XEL

Table 4.1: Guidelines for adaptively modified energy landscaping.

Most guidelines are consistent with each other, but some are not. Guideline *iii.* suggests the magnitude of λ should be less than the bounding value, which can be achieved by increasing n according to Guideline *ii.* Doing so can improve the inaccurate Hessian matrix estimation, according to Guideline *iv.* Since Guidelines *i.* and v. give conflicting conditions, one of them will be applied at a time.

4.3 Adaptive Exponential Energy Landscaping (AXEL)

Since different values of the exponent variable n affect the XEL algorithms, n should be chosen to improve performance. Section 4.3.1 derives two optimization algorithms for AXEL: the adaptive-n and the varying-n XEL algorithms. The difficulty of implementing the adaptive-n XEL algorithms is discussed which leads to the conclusion that only the implementation of the varying-n XEL algorithms is feasible and therefore presented here. Section 4.3.2 introduces and categorizes AXEL schemes to be used with the varying-n XEL or the XEL algorithms depending on the type of scheme. Together the varying-n XEL or the XEL algorithms and the AXEL schemes compose the method of AXEL referred to simply as AXEL.

4.3.1 Optimization with AXEL

The AXEL step can be derived such that 1) it updates both dihedral angles and the n value or 2) it only updates the dihedral angles. In 1) the adaptive-n XEL algorithms simultaneously calculate the updates for dihedral angles and n. In 2) the update for n is determined separately by an AXEL scheme (discussed in Section 4.3.2), and then the varying-n XEL algorithms calculate the update for the dihedral angles with the new n. Both adaptive-n and varying-n XEL algorithms are derived for Newton's method, the QNA, and the BFGS algorithm.

4.3.1.1 Adaptive-n XEL algorithms

Adaptive-n Newton-XEL

When the energy landscape is adaptively modified, the function of the modified energy depends not only on the dihedral angles but also on the n value. The Taylor series expansion for the modified energy landscape about the independent variables θ and n is

$$E^*(\theta_k + h_\theta^*, n_k + h_n^*) = E^* + \left(\frac{\partial E^*}{\partial \theta}\right)^T h_\theta^* + \left(\frac{\partial E^*}{\partial n}\right)^T h_n^* + O\left(h_\theta^{*2}\right) + O\left(h_n^{*2}\right)$$
(4.29)

where h_{θ}^* and h_n^* are changes in dihedral angles and *n* respectively. Taking the partial derivative with respect to θ of both sides gives

$$\frac{\partial E^*(\theta_k + h_{\theta}^*, n_k + h_n^*)}{\partial \theta} = \frac{\partial E^*}{\partial \theta} + \frac{\partial^2 E^*}{\partial \theta^2} h_{\theta}^* + \frac{\partial^2 E^*}{\partial \theta \partial n} h_n^* + O\left(h_{\theta}^{*2}\right) + O\left(h_n^{*2}\right)$$

Once the higher order terms are dropped, the step for dihedral angles that approximates $\frac{\partial E^*(\theta_k + h_{\theta}^*, n_k + h_n^*)}{\partial \theta} = 0$ is

$$0 = \frac{\partial E^*}{\partial \theta} + \frac{\partial^2 E^*}{\partial \theta^2} \hat{h}^*_{\theta} + \frac{\partial^2 E^*}{\partial \theta \partial n} \hat{h}^*_n \tag{4.30}$$

$$\hat{h}_{\theta}^{*} = -\left(\frac{\partial^{2}E^{*}}{\partial\theta^{2}}\right)^{-1} \left(\frac{\partial E^{*}}{\partial\theta} + \frac{\partial^{2}E^{*}}{\partial\theta\partial n}\hat{h}_{n}^{*}\right)$$
(4.31)

where \hat{h}_{θ}^* is the step for dihedral angles, which is the approximation of h_{θ}^* .

Since Equation (4.30) contains two unknowns another equation is needed to simultaneously solve for \hat{h}^*_{θ} and \hat{h}^*_n . This is obtained from taking the partial derivative with respect to *n* of the Taylor series expansion for $E^*(\theta_k, n_k)$, Equation (4.29),

$$\frac{\partial E^*(\theta_k + h_{\theta}^*, n_k + h_n^*)}{\partial n} = \frac{\partial E^*}{\partial n} + \frac{\partial^2 E^*}{\partial n \partial \theta} h_{\theta}^* + \frac{\partial^2 E^*}{\partial n^2} h_n^* + O\left(h_{\theta}^{*2}\right) + O\left(h_n^{*2}\right)$$

Once the higher order terms are dropped, the step for the exponent n that approximates $\frac{\partial E^*(\theta_k + h_{\theta}^*, n_k + h_n^*)}{\partial n} = 0$ is

$$0 = \frac{\partial E^*}{\partial n} + \frac{\partial^2 E^*}{\partial n \partial \theta} \hat{h}^*_{\theta} + \frac{\partial^2 E^*}{\partial n^2} \hat{h}^*_n \tag{4.32}$$

$$\hat{h}_{n}^{*} = -\left(\frac{\partial^{2}E^{*}}{\partial n^{2}}\right)^{-1} \left(\frac{\partial E^{*}}{\partial n} + \frac{\partial^{2}E^{*}}{\partial n\partial\theta}\hat{h}_{\theta}^{*}\right)$$
(4.33)

where \hat{h}_n^* is the step for n, which is the approximation of h_n^* .

To solve for h_{θ}^* and h_n^* simultaneously, stacking and rearranging Equations (4.30) and (4.32) gives

$$-\begin{bmatrix} \frac{\partial E^*}{\partial \theta} \\ \frac{\partial E^*}{\partial n} \end{bmatrix} = \begin{bmatrix} \frac{\partial^2 E^*}{\partial \theta^2} & \frac{\partial^2 E^*}{\partial \theta \partial n} \\ \frac{\partial^2 E^*}{\partial n \partial \theta} & \frac{\partial^2 E^*}{\partial n^2} \end{bmatrix} \begin{bmatrix} \hat{h}_{\theta}^* \\ \hat{h}_{n}^* \end{bmatrix}$$

Defining

$$\Theta \equiv \begin{bmatrix} \theta \\ n \end{bmatrix}$$

$$\kappa^* \equiv \begin{bmatrix} \frac{\partial^2 E^*}{\partial \theta^2} & \frac{\partial^2 E^*}{\partial \theta \partial n} \\ \frac{\partial^2 E^*}{\partial n \partial \theta} & \frac{\partial^2 E^*}{\partial n^2} \end{bmatrix}$$

$$(4.34)$$

$$\frac{\partial E^*}{\partial \Theta} \equiv \begin{bmatrix} \frac{\partial E^*}{\partial \theta} \\ \frac{\partial E^*}{\partial n} \end{bmatrix}$$

$$(4.35)$$

gives the adaptive-n Newton algorithm on AXEL

$$\frac{\partial E^*}{\partial \Theta} = \kappa^* \hat{h}_{\Theta}^*$$
$$\hat{h}_{\Theta}^* = -(\kappa^*)^{-1} \frac{\partial E^*}{\partial \Theta}$$
(4.36)

where
$$\hat{h}_{\Theta}^* \equiv \begin{bmatrix} \hat{h}_{\theta}^* \\ \hat{h}_n^* \end{bmatrix}$$
 is the adaptive-*n* Newton step on AXEL.
 ∂E^*

The terms in the matrix κ^* and the vector $\frac{\partial E^*}{\partial \Theta}$ are found from Equations (4.3) and (4.4),

$$\frac{\partial E^*}{\partial \theta} = -\tau^* = -\beta\tau \tag{4.3}$$

and

$$\frac{\partial^2 E^*}{\partial \theta^2} = K^* = \beta \left((n-1)E^{-1}\tau\tau^T + K \right)$$
[4.4]

The term $\frac{\partial^2 E^*}{\partial n \partial \theta}$, which is equivalent to $\frac{\partial^2 E^*}{\partial \theta \partial n}$, is derived by taking the partial derivative of $\frac{\partial E^*}{\partial \theta}$ with respect to n,

$$\frac{\partial^2 E^*}{\partial n \partial \theta} = \frac{\partial}{\partial n} \frac{\partial E^*}{\partial \theta} = \frac{\partial (-\beta \tau)}{\partial n}$$

Since $\beta = n |E|^{n-1}$ and τ does not depend on n,

$$\frac{\partial^2 E^*}{\partial n \partial \theta} = -\frac{\partial (n |E|^{n-1})}{\partial n} \tau$$
$$= -\left(|E|^{n-1} + n \frac{\partial (|E|^{n-1})}{\partial n} \right) \tau$$
$$= -\left(|E|^{n-1} + n |E|^{n-1} \ln |E| \right) \tau$$
$$= -\left(1 + n \ln |E| \right) |E|^{n-1} \tau$$
(4.37)

The terms $\frac{\partial E^*}{\partial n}$ and $\frac{\partial^2 E^*}{\partial n^2}$ follow as,

$$\frac{\partial E^*}{\partial n} = \frac{\partial \left(E \left| E \right|^{n-1} \right)}{\partial n}$$
$$= E \cdot \ln |E| \cdot |E|^{n-1}$$
(4.38)

and

$$\frac{\partial^2 E^*}{\partial n^2} = \frac{\partial}{\partial n} \left(\frac{\partial E^*}{\partial n} \right)$$
$$= \frac{\partial \left(E \cdot \ln |E| \cdot |E|^{n-1} \right)}{\partial n}$$
$$= E \left(\ln |E| \right)^2 |E|^{n-1}$$
(4.39)

Therefore, substituting Equations (4.4), (4.37), and (4.39) into (4.34) and (4.35) yields

$$\kappa^* = \begin{bmatrix} \beta \left((n-1)E^{-1}\tau\tau^T + K \right) & -(1+n\ln|E|) |E|^{n-1}\tau \\ -(1+n\ln|E|) |E|^{n-1}\tau^T & E(\ln|E|)^2 |E|^{n-1} \end{bmatrix}$$
(4.40)

and

$$\frac{\partial E^*}{\partial \Theta} = \begin{bmatrix} -\beta \tau \\ E \cdot \ln |E| \cdot |E|^{n-1} \end{bmatrix}$$
(4.41)

Equations (4.36), (4.40), and (4.41) define the adaptive-*n* Newton-XEL step.

Adaptive-n QNA-XEL

Two different algorithms for adaptive-n QNA-XEL can be obtained by estimating two matrices that define the Hessian matrix. Similar to the adaptive-n Newton-XEL, the first adaptive-n QNA-XEL is given by

$$\hat{h}_{\Theta}^* = -\left(\hat{\kappa}^*\right)^{-1} \frac{\partial E^*}{\partial \Theta} \tag{4.42}$$

and

$$\hat{\kappa}^* = \begin{bmatrix} \beta \left((n-1)E^{-1}\tau\tau^T + \hat{K} \right) & -(1+n\ln|E|) |E|^{n-1}\tau \\ -(1+n\ln|E|) |E|^{n-1}\tau^T & E(\ln|E|)^2 |E|^{n-1} \end{bmatrix}$$
(4.43)

where $\hat{\kappa}^*$ is an estimation of κ^* and \hat{K} is an estimation of K which was previously described by the Broyden's method as

$$\hat{K}_{k} = \hat{K}_{k-1} + \left(\hat{h}_{\theta \, k-1}^{*} \, \hat{h}_{\theta \, k-1}^{*}\right)^{-1} \left(\Delta \tau_{k-1} - \hat{K}_{k-1} \, \hat{h}_{\theta \, k-1}^{*}\right) \, \hat{h}_{\theta \, k-1}^{*} \qquad [3.7]$$

Note that $\hat{h}_{\theta k}^*$ here is equal to \hat{h}_k in Equation (3.7).

The Broyden-XEL Equation (4.9) can also be used to calculate \hat{K}^* but the $\Delta \tau^*$ and β terms have to be similarly redefined,

$$\hat{K}_{k}^{*} = \hat{K}_{k-1}^{*} + \left(\hat{h}_{\theta \ k-1}^{*} \ \hat{h}_{\theta \ k-1}^{*}\right)^{-1} \left(\Delta \tau^{*}{}_{k-1}^{k} - \hat{K}_{k-1}^{*} \ \hat{h}_{\theta \ k-1}^{*}\right) \hat{h}_{\theta \ k-1}^{*T}$$
(4.44)

where

$$\Delta \tau^{*k}_{k-1} = \beta_k^k \tau_k - \beta_{k-1}^k \tau_{k-1}$$
$$\beta_k^k = n_k |E_k|^{n_k - 1}$$
$$\beta_{k-1}^k = n_k |E_{k-1}|^{n_k - 1}$$

This definition of β^k and $\Delta \tau^{*k}$ uses the current value of n as if n is constant. Allowing only changes in θ to affect the \hat{K} calculation is appropriate since \hat{K} is an estimation of the stiffness matrix that is only of function of θ . Alternatively, the second adaptive-n QNA-XEL uses Broyden's method to estimate the entire κ^* matrix as

$$\hat{\kappa}_{k}^{*} = \hat{\kappa}_{k-1}^{*} - \left(\hat{h}_{\Theta k}^{*T} \hat{h}_{\Theta k}^{*}\right)^{-1} \left(\frac{\partial E^{*}(\Theta_{k})}{\partial \Theta} - \frac{\partial E^{*}(\Theta_{k-1})}{\partial \Theta} + \hat{\kappa}_{k-1}^{*} \hat{h}_{\Theta k}^{*}\right) \hat{h}_{\Theta k}^{*T} \quad (4.45)$$

However, estimating only \hat{K} and calculating other terms individually may give a more accurate estimation of κ^* .

Adaptive-n BFGS-XEL

Similar to previous BFGS algorithms, the adaptive-*n* BFGS-XEL algorithm estimates $(\kappa^*)^{-1}$.

$$\hat{h}_{\Theta k}^{*} = \hat{H}_{k}^{\kappa} \frac{\partial E^{*} (\Theta_{k})}{\partial \Theta}$$

$$\gamma_{k-1}^{*} = \frac{\partial E^{*} (\Theta_{k})}{\partial \Theta} - \frac{\partial E^{*} (\Theta_{k-1})}{\partial \Theta}$$

$$\hat{h}_{\Theta k-1}^{*} = \Theta_{k} - \Theta_{k-1}$$

$$(4.46)$$

$$\hat{H}_{k}^{\kappa} = \hat{H}_{k-1}^{\kappa} - \left(\frac{\hat{H}_{k-1}^{\kappa}\gamma_{k-1}^{*}\hat{h}_{\Theta \ k-1}^{*} + \hat{h}_{\Theta \ k-1}^{*}\gamma_{k-1}^{*}\hat{H}_{k-1}^{\kappa}}{\hat{h}_{\Theta \ k-1}^{*}\gamma_{k-1}^{*}}\right) + \left(1 + \frac{\gamma_{k-1}^{*}\hat{H}_{k-1}^{\kappa}\gamma_{k-1}^{*}}{\hat{h}_{\Theta \ k-1}^{*}\gamma_{k-1}^{*}}\right)\frac{\hat{h}_{\Theta \ k-1}^{*}\hat{h}_{\Theta \ k-1}^{*}\gamma_{k-1}^{*}}{\hat{h}_{\Theta \ k-1}^{*}\gamma_{k-1}^{*}}$$
(4.47)

where \hat{H}^{κ} is the estimation of $(\kappa^*)^{-1}$.

Discussion on adaptive-n algorithms

In addition to the dihedral angles, all adaptive-n XEL algorithms introduce the variable n for optimization. This additional variable inevitably increases the complexity of the problem. Unlike the algorithms with a fixed n, the optimization in adaptive-nXEL algorithms is done on an adaptively modified energy landscape. This means the value of the modified energy or the cost function is driven by not only the dihedral angles but also by changes in n. In fact, the change in n can have a larger impact on the energy value than changes in the dihedral angles. For n > 1, increasing n deepens the valleys on the energy landscape. Therefore, while the modified energy value appears to be reduced because of changing n, the dihedral angles may be varied only slightly, or not at all. This presents an optimization difficulty, which is seen in some preliminary simulations implementing these adaptive-n algorithms. Results from these simulations show that the optimization paths sometimes move with very small steps and occasionally diverge.

Although an adaptive algorithm for determining n value is desired, simultaneously optimizing n and dihedral angles may not be a good solution. The next section derives the varying-n XEL algorithms that determine the change in n and then separately determine the changes in the dihedral angles.

4.3.1.2 Varying-n XEL algorithms

Since the protein conformation prediction problem is already difficult, the strategic goal of the varying-n XEL algorithms is not to aggravate the complexity of the problem while taking into account the effect of the changing n. The n value is changed according to different AXEL schemes discussed in Section 4.3.2.

Varying-n Newton-XEL

Consider the step for dihedral angles that approximates $\frac{\partial E^*(\theta_k + h_{\theta}^*, n_k + h_n^*)}{\partial \theta} = 0,$

$$\hat{h}_{\theta}^{*} = -\left(\frac{\partial^{2}E^{*}}{\partial\theta^{2}}\right)^{-1} \left(\frac{\partial E^{*}}{\partial\theta} + \frac{\partial^{2}E^{*}}{\partial n\partial\theta}\hat{h}_{n}^{*}\right)$$

$$[4.31]$$

Clearly, when n is constant or $\hat{h}_n^* = 0$, this step becomes the Newton-XEL step in Equation (4.5). Although E^* is now a function of variables n and θ , the locations of the minima where $\frac{\partial E^*}{\partial \theta} = 0$ remain the same. Therefore, \hat{h}_{θ}^* can be used to locate the minimum on the unmodified energy landscape $\frac{\partial E}{\partial \theta} = 0$. Substituting all necessary terms into Equation (4.31) yields the varying-*n* Newton-XEL step,

$$\hat{h}_{\theta}^{*} = -\left(\beta\left((n-1)E^{-1}\tau\tau^{T}+K\right)\right)^{-1}\left(-\beta\tau-(1+n\ln|E|)|E|^{n-1}\tau\hat{h}_{n}^{*}\right)$$

$$= \beta^{-1}\left((n-1)E^{-1}\tau\tau^{T}+K\right)^{-1}\cdot\beta\cdot\left(1+\left(\frac{1}{n}+\ln|E|\right)\hat{h}_{n}^{*}\right)\tau$$

$$= \left((n-1)E^{-1}\tau\tau^{T}+K\right)^{-1}\left(1+\left(\frac{1}{n}+\ln|E|\right)\hat{h}_{n}^{*}\right)\tau$$

$$= (K^{*})^{-1}\left(1+\left(\frac{1}{n}+\ln|E|\right)\hat{h}_{n}^{*}\right)\tau^{*}$$
(4.48)

Compared to the Newton-XEL step \hat{h}^* (4.5), the varying-*n* Newton-XEL step in (4.48) has an additional term, $\left(\frac{1}{n} + \ln |E|\right)\hat{h}_n^*$, which reflects the change in the energy landscape due to the changing *n*. Since $1 + \left(\frac{1}{n} + \ln |E|\right)\hat{h}_n^*$ is a scalar, the varying-*n* Newton-XEL step is a scalar multiple of the Newton-XEL step. Recalling Equation (4.19) gives

$$\hat{h}_{\theta}^{*} = \left(1 + \left(\frac{1}{n} + \ln|E|\right)\hat{h}_{n}^{*}\right)\hat{h}^{*}$$
$$= \left(1 + \left(\frac{1}{n} + \ln|E|\right)\hat{h}_{n}^{*}\right)\lambda\hat{h}$$
$$= \lambda^{\Theta}\hat{h}$$

where $\lambda^{\Theta} \equiv \left(1 + \left(\frac{1}{n} + \ln |E|\right) \hat{h}_n^*\right) \lambda$ is a scalar multiplier of the Newton-XEL with Θ as an independent variable. Since the varying-*n* Newton-XEL step is a scalar multiple of the Newton step, its error residue can be shown to be of the same order as Newton error residue. Similar to a guideline that limits $|\lambda|$ (4.28), $|\lambda^{\Theta}| < \delta$ can be followed.

Varying-n QNA-XEL

Similarly, the varying-n QNA-XEL step is defined as

$$\hat{h}_{\theta}^{*} = \left((n-1)E^{-1}\tau\tau^{T} + \hat{K} \right)^{-1} \left(1 + \left(\frac{1}{n} + \ln|E| \right) \hat{h}_{n}^{*} \right) \tau = \left(\hat{K}^{*} \right)^{-1} \left(1 + \left(\frac{1}{n} + \ln|E| \right) \hat{h}_{n}^{*} \right) \tau^{*}$$
(4.49)

where \hat{K}^* is calculated using the redefined Broyden-XEL Equation (4.44). Again, the varying-*n* QNA-XEL step has the additional term $\left(\frac{1}{n} + \ln |E|\right)\hat{h}_n^*$ compared to the QNA-XEL step.

Varying-n BFGS-XEL

The varying-n BFGS-XEL step is defined as

$$\hat{h}_{\theta}^{*} = \hat{H}^{*} \left(1 + \left(\frac{1}{n} + \ln |E| \right) \hat{h}_{n}^{*} \right) \tau^{*}$$
(4.50)

where \hat{H}^* is the estimation of $\left(\frac{\partial^2 E^*}{\partial \theta^2}\right)^{-1}$ previously defined in Equation (4.11). For the varying-*n* case the β and $\Delta \tau^*$ terms have to be redefined as,

$$\hat{h}_{\theta k-1}^* = \theta_k - \theta_{k-1}$$
$$\beta_k^k = n_k \cdot |E_k|^{n_k - 1}$$
$$\beta_{k-1}^k = n_k \cdot |E_{k-1}|^{n_k - 1}$$
$$\Delta \tau^*_{k-1}^k = \beta_k^k \tau_k - \beta_{k-1}^k \tau_{k-1}$$

$$\hat{H}_{k}^{*} = \hat{H}_{k-1}^{*} - \left(\frac{\hat{H}_{k-1}^{*} \Delta \tau^{*}{}_{k-1}^{k} \hat{h}_{\theta \ k-1}^{*} + \hat{h}_{\theta \ k-1}^{*} \Delta \tau^{*}{}_{k-1}^{k}}{\hat{h}_{\theta \ k-1}^{*} \Delta \tau^{*}{}_{k-1}^{k}}\right) + \left(\frac{\Delta \tau^{*}{}_{k-1}^{k} \hat{H}_{k-1}^{*} \Delta \tau^{*}{}_{k-1}^{k}}{\hat{h}_{\theta \ k-1}^{*} \Delta \tau^{*}{}_{k-1}^{k}} - 1\right) \frac{\hat{h}_{\theta \ k-1}^{*} \hat{h}_{\theta \ k-1}^{*}}{\hat{h}_{\theta \ k-1}^{*} \Delta \tau^{*}{}_{k-1}^{k}} - 1\right) (4.51)$$

Again, compared to the BFGS-XEL, the varying-*n* BFGS-XEL contains the additional term $\left(\frac{1}{n} + \ln |E|\right) \hat{h}_n^*$. It turns out that this term is very important to protein predictions that use a varying *n* during the minimization process. In fact, a comparative study shows that when *n* is not constant the varying-*n* BFGS-XEL can locate a conformation with up to 9% lower energy than that located by the BFGS-XEL.

Figures 4.6, 4.7, and 4.8 give pseudo-codes for the varying-n Newton-XEL, the varying-n QNA-XEL, and the varying-n BFGS-XEL, respectively, used with a line search and an AXEL scheme.

Pseudo-code: Varying-n Newton-XEL with a line search

Initialize ϵ , n_1 , and θ_1 Calculate E_0 for $k = 1, ..., k_{max}$ do

Calculate E_k , τ_k , and K_k if $\left| \frac{E_k - E_{k-1}}{E_{k-1}} \right| < \epsilon$ then

break // Convergence criterion met.

end if

 $n_{k+1} = \texttt{AXEL_scheme}() // \text{Determine the new exponent with an AXEL scheme}$

$$\begin{split} \hat{h}_{n\,k}^{*} &= n_{k+1} - n_{k} \\ \beta_{k} &= n_{k} |E_{k}|^{n_{k}-1} \\ K_{k}^{*} &= \beta_{k} \left((n_{k} - 1)E_{k}^{-1}\tau_{k}\tau_{k}^{T} + K_{k} \right) \\ \tau_{k}^{*} &= \beta_{k}\tau_{k} \\ \hat{h}_{\theta\,k}^{*} &= (K_{k}^{*})^{-1} \left(1 + \left(\frac{1}{n_{k}} + \ln |E_{k}| \right) \hat{h}_{n\,k}^{*} \right) \tau_{k}^{*} \\ // \text{ Determine the varying-} n \text{ Newton-XEL step} \\ \alpha_{k} &= \texttt{line_search}(\hat{h}_{\theta\,k}^{*}) // \text{ Determine the line search gain} \\ \theta_{k+1} &= \theta_{k} + \alpha_{k} \hat{h}_{\theta\,k}^{*} // (k+1)^{\text{th}} \text{ solution} \end{split}$$

end for

Figure 4.6: A pseudo-code for varying-*n* Newton-XEL with a line search. Note that for varying-*n* Newton-XEL (*Hessian only*) $\tau_k^* = \tau_k$.

Initialize ϵ , n_1 , θ_0 , θ_1 , and \hat{K}_0^*

Calculate E_0 , and τ_0

for $k = 1, ..., k_{max}$ do

Calculate E_k , and τ_k

if
$$\left|\frac{E_k - E_{k-1}}{E_{k-1}}\right| < \epsilon$$
 then

break // Convergence criterion met.

end if

 $n_{k+1} = \text{AXEL_scheme}() // \text{Determine the new exponent with an AXEL scheme}$

$$\hat{h}_{n\,k}^* = n_{k+1} - n_k$$

Calculate \hat{K}_k^*

$$\hat{h}_{\theta k-1}^{*} = \theta_{k} - \theta_{k-1}$$

$$\beta_{k}^{k} = n_{k} \cdot |E_{k}|^{n_{k}-1}$$

$$\beta_{k-1}^{k} = n_{k} \cdot |E_{k-1}|^{n_{k}-1}$$

$$\Delta \tau^{*}{}_{k-1}^{k} = \beta_{k}^{k} \tau_{k} - \beta_{k-1}^{k} \tau_{k-1}$$

$$\hat{K}_{k}^{*} = \hat{K}_{k-1}^{*} + \left(\hat{h}_{\theta k-1}^{*T} \hat{h}_{\theta k-1}^{*}\right)^{-1} \left(\Delta \tau^{*}{}_{k-1}^{k} - \hat{K}_{k-1}^{*} \hat{h}_{\theta k-1}^{*}\right) \hat{h}_{\theta k-1}^{*T}$$

$$\tau_k^* = \beta_k^k \tau_k$$
$$\hat{h}_{\theta k}^* = \left(\hat{K}_k^*\right)^{-1} \left(1 + \left(\frac{1}{n_k} + \ln|E_k|\right) \hat{h}_{n k}^*\right) \tau_k^*$$

// Determine the varying-*n* QNA-XEL step $\alpha_k = \texttt{line_search}(\hat{h}^*_{\theta k}) // \text{Determine the line search gain}$ $\theta_{k+1} = \theta_k + \alpha_k \hat{h}^*_{\theta k} // (k+1)^{\text{th}} \text{ solution}$

end for

Figure 4.7: A pseudo-code for varying-*n* QNA-XEL with a line search. Note that for varying-*n* QNA-XEL (*Hessian only*) $\tau_k^* = \tau_k$.

Pseudo-code: Varying-*n* BFGS-XEL with a line search

Initialize ϵ , n_1 , θ_0 , θ_1 , and \hat{H}_0^*

Calculate E_0 and τ_0

for $k = 1, ..., k_{max}$ do

Calculate E_k , and τ_k

if
$$\left|\frac{E_k - E_{k-1}}{E_{k-1}}\right| < \epsilon$$
 then

break // Convergence criterion met.

end if

 $n_{k+1} = \text{AXEL_scheme}() // \text{Determine the exponent with an AXEL scheme}$ $\hat{h}_{n\,k}^* = n_{k+1} - n_k$

Calculate \hat{H}_k^*

$$\begin{split} \hat{h}_{\theta \, k-1}^{*} &= \theta_{k} - \theta_{k-1} \\ &\beta_{k}^{k} = n_{k} \cdot |E_{k}|^{n_{k}-1} \\ &\beta_{k-1}^{k} = n_{k} \cdot |E_{k-1}|^{n_{k}-1} \\ &\Delta \tau^{*}{}_{k-1}^{k} = \beta_{k}^{k} \tau_{k} - \beta_{k-1}^{k} \tau_{k-1} \\ &\hat{H}_{k}^{*} = \hat{H}_{k-1}^{*} - \left(\frac{\hat{H}_{k-1}^{*} \Delta \tau^{*}{}_{k-1}^{k} \hat{h}_{\theta \, k-1}^{*} + \hat{h}_{\theta \, k-1}^{*} \Delta \tau^{*}{}_{k-1}^{k}}{\hat{h}_{\theta \, k-1}^{*} \Delta \tau^{*}{}_{k-1}^{k}} - 1 \right) \frac{\hat{h}_{\theta \, k-1}^{*} \hat{h}_{\theta \, k-1}^{*}}{\hat{h}_{\theta \, k-1}^{*} \Delta \tau^{*}{}_{k-1}^{k}} - 1 \\ &+ \left(\frac{\Delta \tau^{*}{}_{k-1}^{k} \hat{H}_{k-1}^{*} \Delta \tau^{*}{}_{k-1}^{k}}{\hat{h}_{\theta \, k-1}^{*} \Delta \tau^{*}{}_{k-1}^{k}} - 1 \right) \frac{\hat{h}_{\theta \, k-1}^{*} \hat{h}_{\theta \, k-1}^{*} \Delta \tau^{*}{}_{k-1}^{k}}{\hat{h}_{\theta \, k-1}^{*} \Delta \tau^{*}{}_{k-1}^{k}} \\ &\tau_{k}^{*} = \beta_{k}^{k} \tau_{k} \\ \hat{h}_{\theta \, k}^{*} &= \hat{H}_{k}^{*} \left(1 + \left(\frac{1}{n_{k}} + \ln |E_{k}| \right) \hat{h}_{n \, k}^{*} \right) \tau_{k}^{*} \\ // \text{ Determine the varying-} n \text{ BFGS-XEL step} \\ &\alpha_{k} = \texttt{line_search}(\hat{h}_{\theta \, k}^{*}) // \text{ Determine the line search gain} \end{split}$$

 $\theta_{k+1} = \theta_k + \alpha_k \hat{h}^*_{\theta k} / (k+1)^{\text{th}}$ solution

end for

Figure 4.8: A pseudo-code for varying-*n* BFGS-XEL with a line search. Note that for varying-*n* BFGS-XEL (*Hessian only*) $\tau_k^* = \tau_k$.

4.3.2 AXEL Schemes

The simulation of conformation prediction typically consists of two parts, probabilistic search and minimization. AXEL schemes can be categorized into two groups based on when n changes. Some of these schemes change n during the minimization, so they are applicable to varying-n XEL algorithms. Others change n less frequently outside the minimization stage and therefore they are more applicable to XEL algorithms.

- I. Global: n changes depending on the global location of the probabilistic search which is the searched location with respect to the global solution. For example, as the search progresses the searched location is expected to be closer to the global solution so larger n may be needed to improve quality. These changes of n happen during the course of the simulation, but n's are constant during each minimization step. XEL algorithms are used in minimization.
 - *i.* **Periodic:** *n* changes at specific times. The performance of the XEL simulations are compared to determine which *n* performs the best during the beginning, the middle, and the end of the simulation. The AXEL scheme changes the *n* value accordingly.
- II. Local: n changes depending on the local location of the search. These changes happen during the course of each minimization step. Varying-n XEL algorithms are used in minimization.
 - *i.* **Periodic:** *n* changes at specific times.
 - *ii.* Occasional: *n* changes when criteria are met.

Global and local schemes are implemented so that the global and local effects of the AXEL on the prediction simulation can be studied. While global schemes are only applicable to a protein prediction algorithm that involves several minimization steps for many starting configurations, local schemes are applicable to an algorithm with one minimization step as well.

4.4 Conclusion for the Energy Landscaping

The method of XEL has been presented and discussed, and five guidelines on how to change the landscape are suggested. The derivations of Newton's method, the QNA, and the BFGS method for XEL and AXEL have been presented. Lastly, the AXEL schemes used with the varying-n XEL algorithms are categorized. The method of adaptively modified energy landscape comprises of the varying-n XEL or the XEL algorithms and the AXEL schemes. Tables 4.2 through 4.5 summarize all presented methods from Chapters 3 and 4. Note that in some equations subscript k is omitted for displaying purposes.

Method	Equation	
Newton	$\hat{h}_k = K_k^{-1} \tau_k$	[3.5]
	$K_k^* = \beta_k \left((n-1)E_k^{-1}\tau_k \tau_k^T + K_k \right)$	[4.4]
Newton-XEL	$\tau_k^* \equiv \beta_k \tau_k \text{(or } \tau_k^* \equiv \tau_k, \text{ Hessian only)}$ $\hat{h}_k^* = (K_k^*)^{-1} \tau_k^*$	[4.5]
	$\hat{K}_{k} = \hat{K}_{k-1} + \left(\hat{h}_{k-1}^{T}\hat{h}_{k-1}\right)^{-1}$	
QNA	$\cdot \left(\Delta au_{k-1} - \hat{K}_{k-1} \hat{h}_{k-1} ight) \hat{h}_{k-1}^T$	[3.7]
	$\hat{h}_k = \hat{K}_k^{-1} \tau_k$	[3.6]
	$\hat{K}_{k}^{*} = \hat{K}_{k-1}^{*} + \left(\hat{h}_{k-1}^{*}{}^{T}\hat{h}_{k-1}^{*}\right)^{-1}$	
ONA-XEL	$\cdot \left(\Delta \tau_{k-1}^* - \hat{K}_{k-1}^* \hat{h}_{k-1}^* \right) \hat{h}_{k-1}^* $	[4.9]
QIA-ADD	$\tau_k^* \equiv \beta_k \tau_k$ (or $\tau_k^* \equiv \tau_k$, Hessian only)	
	$\hat{h}_k^* = \left(\hat{K}_k^*\right)^{-1} \tau_k^*$	[4.6]
	$\hat{K}_{k} = \hat{K}_{k-1} + \left(\hat{h}_{k-1}^{*} \hat{h}_{k-1}^{*}\right)^{-1}$	
QNA-XEL	$\cdot \left(\Delta \tau_{k-1} - \hat{K}_{k-1} \ \hat{h}_{k-1}^* \right) \ \hat{h}_{k-1}^*^T$	[3.7]
(Broyden's method)	$\hat{K}_k^* = \beta_k (n-1) E_k^{-1} \tau_k \tau_k^T + \hat{K}_k$	[4.7]
	$\hat{h}_k^* = \left(\hat{K}_k^*\right)^{-1} \tau_k^*$	[4.6]

Table 4.2: A summary of presented methods: Newton, Newton-XEL, QNA, and QNA-XEL (1 of 4)

Table 4.3: A summary of presented methods: BFGS and BFGS-XEL (2 of 4)

Equation (cont.)	$\hat{H}_{k} = \hat{H}_{k-1} - \left(\frac{\hat{H}_{k-1}\Delta\tau_{k-1}\hat{h}_{k-1}^{T} + \hat{h}_{k-1}\Delta\tau_{k-1}^{T}\hat{H}_{k-1}}{\hat{h}_{k-1}^{T}\Delta\tau_{k-1}}\right) + \left(\frac{\Delta\tau_{k-1}^{T}\hat{H}_{k-1}\Delta\tau_{k-1}}{\hat{h}_{k-1}^{T}\Delta\tau_{k-1}} - 1\right)\frac{\hat{h}_{k-1}\hat{h}_{k-1}^{T}}{\hat{h}_{k-1}^{T}\Delta\tau_{k-1}} $ [3.8] $\hat{h}_{k} = \hat{H}_{k}\tau_{k} $ [3.9]	$\begin{split} \hat{H}_{k}^{*} &= \hat{H}_{k-1}^{*} - \left(\frac{\hat{H}_{k-1}^{*} \Delta \tau_{k-1}^{*} \hat{h}_{k-1}^{*} - \Delta \tau_{k-1}^{*} \hat{H}_{k-1}^{*}}{\hat{h}_{k-1}^{*} \Delta \tau_{k-1}^{*}} \right) \\ &+ \left(\frac{\hat{\Lambda}_{k-1}^{*} T \Delta \tau_{k-1}^{*}}{\hat{h}_{k-1}^{*} \Delta \tau_{k-1}^{*}} - 1 \right) \frac{\hat{h}_{k-1}^{*} \hat{h}_{k-1}^{*} \hat{h}_{k-1}^{*}}{\hat{h}_{k-1}^{*} \Delta \tau_{k-1}^{*}} \end{split} $ [4.11] $\tau_{k}^{*} \equiv \beta_{k} \tau_{k} (\text{or } \tau_{k}^{*} \equiv \tau_{k}, \text{Hessian only}) \\ \hat{h}_{k}^{*} = \hat{H}_{k}^{*} \tau_{k}^{*} \end{split}$
Method (cont.)	BFGS	BFGS-XEL
Method (cont.)	Equation (cont.)	
--	--	-------------------
Adaptive- <i>n</i> Newton-	$\kappa^* = \begin{bmatrix} \beta \left((n-1)E^{-1}\tau\tau^T + K \right) & -(1+n\ln E) E ^{n-1}\tau \\ -(1+n\ln E) E ^{n-1}\tau^T & E(\ln E)^2 E ^{n-1} \end{bmatrix}$	[4.40]
XEL	$\hat{h}_{\Theta}^{*} = -\left(\kappa^{*}\right)^{-1} \frac{\partial E^{*}}{\partial \Theta}$	[4.36]
	$\hat{\kappa}^* = \begin{bmatrix} \beta \left((n-1)E^{-1}\tau\tau^T + \hat{K} \right) & -(1+n\ln E) E ^{n-1}\tau \\ -(1+n\ln E) E ^{n-1}\tau^T & E(\ln E)^2 E ^{n-1} \end{bmatrix}$	[4.43]
Adaptive-n	$a) \hat{K}_{k} = \hat{K}_{k-1} + \left(\hat{h}_{\theta \ k-1}^{*T} \hat{h}_{\theta \ k-1}^{*}\right)^{-1} \left(\Delta \tau_{k-1} - \hat{K}_{k-1} \hat{h}_{\theta \ k-1}^{*}\right) \hat{h}_{\theta}^{*}$	$_{k-1}^{T}[3.7]$
QNA-XEL	b) $\hat{K}_{k}^{*} = \hat{K}_{k-1}^{*} + \left(\hat{h}_{\theta \ k-1}^{*} \ \hat{h}_{\theta \ k-1}^{*}\right)^{-1}$	
(a) Broyden's method for \hat{K} (b) Redefined	$\cdot \left(\Delta \tau^{*}{}^{k}_{k-1} - \hat{K}^{*}_{k-1} \hat{h}^{*}_{\theta k-1} \right) \hat{h}^{*}_{\theta k-1}^{T}$	[4.44]
Broyden-XEL c) Broyden's method for $\hat{\kappa}^*$	c) $\hat{\kappa}_{k}^{*} = \hat{\kappa}_{k-1}^{*} - \left(\hat{h}_{\Theta_{k}}^{*} \hat{h}_{\Theta_{k}}^{*}\right)^{-1}$	
	$\left(\frac{\partial E^*\left(\Theta_k\right)}{\partial \Theta} - \frac{\partial E^*\left(\Theta_{k-1}\right)}{\partial \Theta} + \hat{\kappa}_{k-1}^* \hat{h}_{\Theta k}^*\right) \hat{h}_{\Theta k}^* \overset{T}{\rightarrow}$	[4.45]
	$\hat{h}_{\Theta}^* = -\left(\hat{\kappa}^*\right)^{-1} \frac{\partial E^*}{\partial \Theta}$	[4.42]
	$\hat{H}_{k}^{\kappa} = \hat{H}_{k-1}^{\kappa} - \left(\frac{\hat{H}_{k-1}^{\kappa}\gamma_{k-1}^{*}\hat{h}_{\Theta k-1}^{*}^{T} + \hat{h}_{\Theta k-1}^{*}\gamma_{k-1}^{*}^{T}\hat{H}_{k-1}^{\kappa}}{\hat{h}_{\Theta k-1}^{*}\gamma_{k-1}^{*}}\right)$	
Adaptive- <i>n</i> BFGS-XEL	$+ \left(1 + \frac{\gamma_{k-1}^{*} {}^{T} \hat{H}_{k-1}^{\kappa} \gamma_{k-1}^{*}}{\hat{h}_{\Theta k-1}^{*} \gamma_{k-1}^{*}}\right) \frac{\hat{h}_{\Theta k-1}^{*} \hat{h}_{\Theta k-1}^{*} {}^{T} {}^{T} {}^{K} \hat{h}_{\Theta k-1}^{*} \gamma_{k-1}^{*}}{\hat{h}_{\Theta k-1}^{*} \gamma_{k-1}^{*}}$	[4.47]
	$\hat{h}_{\Theta k}^{*} = \hat{H}_{k}^{\kappa} \frac{\partial E^{*} \left(\Theta_{k}\right)}{\partial \Theta}$	[4.46]
	where $\frac{\partial E^*}{\partial \Theta} = \begin{bmatrix} -\beta \tau \\ E \cdot \ln E \cdot E ^{n-1} \end{bmatrix}$	[4.41]

Table 4.4: A summary of presented methods: Adaptive-n XEL algorithms (3 of 4)

	[4.48]	[3.7]	[4.7]	[4.44]	[4.49]		[4.51]	[4.50]
Equation (cont.)	$\hat{h}_{\theta k}^{*} = (K_{k}^{*})^{-1} \left(1 + \left(\frac{1}{n_{k}} + \ln E_{k} \right) \hat{h}_{n k}^{*} ight) au_{k}^{*}$	a) $\hat{K}_{k} = \hat{K}_{k-1} + \left(\hat{h}_{\thetak-1}^{*T} \hat{h}_{\thetak-1}^{*}\right)^{-1} \left(\Delta \tau_{k-1} - \hat{K}_{k-1}^{*\theta} \hat{h}_{\thetak-1}^{*T}\right) \hat{h}_{\thetak-1}^{*T}$	$\hat{K}_k^* = eta_k(n-1)E_k^{-1} au_k au_k^T + \hat{K}_k$	$b) \qquad \hat{K}_{k}^{*} = \hat{K}_{k-1}^{*} + \left(\hat{h}_{\theta \ k-1}^{* \ T} \ \hat{h}_{\theta \ k-1}^{*}\right)^{-1} \left(\Delta \tau^{* \ k}_{-1} - \hat{K}_{k-1}^{*} \ \hat{h}_{\theta \ k-1}^{*}\right) \hat{h}_{\theta \ k-1}^{* \ T}$	$\hat{h}_{\thetak}^{*} = \left(\hat{K}_{k}^{*}\right)^{-1} \left(1 + \left(\frac{1}{n_{k}} + \ln E_{k} \right)\hat{h}_{nk}^{*}\right)\tau_{k}^{*}$	$\hat{H}_{k}^{*} = \hat{H}_{k-1}^{*} - \left(\frac{\hat{H}_{k-1}^{*} \ \Delta \tau^{*} {}_{k-1}^{*} \ \hat{h}_{\theta \ k-1}^{*} + \hat{h}_{\theta \ k-1}^{*} \ \Delta \tau^{*} {}_{k-1}^{*} \ \hat{H}_{k-1}^{*}}{\hat{h}_{\theta \ k-1}^{*} \ \Delta \tau^{*} {}_{k-1}^{*}} \right)$	$+ \left(\frac{\Delta \tau^* \underset{k=1}{k-1}^T \hat{H}^*_{k-1} \ \Delta \tau^* \underset{k=1}{k-1}}{\hat{h}^*_{\theta \ k-1} \ \Delta \tau^* \underset{k=1}{k-1}} - 1 \right) \frac{\hat{h}^*_{\theta \ k-1} \ \hat{h}^*_{\theta \ k-1}}{\hat{h}^*_{\theta \ k-1} \ \Delta \tau^* \underset{k=1}{k-1}}$	$\hat{h}^*_{ heta k} = \hat{H}^*_k \left(1 + \left(rac{1}{n_k} + \ln E_k ight) \hat{h}^*_{n k} ight) au^*_k$
Method (cont.)	Varying- <i>n</i> Newton-XEL	Varvino- <i>n</i>	QNA-XEL	a) Broyden's method for \hat{K} b) Redefined Broyden-XEL			Varying- <i>n</i> BFGS-XEL	

Table 4.5: A summary of presented methods: Varying-n XEL algorithms (4 of 4)

CHAPTER V

SIMULATION RESULTS AND DISCUSSIONS

To investigate the effects of the method of exponential energy landscaping (XEL) on Newton's method, the quasi-Newton algorithm (QNA), and the Broyden-Fletcher-Goldfarb-Shanno (BFGS) algorithm, several simulations are performed with two different implementations.

- (i) The investigated algorithms are implemented without a probabilistic search on a simulated two-dimensional energy landscape.
- (ii) The investigated algorithms are implemented with a probabilistic search on a multi-dimensional energy landscape.

While the variations of the QNA-XEL and the BFGS-XEL algorithms are implemented in both (i) and (ii), the variations of the Newton-XEL are only implemented in (i) because the second derivatives on a multi-dimensional energy landscape are not available in the platform used for (ii). Observing the results from the simulations in (i), adaptive exponential energy landscape (AXEL) schemes are developed and implemented in (ii).

The performance comparison of different simulations is done in terms of quality and speed evaluated through different quantities for different implementations.

- (i) For quality comparisons the converged energy and the number of paths that locate the global minimum are evaluated, and for speed comparisons the number of iterations is evaluated.
- (ii) For quality comparisons the score improvement, the lowest score, and the similarity to the native conformation are evaluated, and for speed comparisons the

number of iterations is evaluated. In addition to quality and speed, efficiency defined as a ratio between quality and speed is evaluated.

Section 5.1 presents results from the following algorithms implemented in (*i*): Newton-XEL, Newton-XEL (*Hessian only*), QNA-XEL, QNA-XEL (*Hessian only*), BFGS-XEL, and BFGS-XEL (*Hessian only*). The XEL algorithms are derived from both the first derivative and the Hessian matrix of the XEL but the XEL (*Hessian only*) algorithms are derived from the Hessian matrix only. Section 5.2 presents results from the following algorithms implemented in (*ii*): QNA-XEL, QNA-XEL (*Hessian only*), BFGS-XEL, and BFGS-XEL (*Hessian only*). Section 5.3 discusses simulations and results from AXEL. Lastly, Section 5.4 concludes the chapter.

5.1 XEL Algorithms without Probabilistic Search

To study the effect of XEL on Newton's method, QNA, and BFGS, optimization paths are generated for each XEL algorithm on two simulated two-degree-of-freedom energy landscapes: simple and complex energy landscapes. As shown in Figure 5.1, the simple energy landscape (a) is a fourth order polynomial function with only one minimum which is a global minimum and the complex landscape (b) is a parabolic based surface with multiple local minima and one global minimum. The simple energy landscape and its derivatives as functions of the dihedral angles θ are described below. The functions of the complex energy landscape and its derivatives are not displayed because of their lengths.

$$E(\theta_{1},\theta_{2}) = 0.0128(\theta_{1}^{4} + \theta_{2}^{4}) - 16$$

$$\frac{\partial E(\theta_{1},\theta_{2})}{\partial \theta} = \begin{bmatrix} 0.0512 \ \theta_{1}^{3} \\ 0.0512 \ \theta_{2}^{3} \end{bmatrix}$$

$$\frac{\partial E^{2}(\theta_{1},\theta_{2})}{\partial \theta^{2}} = \begin{bmatrix} 0.1536 \ \theta_{1}^{2} & 0 \\ 0 & 0.1536 \ \theta_{2}^{2} \end{bmatrix}$$
(5.1)

where E is the energy function and θ_i 's are dihedral angles. Note that dihedral angles exist only on a protein chain or a polypeptide and the smallest polypeptide has two residues with four degrees of freedom. Therefore, there are no dihedral angles in 2D system but torsion angles. However, these angles are called dihedral angles here for consistency.



Figure 5.1: The simple energy landscape (a) and the complex landscape (b) used to generate optimization paths for all algorithms.

A grid search is done as paths are generated for 25 different locations equally distributed on the surface shown in Figure 5.2. Seven values of n used in these simulations are 0.125, 0.25, 0.5, 1 (unmodified case), 2, 3, and 4. The range of n values is selectively investigated because it sufficiently represents the performance of the algorithms. Also, as n becomes lower than 0.125 the optimization algorithm diverges and as n becomes higher than 4, total steps become too small to reach a solution in a reasonable time.

In addition to different optimization algorithms, two line search algorithms are used to demonstrate that a line search can have a significant impact to the XEL.



25 Starting Points on the Energy Landscape

Figure 5.2: Twenty-five starting points for paths generation. Each line indicates the initial optimization step, which moves from the the dot to another end of the line.

The first implemented line search is an integrated golden section and quadratic interpolation bracketing algorithm with Brent's method [52]. This algorithm uses a bracketing algorithm combined with the golden section and quadratic interpolation to initialize a search interval and uses the Brent's method to locate the minimum. The second implemented line search is a heuristic exhaustive line search algorithm. This algorithm divides and checks the energies of a search interval and if an optimal line search gain can not be located it iteratively increases the range of the search up to five times the size of the computed step.

Two sets of the convergence criteria are used: 1) for the simple energy landscape the minimization stops when the magnitude of the gradient is less than 0.001, and 2) for the complex energy landscape the minimization stops when the magnitude of the gradient is less than 0.1. Since the simple energy landscape is very flat near the global minimum, the stricter criterion is in place so that it would allow the paths to reach the minimum. The magnitude of the gradient is chosen to be the measure of convergence because the usual measure, the relative change of the energy function between the current and the previous step, can mistakenly assume paths with a very small step as converged. As mentioned, the performance comparison of different simulations is discussed in terms of quality and speed. Before the quality and speed are fully discussed, Section 5.1.1 discusses the characteristics of paths generated by the XEL algorithms with different values of n. Section 5.1.2 discusses the quality and speed of XEL algorithms by comparing the energy to which the paths converge and the number of iterations to the unmodified case. Section 5.1.3 compares the quality in terms of the ability of the algorithms to locate a global minimum.

5.1.1 Path Comparison

To study the effects of XEL on each algorithm, paths generated by Newton-XEL, Newton-XEL (*Hessian only*), QNA-XEL, QNA-XEL (*Hessian only*), BFGS-XEL, and BFGS-XEL (*Hessian only*) are evaluated. The equations of these algorithms are summarized in Tables 4.2 and 4.3, pages 83–84.

5.1.1.1 Newton-XEL and Newton-XEL (Hessian only)

Simple energy landscape

With a bracketing and Brent line search on the simple energy landscape, Newton-XEL and Newton-XEL (Hessian only) (4.5) paths, if exist, from all cases are the same as they are able to locate the minimum in one total step. This can be observed from Figures 5.3 and 5.4 with representative examples of paths from Newton-XEL and Newton-XEL (Hessian only) respectively. Note that Figures 5.3 c) and 5.4 c) do not have a path but only the starting points. Since n affects only the magnitude of the Newton step, as described in Section 4.2.2, and the Newton step of this energy landscape is coincidently parallel to a line from the current position to the global minimum, every Newton-XEL step always points to the global minimum. Although the magnitudes of these steps are different, the bracketing and Brent line search helps locating the minimum in one total step. If the Newton step does not always point to the global minimum, the Newton-XEL steps with different magnitudes would yield

different increments and more divergent paths of the Newton-XEL would have been observed. The absence of deviation in both Newton-XEL and Newton-XEL (*Hessian only*) paths demonstrates that the effect of XEL on the Newton's method can be very limited due to the lack of directional effects of XEL on the Newton's method.

Figures 5.3 c) and 5.4 c) show that at some starting points no path is generated. This is because at least one of the second derivatives (5.1) with respect to θ_1 or θ_2 is zero and therefore the Hessian matrix is not invertible at these starting points. This happens in 9 out of 25 starting points. This demonstrates one restriction of the Newton-XEL, which is inherited from the Newton's method.

With a heuristic exhaustive line search, Newton-XEL and Newton-XEL (*Hessian* only) paths on the simple energy landscape also make a straight line to the global minimum but with multiple total steps. This can be observed from Figures 5.5 and 5.6 with representative examples of paths from Newton-XEL and Newton-XEL (*Hessian* only) respectively. While the special characteristic of the energy landscape is that straight paths are taken to the minimum, most paths are actually composed of small total steps which are different from those with the bracketing and Brent line search that have only one total step. Because the heuristic exhaustive line search preserves the effect of n on the magnitude of the step, the line search results in different gains for different values of n.

With a heuristic exhaustive line search, most Newton-XEL and Newton-XEL (*Hessian only*) paths are the same except some n = 0.125 paths from Newton-XEL (*Hessian only*) that overshoot the minimum at the first total step and locate it soon thereafter (Figure 5.6 e). This is because with the Newton-XEL (*Hessian only*) n has a greater effect on the magnitude so the n = 0.125 total step becomes much larger than that of the Newton-XEL.











f) through j) show more detail of the paths on a) through e), respectively. Paths are generated by a heuristic exhaustive line **Figure 5.5:** Paths from 1^{st} through 5^{th} starting points a) through e) on a simple energy landscape generated with Newton-XEL. search.





Complex energy landscape

With the bracketing and Brent line search on the complex energy landscape, n has little effect on paths from Newton-XEL and Newton-XEL (*Hessian only*) as they rarely differ. This can be observed from Figures 5.7 and 5.8 with representative examples of paths from Newton-XEL and Newton-XEL (*Hessian only*) respectively. Paths from Newton-XEL (*Hessian only*) show more variation than those from Newton-XEL which demonstrates that n has more effect on Newton-XEL (*Hessian only*) than Newton-XEL. This is because the extra term β^{-1} in Newton-XEL (*Hessian only*) and with the exponent n - 1 in $\beta = n |E|^{n-1}$ magnifies the dependency of n is magnified as discussed briefly in Section 4.1.4. Results here show that n < 1 yields a longer total step than n > 1 but the effects of n are not as pronounced as expected because the bracketing and Brent line search reduces the effect of n on the magnitudes of the step.

With the heuristic exhaustive line search, paths from different values of n on the complex energy landscape show more variation than those with the bracketing and Brent line search. This can be observed from Figures 5.9 and 5.10 with representative examples of paths from Newton-XEL and Newton-XEL (*Hessian only*) respectively. Again this is because the heuristic exhaustive line search does not significantly reduce the effect of n on the magnitude of the step. Paths from Newton-XEL (*Hessian only*) have larger total steps when n < 1 which allow paths to travel farther and more frequently reach the global minimum. However, with smaller total steps Newton-XEL paths tend to converge to the closest local minima.



Figure 5.7: Paths from 1^{st} through 5^{th} starting points a) through e) on a complex energy landscape generated with Newton-XEL. f) through j) show more detail of the paths on a) through e), respectively. Paths are generated by a bracketing and Brent line search.











XEL (Hessian only). f) through j) show more detail of the paths on a) through e), respectively. Paths are generated by a **Figure 5.10:** Paths from 1st through 5th starting points a) through e) on a complex energy landscape generated with Newtonheuristic exhaustive line search.

5.1.1.2 QNA-XEL and QNA-XEL (Hessian only)

Simple energy landscape

With the bracketing and Brent line search on the simple energy landscape, QNA-XEL and QNA-XEL (*Hessian only*) (4.6) paths with different values of n differ significantly from each another. This can be observed from Figures 5.11 and 5.12 with representative examples of paths from QNA-XEL and QNA-XEL (*Hessian only*) respectively. Because both algorithms use Broyden-XEL (4.9), the step is significantly varied by the different values of n. However, the total steps are not varied by the different values of n because its effect is reduced by the bracketing and Brent line search.

Figures 5.11 and 5.12 also show that paths from QNA-XEL and QNA-XEL (*Hessian only*) are identical. Even though each algorithm gives a different magnitude of the step, the bracketing and Brent line search results in the same total step. The figures also show that many paths do not converge to the minimum. The paths stop before they reach the minimum because a new step can not be calculated when the line search gives a zero line search gain. This occurs when the step is almost perpendicular to the gradient which means the line search has a very small search interval so it converges with zero gain.

With a heuristic exhaustive line search, QNA-XEL and QNA-XEL (*Hessian only*) paths differ with n. This can be observed from Figures 5.13 and 5.14 with representative examples of paths from QNA-XEL and QNA-XEL (*Hessian only*) respectively. Paths from both algorithms show that when n > 1 the paths proceed directly toward the global minimum, similar to the Newton-XEL paths. This shows that the estimation of the Hessian matrix, \hat{K}^* , is very accurate. Since both the actual modified Hessian matrix, K^* , and the initial estimation of the Hessian matrix \hat{K}_0^* for this energy function are diagonal, they are initially close. Because n > 1 paths have smaller total steps and the K^* is not varied too much along the path, it can be estimated more accurately.

Paths with a heuristic exhaustive line search also show that for $n \leq 1$ \hat{K}^* becomes inaccurate due to the larger total step at the beginning of the path. As a result, paths are sometimes almost perpendicular to the negative gradient direction and occasionally diverge. As the QNA-XEL path becomes almost parallel to the negative gradient, the total step becomes very small because of the line search limiting the gain to a point with lowest energy. This means the change in gradient becomes very small and \hat{K}^* is hardly updated. If \hat{K}^* is close to the actual K^* , the small total step may not cause any problem. However, unlike at the beginning of the n > 1 paths, \hat{K}^* here is far from the actual K^* , so it takes a long time for \hat{K}^* to converge to K^* with too small a step size. These characteristics are not observed on paths with the bracketing and Brent line search because the magnitudes of their steps are not significantly varied by the different values of n. While most paths with the bracketing and Brent line search end at a close proximity of the minimum, they have many fewer total steps compared to paths with a heuristic exhaustive line search which implies higher speed.

While paths of the QNA-XEL (*Hessian only*) do not significantly vary from those of the QNA-XEL, Figure 5.14 shows that when n = 0.125 the step can diverge. This suggests that the QNA-XEL (*Hessian only*) may be less stable than the QNA-XEL for n < 1 as the total step can become too large.



f) through j) show more detail of the paths on a) through e), respectively. Paths are generated by a bracketing and Brent line **Figure 5.11:** Paths from 1st through 5th starting points a) through e) on a simple energy landscape generated with QNA-XEL. search.



 $(Hessian \ only)$. f) through j) show more detail of the paths on a) through e), respectively. Paths are generated by a bracketing Figure 5.12: Paths from 1st through 5th starting points a) through e) on a simple energy landscape generated with QNA-XEL and Brent line search.



f) through j) show more detail of the paths on a) through e), respectively. Paths are generated by a heuristic exhaustive line **Figure 5.13:** Paths from 1st through 5th starting points a) through e) on a simple energy landscape generated with QNA-XEL. search.





Complex energy landscape

With the bracketing and Brent line search on the complex energy landscape, QNA-XEL and QNA-XEL (*Hessian only*) (4.6) paths with different values of n can differ significantly. This can be observed from Figures 5.15 and 5.16 with representative examples of paths from QNA-XEL and QNA-XEL (*Hessian only*) respectively. This is again because both algorithms use Broyden-XEL and the step is significantly varied by different values of n. Although QNA-XEL and QNA-XEL (*Hessian only*) give different magnitudes of the step which is varied by different values of n, the total steps and paths are similar because of the bracketing and Brent line search.

With a heuristic exhaustive line search, paths resulting from different values of n also differ from each other. This can be observed from Figures 5.17 and 5.18 with representative examples of paths from QNA-XEL and QNA-XEL (*Hessian only*) respectively. Similar to Newton-XEL paths on a complex energy landscape, QNA-XEL paths have smaller total steps for n > 1 and tend to converge to the closest local minima. On the contrary, paths have larger total steps for n < 1 and travel farther and more frequently reach the global minimum. However, when the total steps become too large paths are almost perpendicular to the gradient direction because of inaccurate \hat{K}^* . This case can be identified by the portions of paths that are perpendicular to the gradient and have small total steps which occurs more often as n gets smaller. For n = 0.125, paths sometimes diverge.

QNA-XEL and QNA-XEL (*Hessian only*) paths with a heuristic exhaustive line search show more deviation than those with the bracketing and Brent line search which implies that the heuristic exhaustive line search allows the magnitude of the step to have more effect on the paths. Paths with the heuristic exhaustive line search also have more total steps which implies that the heuristic exhaustive line search is less efficient than the bracketing and Brent line search.



Figure 5.15: Paths from 1st through 5th starting points a) through e) on a complex energy landscape generated with QNA-XEL. f) through j) show more detail of the paths on a) through e), respectively. Paths are generated by a bracketing and Brent line search.



Figure 5.16: Paths from 1st through 5th starting points a) through e) on a complex energy landscape generated with QNA-XEL $(Hessian \ only)$. f) through j) show more detail of the paths on a) through e), respectively. Paths are generated by a bracketing and Brent line search.



XEL. f) through j) show more detail of the paths on a) through e), respectively. Paths are generated by a heuristic exhaustive Figure 5.17: Paths from 1st through 5th starting points a) through e) on a complex energy landscape generated with QNAline search.



 $(Hessian \ only)$. f) through j) show more detail of the paths on a) through e), respectively. Paths are generated by a heuristic Figure 5.18: Paths from 1st through 5th starting points a) through e) on a complex energy landscape generated with QNA-XEL exhaustive line search.

5.1.1.3 BFGS-XEL and BFGS-XEL (Hessian only)

Simple energy landscape

With a bracketing and Brent line search on the simple energy landscape, BFGS-XEL and BFGS-XEL (*Hessian only*) (4.10) paths with different values of n can differ from one another. This can be observed from Figures 5.19 and 5.20 with representative examples of paths from BFGS-XEL and BFGS-XEL (*Hessian only*) respectively. The total step sizes of paths from both algorithms are not varied by the different values of n and their paths are identical due to the bracketing and Brent line search.

With a heuristic exhaustive line search, paths with different values of n on the simple energy landscape also differ from one another. This can be observed from Figures 5.21 and 5.22 with representative examples of paths from BFGS-XEL and BFGS-XEL (*Hessian only*) respectively. Unlike the Newton-XEL and the QNA-XEL paths on the simple energy landscape, after a few initial total steps in the BFGS-XEL paths the total step sizes are not varied with different values of n. After a few initial iterations, most $n \ge 1$ paths point straight toward the global minimum, but n < 1 paths do not. This indicates that $n \ge 1$ cases have more accurate estimations of the Hessian inverse, \hat{H}^* than those of the n < 1 case which means that larger n can be used to improve the accuracy of \hat{H}^* .

With a heuristic exhaustive line search, paths from BFGS-XEL and BFGS-XEL (*Hessian only*) are significantly different. The step sizes of BFGS-XEL (*Hessian only*) paths are more influenced by n as the total steps are larger when n < 1 and less when n > 1. This difference is more pronounced than that between Newton-XEL and Newton-XEL (*Hessian only*) and between QNA-XEL and QNA-XEL (*Hessian only*). BFGS-XEL and BFGS-XEL (*Hessian only*) paths with a heuristic exhaustive line search also show more deviation than those with the bracketing and Brent line search which implies that the heuristic exhaustive line search allows the magnitude of the step to have more effect on the path. Paths with the heuristic exhaustive line search







Figure 5.20: Paths from 1st through 5th starting points a) through e) on a simple energy landscape generated with BFGS-XEL (Hessian only). f) through j) show more detail of the paths on a) through e), respectively. Paths are generated by a bracketing and Brent line search.



f) through j) show more detail of the paths on a) through e), respectively. Paths are generated by a heuristic exhaustive line Figure 5.21: Paths from 1st through 5th starting points a) through e) on a simple energy landscape generated with BFGS-XEL. search.





also have more steps which implies that the heuristic exhaustive line search is less efficient that the bracketing and Brent line search.

Complex energy landscape

With the bracketing and Brent line search on the complex energy landscape, BFGS-XEL and BFGS-XEL (*Hessian only*) paths with different values of n differ from one another. This can be observed from Figures 5.23 and 5.24 with representative examples of paths from BFGS-XEL and BFGS-XEL (*Hessian only*) respectively. While the step size appears to be in independent of n, the deviation between paths from BFGS-XEL (*Hessian only*) is little noticeable.

With a heuristic exhaustive line search on the complex energy landscape, paths with different values of n differ from one another. This can be observed from Figures 5.25 and 5.26 with representative examples of paths from BFGS-XEL and BFGS-XEL (*Hessian only*) respectively. Unlike Newton-XEL and the QNA-XEL paths on a complex energy landscape, the relationship between the step size and n on the BFGS-XEL paths is less distinguishable.

With a heuristic exhaustive line search, BFGS-XEL and BFGS-XEL (*Hessian* only) paths also show slightly more deviation than those with the bracketing and Brent line search which implies that the heuristic exhaustive line search allows the magnitude of the step to have more effect on paths. Paths from both line search algorithms appear to have about the same number of steps which implies that both line search algorithms have about the same effect on the the BFGS-XEL and BFGS-XEL (*Hessian only*) in terms of speed.







XEL (Hessian only). f) through j) show more detail of the paths on a) through e), respectively. Paths are generated by a Figure 5.24: Paths from 1st through 5th starting points a) through e) on a complex energy landscape generated with BFGSbracketing and Brent line search.






XEL (Hessian only). f) through j) show more detail of the paths on a) through e), respectively. Paths are generated by a Figure 5.26: Paths from 1st through 5th starting points a) through e) on a complex energy landscape generated with BFGSheuristic exhaustive line search.

5.1.1.4 Conclusion of path comparison

The characteristics of paths on the simple and the complex energy landscape from the Newton-XEL, Newton-XEL (Hessian only), QNA-XEL, QNA-XEL (Hessian only), BFGS-XEL, and BFGS-XEL (Hessian only) with a bracketing and Brent line search and a heuristic exhaustive line search are compared and summarized on Table 5.1. Results show that paths are dependent on n with an exception of Newton-XEL paths on a simple energy landscape. The step size can show some dependency on n but it can be reduced by the bracketing and Brent line search and it is less apparent on the complex landscape. The XEL (Hessian only) algorithms yield different paths from the XEL algorithms because they give different magnitudes of the step but the effect can be reduced by the bracketing and Brent line search. The number of steps from paths with the bracketing and Brent line search are less than those from paths with the heuristic exhaustive line search which means the bracketing and Brent line search is more efficient. Lastly, the QNA-XEL and the BFGS-XEL paths suggest that n > 1 results in more accurate Hessian matrix estimation which is in agreement with Guideline *iv*. in Table 4.1.

To compare the quality and speed of different algorithms, the next section compares energy and number of iterations as the percentage of paths with lower, equal, or higher values compared to n = 1.

5.1.2 Comparison of Energy and Number of Iterations to the Unmodified Case

For each simulation, the converged energy and the number of iterations (or steps) are evaluated and compared to those of the unmodified case (n = 1) to assess quality and speed. The percentage of paths from each XEL algorithm with different n cases that has lower, equal, or higher energy or number of iterations than those of n = 1 on each energy landscape are determined. The values are considered "equal" if they are within $\pm 5\%$ of those of n = 1, "lower" if they are lower than the range, and "higher"

Characte		Paths are ' different ve	Step size are different va	er of steps om the oth	Paths differ (<i>Hessian</i>	Paths are ' different va	Step size are different ve	oer of steps om the oth	Paths diffe (<i>Hessian</i>
ristic		varied u lues of n	e varied alues of n	compared to er line search	r from only)	varied u alues of n	e varied n	compared to er line search	r from only)
New	Brent	No	No	Less	$N_{\rm O}$	Few	No	Less	Some
ton-XEL	Exhaustive	No	Yes	More	Few	Most	No	More	Most
QN Q	Brent	Yes	No	Less	No	Yes	No	Less	Some
VA-XEL	Exhaustive	Yes	Yes	More	Few	Yes	Yes	More	Most
BF	Brent	Yes	No	Less	No	Yes	No	Same	Some
'GS-XEL	Exhaustive	Yes	Some	More	Yes	Yes	No	Same	Most

Table 5.1: A summary of the path characteristics from implemented algorithms.

if they are above the range. Results are shown in Figures 5.27 through 5.38.

5.1.2.1 Newton-XEL and Newton-XEL (Hessian only)

Figures 5.27 through 5.30 display the percentage of paths from Newton-XEL and Newton-XEL (*Hessian only*) with different n cases that have smaller, equal, or higher number of iterations or energy than those of n = 1 on the simple and the complex energy landscape. While paths resulting in Figures 5.27 and 5.28 are generated by a bracketing and Brent line search, paths resulting in Figures 5.29 and 5.30 are generated by a heuristic exhaustive line search. The higher percentage of paths with smaller energy or number of iterations implies better quality and speed.

With a bracketing and Brent line search, the Newton-XEL (Figure 5.27 a and c) on the simple energy landscape have the same quality and speed for all n as all paths have equal energy and number of iterations compared to n = 1. On the complex energy landscape (Figure 5.27 b and d) quality and speed become slightly worse. Although up to 20% of paths for all n cases from the Newton-XEL have lower energy than n = 1, 12% to 40% have higher energy and up to 30% have higher number of iterations.

With a bracketing and Brent line search, the Newton-XEL (*Hessian only*) (Figure 5.28 a and c) on the simple energy landscape also have the same quality and speed for all n as all paths have equal energy and number of iterations compared to n = 1. On the complex energy landscape (Figure 5.28 b and d) quality becomes slightly better in some n cases and speed becomes slightly worse. When n < 1, 8% to 16% of paths have higher energy than n = 1 but 24% to 36% have lower energy, and when n > 1, 24% to 30% have higher energy and 12% to 16% have lower energy. For all n cases 12% to 32% of paths have a higher number of iterations while only 4% to 12% have lower number of iterations. Results from Newton-XEL (*Hessian only*) show slightly more deviation for different values of n because n has more effect on the magnitude



Figure 5.27: Newton-XEL: The percentage of paths that yields the lower, the equal, or the higher number of iterations in c) and d) or energy in a) and b) compared to n = 1. Shown in a) and c) are the percentage of paths on the simple energy landscape and in b) and d), on the complex energy landscape. Paths are generated by a bracketing and Brent line search.



Figure 5.28: Newton-XEL (*Hessian only*): The percentage of paths that yields the lower, the equal, or the higher number of iterations in c) and d) or energy in a) and b) compared to n = 1. Shown in a) and c) are the percentage of paths on the simple energy landscape and in b) and d), on the complex energy landscape. Paths are generated by a bracketing and Brent line search.



Figure 5.29: Newton-XEL: The percentage of paths that yields the lower, the equal, or the higher number of iterations in c) and d) or energy in a) and b) compared to n = 1. Shown in a) and c) are the percentage of paths on the simple energy landscape and in b) and d), on the complex energy landscape. Paths are generated by a heuristic exhaustive line search.



Figure 5.30: Newton-XEL (*Hessian only*): The percentage of paths that yields the lower, the equal, or the higher number of iterations in c) and d) or energy in a) and b) compared to n = 1. Shown in a) and c) are the percentage of paths on the simple energy landscape and in b) and d), on the complex energy landscape. Paths are generated by a heuristic exhaustive line search.

of the step of the Newton-XEL (*Hessian only*) than that of the Newton-XEL.

With a heuristic exhaustive line search, the Newton-XEL (Figure 5.29 a and c) on the simple energy landscape has the same quality for all n but higher speed for n < 1 and lower speed for n > 1. All paths have equal energy compared to n = 1 but 16% to 48% of n < 1 paths have a lower number of iterations while 48% of n > 1 paths have a higher number of iterations. The lower speed is due to the small step size of n > 1 cases. On the complex energy landscape (Figure 5.29 b and d) quality and speed become slightly worse. Although up to 16% of paths for all n cases from the Newton-XEL have lower energy than n = 1 and up to 24% have lower number of iterations. The lower speed is due to the small step of iterations, 12% to 32% have higher energy and 16% to 44% have higher number of iterations.

With a heuristic exhaustive line search, the Newton-XEL (*Hessian only*) (Figure 5.30 a and c) on the simple energy landscape have the same quality for all n but higher speed for n < 1 and lower speed for n > 1. All paths have equal energy compared to n = 1 but 64% of n < 1 paths have a lower number of iterations while 64% of n > 1 paths have a higher number of iterations. On the complex energy landscape (Figure 5.30 b and d) quality becomes better for n < 1 cases and worse for n > 1 and speed becomes significantly worse. As n < 1, 28% to 60% of paths have lower energy than n = 1 but as n > 1, 44% to 48% have higher energy. Their poorer performance is due to their smaller step size, which does not allow paths to travel beyond local minima. While some n < 1 paths have a lower number of iterations, 100% of the n > 1 paths have a higher number of iterations. While the low speed in the n > 1paths is due to their small step size, the reduction in speed in n < 1 cases is due to their large step size and an increase in the complexity of the energy landscape. Since a larger step tends to overshoot a minimum, n < 1 paths take several iterations to reach a minimum as they keep passing it which is inefficient. This is less severe in the simple energy landscape because its global minimum is very flat. When a path gets in the proximity of the global minimum, the convergence criterion is met and the path stops. Since the complex energy landscape is rougher than the simple energy landscape, the path needs to get much closer to the minumum before the convergence criterion is met.

The results show that Newton-XEL (*Hessian only*) are more dependent on n than those of Newton-XEL. Results also show that the bracketing and Brent line search may be a better choice of a line search because it can slightly improve quality without a substantial reduction of speed. While the heuristic exhaustive line search can achieve better quality and speed in some cases, it can also significantly worsen the quality and speed of the results.

5.1.2.2 QNA-XEL and QNA-XEL (Hessian only)

Figures 5.31 through 5.34 display the percentage of paths from QNA-XEL and QNA-XEL (*Hessian only*) with different n cases that has smaller, equal, or higher number of iterations or energy than those of n = 1 on the simple and the complex energy landscape. While paths resulting in Figures 5.31 and 5.32 are generated by a bracketing and Brent line search, paths resulting in Figures 5.33 and 5.34 are generated by a heuristic exhaustive line search. The higher percentage of paths with smaller energy or number of iterations implies better quality and speed.

With a bracketing and Brent line search, the QNA-XEL (Figure 5.31 a and c) on the simple energy landscape has slightly better quality for all n, lower speed for n < 1, and about the same speed for n > 1 compared to n = 1. As 8% to 16% of paths of most n cases have lower energy compared to n = 1, only a few n cases have 8% to 16% of paths with higher energy. In terms of speed 16% to 48% of paths for all n cases have higher number of iterations but up to 40% of paths with n > 1 have lower number of iterations. On the complex energy landscape (Figure 5.31 b and d) quality on average is about the same as n = 1 but speed becomes worse. On average



Figure 5.31: QNA-XEL: The percentage of paths that yields the lower, the equal, or the higher number of iterations in c) and d) or energy in a) and b) compared to n = 1. Shown in a) and c) are the percentage of paths on the simple energy landscape and in b) and d), on the complex energy landscape. Paths are generated by a bracketing and Brent line search.



Figure 5.32: QNA-XEL (*Hessian only*): The percentage of paths that yields the lower, the equal, or the higher number of iterations in c) and d) or energy in a) and b) compared to n = 1. Shown in a) and c) are the percentage of paths on the simple energy landscape and in b) and d), on the complex energy landscape. Paths are generated by a bracketing and Brent line search.



Figure 5.33: QNA-XEL: The percentage of paths that yields the lower, the equal, or the higher number of iterations in c) and d) or energy in a) and b) compared to n = 1. Shown in a) and c) are the percentage of paths on the simple energy landscape and in b) and d), on the complex energy landscape. Paths are generated by a heuristic exhaustive line search.



Figure 5.34: QNA-XEL (*Hessian only*): The percentage of paths that yields the lower, the equal, or the higher number of iterations in c) and d) or energy in a) and b) compared to n = 1. Shown in a) and c) are the percentage of paths on the simple energy landscape and in b) and d), on the complex energy landscape. Paths are generated by a heuristic exhaustive line search.

quality is about the same as n = 1 because up to 44% of paths yield lower energy but up to 36% of paths have higher energy. In terms of speed while up to 72% of all paths have a higher number of iterations, only 36% of n > 1 paths have a lower number of iterations. Results show that n > 1 yields slightly better quality and higher speed than n < 1 for QNA-XEL with a bracketing and Brent line search.

With a bracketing and Brent line search, results from the QNA-XEL (*Hessian* only) (Figure 5.32 a and c) are almost identical to those from the QNA-XEL which is consistent with observations made in the previous section. The QNA-XEL (*Hessian* only) on the simple energy landscape has slightly better quality for all n and lower speed for n < 1 and about the same speed for n > 1 compared to n = 1. On the complex energy landscape (Figure 5.32 b and d) quality on average is about the same as n = 1 but speed becomes worse. Results also show that n > 1 yields slightly better quality and higher speed than n < 1 for QNA-XEL (*Hessian* only) with a bracketing and Brent line search.

With a heuristic exhaustive line search, the QNA-XEL (Figure 5.33 a and c) on the simple energy landscape has the same quality for n > 1, slightly worse quality for n < 1, and lower speed for all n compared to n = 1. As 32% to 40% of n < 1 paths have higher energy than that of n = 1, all n > 1 paths have the same energy. This is because some n < 1 paths can not reach the global solution before the convergence criterion is met. Due to an inaccurate estimated modified Hessian matrix \hat{K}^* and a very small step size, \hat{K}^* does not get improved so the paths are almost perpendicular to the gradient and advance with very small steps. It is possible that if the convergence criterion becomes stricter, these paths may eventually reach the global minimum. In terms of speed, while only 32% of n < 1 paths have a lower number of iterations, 48% of n < 1 paths and 96% of n > 1 paths have a higher number of iterations.

With a heuristic exhaustive line search, the QNA-XEL on the complex energy landscape (Figure 5.33 b and d) has the same quality for n < 1 but worse quality for n > 1 and lower speed. While up to 40% of n < 1 paths have higher energy, up to 44% have lower energy so on average the quality is about the same as n = 1. For n > 1 the quality is worse as up to 52% of the paths have higher energy than n = 1. Speed is significantly worse than n = 1 as 60% to 100% of all paths have a higher number of iterations. The lower speed is due to inaccuracy of \hat{K}^* in n < 1 cases and too small step size in n > 1 cases.

With a heuristic exhaustive line search, results from QNA-XEL (*Hessian only*) (Figure 5.34 a and c) are considerably similar to those from QNA-XEL. On the simple energy landscape QNA-XEL (*Hessian only*) has the same quality compared to n = 1 for n > 1 but slightly worse quality for n < 1 and lower speed for all n except for n = 0.125. On the complex energy landscape (Figure 5.34 b and d) it has slightly better quality for n < 1 but worse quality for n > 1 and lower speed.

Results show that the heuristic exhaustive line search can significantly reduce the quality and the speed of QNA-XEL and QNA-XEL (*Hessian only*).

5.1.2.3 BFGS-XEL and BFGS-XEL (Hessian only)

Figures 5.35 through 5.38 display the percentage of paths from BFGS-XEL and BFGS-XEL (*Hessian only*) with different n cases that have smaller, equal, or higher number of iterations or energy than those of n = 1 on the simple and the complex energy landscape. While paths resulting in Figures 5.35 and 5.36 are generated by a bracketing and Brent line search, paths resulting in Figures 5.37 and 5.38 are generated by a heuristic exhaustive line search. The higher percentage of paths with a lower energy or a smaller number of iterations implies better quality or speed.

With a bracketing and Brent line search, the BFGS-XEL (Figure 5.35 a and c) on the simple energy landscape has the same quality and speed for all n compared to n = 1. In terms of quality all paths have equal energy compared to n = 1. In terms of speed 8% to 24% of most paths have a higher number of iterations but the



Figure 5.35: BFGS-XEL: The percentage of paths that yields the lower, the equal, or the higher number of iterations in c) and d) or energy in a) and b) compared to n = 1. Shown in a) and c) are the percentage of paths on the simple energy landscape and in b) and d), on the complex energy landscape. Paths are generated by a bracketing and Brent line search.



Figure 5.36: BFGS-XEL (*Hessian only*): The percentage of paths that yields the lower, the equal, or the higher number of iterations in c) and d) or energy in a) and b) compared to n = 1. Shown in a) and c) are the percentage of paths on the simple energy landscape and in b) and d), on the complex energy landscape. Paths are generated by a bracketing and Brent line search.



Figure 5.37: BFGS-XEL: The percentage of paths that yields the lower, the equal, or the higher number of iterations in c) and d) or energy in a) and b) compared to n = 1. Shown in a) and c) are the percentage of paths on the simple energy landscape and in b) and d), on the complex energy landscape. Paths are generated by a heuristic exhaustive line search.



Figure 5.38: BFGS-XEL (*Hessian only*): The percentage of paths that yields the lower, the equal, or the higher number of iterations in c) and d) or energy in a) and b) compared to n = 1. Shown in a) and c) are the percentage of paths on the simple energy landscape and in b) and d), on the complex energy landscape. Paths are generated by a heuristic exhaustive line search.

same paercentage have lower number of iterations. On the complex energy landscape (Figure 5.35 b and d) quality on average is about the same as n = 1 but speed is better. In terms of quality up to 32% of all paths have lower energy but the same approximate percentage of paths have higher energy. In terms of speed while up to only 28% of all paths have a higher number of iterations, up to 40% of n < 1 paths and up to 80% of n > 1 paths have a lower number of iterations. Results show that n > 1 can yield significantly higher speed than n < 1 for BFGS-XEL with a bracketing and Brent line search.

With a bracketing and Brent line search, results from the BFGS-XEL (*Hessian* only) (Figure 5.36 a and c) are substantially similar to those from the BFGS-XEL which is consistent with observations made in the previous section. The BFGS-XEL (*Hessian only*) on the simple energy landscape has the same quality and speed for all n compared to n = 1. On the complex energy landscape (Figure 5.36 b and d) on average quality is about the same as n = 1 but speed is better. Results show that n < 1 yields slightly better quality than n > 1 while n > 1 yields higher speed than n < 1 for BFGS-XEL (*Hessian only*) with a bracketing and Brent line search.

With a heuristic exhaustive line search, the BFGS-XEL (Figure 5.37 a and c) on the simple energy landscape has the same quality for all n compared to n = 1, lower speed for n < 1, and higher speed for n > 1. In terms of quality all paths have equal energy compared to n = 1. In terms of speed up to 56% of n > 1 paths have a lower number of iterations but only 24% have higher number of iterations. For n < 1 up to 80% have higher number of iterations. The low speed in n < 1 is caused by the inaccuracy of the estimated modified Hessian inverse \hat{H}^* so n < 1 paths take longer to converge. On the complex energy landscape (Figure 5.37 b and d) the BFGS-XEL have about the same quality on average but lower speed for n > 1 and higher speed for n < 1. While up to 40% of all paths have either higher or lower energy so on average the quality is about the same as n = 1. In terms of speed up to 60% of the paths have higher energy than n = 1 but up to 56% of n < 1 paths have a lower number of iterations. The effect of n on speed on the complex energy landscape is opposite to that on the simple energy landscape.

With a heuristic exhaustive line search, the BFGS-XEL (*Hessian only*) (Figure 5.38 a and c) on the simple energy landscape has the same quality for all n compared to n = 1, lower speed for n > 1, and higher speed for n < 1. In terms of quality all paths have equal energy compared to n = 1. In terms of speed up to 80% of n > 1 paths have a higher number of iterations but only 8% have higher number of iterations. For n < 1 up to 80% have lower number of iterations. The low speed in n > 1 is possibly caused by the step size being too small. On the complex energy landscape (Figure 5.38 b and d) the BFGS-XEL (*Hessian only*) have better quality for n < 1 but slightly worse quality for n > 1 and lower speed. In terms of quality up to 52% of n < 1 paths have lower energy but up to only 16% have lower energy. In terms of speed n = 0.5 yields higher speed but up to 100% of paths in most n cases have a higher number of iterations. The results show that n < 1 yields better results than n > 1 for the BFGS-XEL (*Hessian only*) with a heuristic exhaustive line search.

The bracketing and Brent line search yields more consistent results for different values of n than the heuristic exhaustive line search and it yields better quality and speed on average.

5.1.2.4 Conclusion of iterations and energy comparison to the unmodified case

The energy and the number of iterations for each path generated by Newton-XEL, Newton-XEL (Hessian only), QNA-XEL, QNA-XEL (Hessian only), BFGS-XEL, and BFGS-XEL (Hessian only) with different values of n are compared to those of n = 1. The results are the percentage of paths with smaller, equal, or higher energy and number of iterations are presented from which the quality and the speed of each algorithm are evaluated. Tables 5.2 and 5.3 summarize the quality and the speed of XEL and XEL (Hessian only) algorithms with n < 1 and n > 1. Quality or speed is "Significantly better" when the averaged percentage of paths with lower energy or number of iterations is $\geq 50\%$ higher than the averaged percentage of paths with higher energy or number of iterations, "Better" when it is $\geq 30\%$ higher, and "Slightly better" when it is $\geq 10\%$ higher. Quality or speed is "Significantly worse" when the averaged percentage of paths with lower energy or number of iterations is $\geq 50\%$ lower than the averaged percentage of paths with higher energy or number of iterations, "Slightly worse" when it is $\geq 30\%$ lower. Otherwise quality or speed is "Same."

Results in Tables 5.2 and 5.3 show that the XEL and the XEL (*Hessian only*) algorithms yield the same quality on the simple energy landscape which demonstrates that the difference in the magnitudes of the step of these algorithms does not effect the quality on a smooth landscape.

The results are inconclusive for which value of n yields better performance for all algorithms. While n > 1 yields better quality on the simple energy landscape, n < 1yields better quality on the complex energy landscape. Although in each algorithm different values of n yield higher speed, n > 1 yields higher speed when used with the bracketing and Brent line search and n < 1 yields higher speed when used with the heuristic exhaustive line search. However, on average the bracketing and Brent line search performs better than the heuristic exhaustive line search. On the complex energy landscape the BFGS-XEL and the BFGS-XEL (*Hessian* only) with the bracketing and Brent line search yield the best overall quality and speed without sacrificing one or another. Thus, they would be the methods of choice at least for these results. The fact that they perform well on the complex energy landscape can imply that they are more robust than the other algorithms and therefore they are most likely to perform better in the real energy landscape.

	ive			tly	<u>۲</u>		۲.	N		
S-XEL	Exhausti	Same	Same	Significan worse	Slightly better	Same	Slightly better	Slightly better	Worse	
BFG	Brent	Same	Same	Same	Slightly worse	Same	Same	Same	Significantly better	
-XEL	Exhaustive	Worse	Same	Slightly worse	Significantly worse	Same	Worse	Significantly worse	Significantly worse	
QNA	Brent	Same	Slightly better	Worse	Same	Same	Same	Significantly worse	Worse	
ton-XEL	Exhaustive	Same	Same	Slightly better	Worse	Same	Slightly worse	Same	Slightly worse	
New	Brent	Same	Same	Same	Same	Same	Slightly worse	Same	Slightly worse	
aristic		n < 1	n > 1	n < 1	n > 1	n < 1	n > 1	n < 1	n > 1	
Charact		Quality Speed			- - -	Quality Speed				
		Landscape					ədvəs	basJ		
		Simple Energy				Complex Energy				

Table 5.2: A summary of the quality and the speed results: XEL algorithms. (1 of 2)

Chi		Landscape		Landscape					
aracteristic		n < 1	ality $n>1$	n < 1eed	n > 1	n < 1 ality	n > 1	n < 1eed	n > 1
(Hess	Brent	Same	Same	Same	Same	Slightly better	Slightly worse	Slightly worse	Slightly worse
ton-XEL sian only)	Exhaustive	Same	Same	Significantly better	Significantly worse	Better	Worse	Worse	Significantly worse
QNA. (Hessia	\mathbf{Brent}	Same	Slightly better	Worse	Same	Same	Same	Significantly worse	Worse
-XEL ta only)	Exhaustive	Worse	Same	Slightly better	Significantly worse	Better	Worse	Significantly worse	Significantly worse
BFG9 (Hessic	\mathbf{Brent}	Same	Same	Same	Slightly worse	Slightly better	Slightly worse	Same	Significantly better
S-XEL in only)	Exhaustive	Same	Same	Significantly better	Significantly worse	Better	Same	Slightly worse	Significantly worse

Table 5.3: A summary of the quality and the speed results: XEL (*Hessian only*) algorithms. (2 of 2)

5.1.3 Global Search Performance

Since Newton's method, the QNA, and the BFGS algorithm are local optimization algorithms, XEL algorithms are not expected to solve the global minimization. However, the XEL may improve the performance in a global search by transforming the energy landscape such that it assists the optimization. Here the global search performance is defined as an ability to locate the global solution. This ability contributes to the quality because a more successful global search yields a lower energy which implies higher quality. To compare the global search performance the number of paths that are able to locate the global minimum from Newton-XEL, Newton-XEL (*Hessian* only), QNA-XEL, QNA-XEL (*Hessian only*), BFGS-XEL, and BFGS-XEL (*Hessian* only) are compared. Figures 5.39 and 5.41 present results from XEL algorithms and Figures 5.40 and 5.42 present results from XEL (*Hessian only*) algorithms. Results in Figures 5.39 and 5.40 are generated by a bracketing and Brent line search, results in Figures 5.41 and 5.42 are generated by a heuristic exhaustive line search. Higher number of paths may indicate higher ability to locate the global minimum which is desired.

With a bracketing and Brent line search, every path generated by the Newton-XEL and the BFGS-XEL on the simple energy landscape (Figure 5.39 a) successfully locates the global minimum, but some n cases of QNA-XEL do not. Note that Newton-XEL has only 21 paths because some starting points have zero Hessian matrix and yield no path. High number of paths can locate the global minimum because of the simplicity of the energy landscape. On the complex energy landscape (Figure 5.39 b) far fewer paths from all algorithms can locate the global minimum. All algorithms where $n \neq 1$ have fewer successful paths than n = 1 and in QNA-XEL and BFGS-XEL n > 1 cases have more successful paths than n < 1.

With a bracketing and Brent line search, the results of the Newton-XEL (*Hessian only*), the QNA-XEL (*Hessian only*), and the BFGS-XEL (*Hessian only*) on the simple



Figure 5.39: The number of paths on a simple or a complex energy landscape generated by Newton-XEL, QNA-XEL, or BFGS-XEL that locate the global minimum. Paths are generated by a bracketing and Brent line search.



Figure 5.40: The number of paths on a simple or a complex energy landscape generated by Newton-XEL (*Hessian only*), QNA-XEL (*Hessian only*), or BFGS-XEL (*Hessian only*) that locate the global minimum. Paths are generated by a bracketing and Brent line search.



Figure 5.41: The number of paths on a simple or a complex energy landscape generated by Newton-XEL, QNA-XEL, or BFGS-XEL that locate the global minimum. Paths are generated by a heuristic exhaustive line search.



Figure 5.42: The number of paths on a simple or a complex energy landscape generated by Newton-XEL (*Hessian only*), QNA-XEL (*Hessian only*), or BFGS-XEL (*Hessian only*) that locate the global minimum. Paths are generated by a heuristic exhaustive line search.

energy landscape (Figure 5.40 a) are the same as those of the Newton-XEL, the QNA-XEL, and the BFGS-XEL. On the complex energy landscape (Figure 5.40 b) far fewer paths from all algorithms can locate the global minimum. All algorithms where $n \neq 1$ have fewer successful paths than n = 1 and in Newton-XEL (*Hessian only*) and BFGS-XEL (*Hessian only*) n < 1 cases have more successful paths than n > 1 but in QNA-XEL (*Hessian only*) n > 1 cases have more successful paths than n < 1.

With a heuristic exhaustive line search, every path generated by the Newton-XEL and the BFGS-XEL on the simple energy landscape (Figure 5.41 a) successfully locates the global minimum, but QNA-XEL with n < 1 have 15 to 17 paths that do so. The high number of paths that reach the global minimum is due to the simplicity of the energy landscape. Some QNA-XEL paths never reach the minimum because an inaccurate estimated modified Hessian matrix gives a step almost perpendicular to the gradient that yields a very small step size. As the result the path never reaches the minimum before the simulation reaches the maximum iteration. On the complex energy landscape (Figure 5.41 b) far fewer paths from all algorithms can locate the global minimum. In each algorithm only a few values of $n \neq 1$ have more successful paths than n = 1. In QNA-XEL n < 1 cases have more successful paths than n > 1but in BFGS-XEL n>1 cases have more successful paths than $n\leq 1$ and in Newton-XEL all cases have about the same number of paths. As n < 1 QNA-XEL has a larger step size and while n < 1 cases have more successful paths than n > 1 QNA-XEL with n = 0.125 have fewer successful paths than n = 0.5 or 0.25. This suggests that a larger step size gives a more favorable condition for a global minimum search but a step size too large can cause \hat{K}^* to diverge.

With a heuristic exhaustive line search, the results of the Newton-XEL (*Hessian only*), the QNA-XEL (*Hessian only*), and the BFGS-XEL (*Hessian only*) on the simple

energy landscape (Figure 5.42 a) are the same as those of the Newton-XEL, the QNA-XEL, and the BFGS-XEL. On the complex energy landscape (Figure 5.42 b) far fewer paths from all algorithms can locate the global minimum. For all algorithms n < 1cases have more successful paths than $n \geq 1$ cases.

Results are inconclusive as to whether the XEL has positive or negative impact on the global search. The XEL does not change the performance in a global search for the Newton-XEL, the Newton-XEL (*Hessian only*), the BFGS-XEL, and the BFGS-XEL (*Hessian only*) on the simple energy landscape. The XEL may reduce or improve the performance on the complex energy landscape, but more often it reduces. The XEL with n < 1 reduces the performance in a global search in the QNA-XEL and the QNA-XEL (*Hessian only*) except when they are used with the heuristic exhaustive line search on the complex energy landscape. With the bracketing and Brent line search the performance is slightly varied by the values of n but with the heuristic exhaustive line search the variation can be significant. The XEL impact on the global search performance is further investigated in the optimization with a probabilistic search presented in Section 5.2.

5.1.4 Conclusion of XEL Algorithms without Probabilistic Search

To study the effect of the XEL on Newton's method, the QNA, and the BFGS, the Newton-XEL, the Newton-XEL (*Hessian only*), the QNA-XEL, the QNA-XEL (*Hessian only*), the BFGS-XEL, and the BFGS-XEL (*Hessian only*) are implemented without a probabilistic search on a simulated two-dimensional energy landscape with a bracketing and Brent or a heuristic exhaustive line search. The results are evaluated in terms of quality and speed by comparing the converged energy and the number of iterations. The paths generated by the investigated algorithms are compared and the the impact of the XEL on the global search is investigated.

A path comparison shows that paths and step sizes are dependent on n but this

dependency can be reduced by the shape of the energy landscape or a line search algorithm. Paths tend to have longer steps when n < 1 but shorter steps when n > 1which is in agreement with Guideline *ii*. in Table 4.1. Some QNA-XEL or BFGS-XEL paths show that poor Hessian matrix estimation can result in poor speed and even divergence while n > 1 results in a more accurate estimation. This is in agreement with Guideline *iv*. and Guideline *iii*. since increasing n results in better convergence and small steps which implies a bounded magnitude of λ .

The percentages of paths that have lower, equal, or higher energy or number of iterations compared to n = 1 show that n > 1 yields better quality on the simple energy landscape but n < 1 yields better quality on the complex energy landscape. Since n > 1 gives a more accurate Hessian matrix estimation, paths can converge quickly to a minimum. This works well on the simple energy landscape with only one minimum but on the complex energy landscape a local minimum nearest to the starting point may not have the low energy and therefore n > 1 may not result in a lower energy and n < 1 with longer steps tends to converge a lower minimum energy by covering more distance. This suggests that n < 1 should be used at the beginning of optimization for farther exploration and n > 1 should be used at the end for convergence. This is in agreement with Guideline v. in Table 4.1.

On average the bracketing and Brent line search yields better quality and speed than the heuristic exhaustive line search and thus it is implemented in the optimization with a probabilistic search presented next in Section 5.2. The algorithms investigated in the next section are the QNA-XEL, the QNA-XEL (*Hessian only*), the BFGS-XEL, and the BFGS-XEL (*Hessian only*). Results in this section show that with the bracketing and Brent line search the QNA-XEL and the QNA-XEL (*Hessian only*) yield better quality and speed when n > 1, the BFGS-XEL yields the same quality with different values of n but slightly higher speed when n > 1, and the BFGS-XEL (*Hessian only*) yields better quality when n < 1 and slightly higher speed when n > 1. These will be compared with those in Section 5.2.

Lastly, results show that the XEL impact on the global search is inconclusive. This impact is further investigated in Section 5.2
5.2 XEL Algorithms with Probabilistic Search

This section investigates the effects of XEL on the optimization algorithm combined with a probabilistic search. The QNA-XEL, the QNA-XEL (*Hessian only*), the BFGS-XEL, and the BFGS-XEL (*Hessian only*) are implemented with the bracketing and Brent line search on the protein conformation prediction software Rosetta. The Newton-XEL and the Newton-XEL (*Hessian only*) is not studied because the second order derivatives are not be available in this platform.

5.2.1 Rosetta

Using both analytical and probabilistic methods, Rosetta is a computer program that predicts protein conformation as it constructs decoys or initial configurations from pre-generated fragments (decoy generation) and refines them by perturbing and then minimizing the predicted configurations (refinement protocol) [11] which is called the Monte Carlo-minimization approach.

Decoy generation consists of three components: fragment generation, scoring functions, and fragment assembly. A decoy is generated during the fragment assembly by piecing the generated fragments together which is guided by the scoring function. Fragments are the structural templates of small windows of a protein sequence and potential shapes that can be assumed by corresponding windows of the protein molecule. Windows are defined as 3- or 9-residue sections of a protein sequence. By comparing the protein sequence to the protein database, fragments are generated for all possible windows in the protein sequence, which are then ranked according to sequence and secondary structure similarity.

The second component essential for both decoy generation and refinement protocol are the scoring functions. Two types of scoring functions, centroid and full atom, are available for course-grained and atomic-level descriptions of the molecule structure. Despite their differences, both include energy functions, such as van der Waals interactions and electrostatics, and non-energy potentials, such as the Ramachandran torsion preferences [10].

Lastly, during fragment assembly a decoy is constructed as the parts of the molecule corresponding to randomly chosen sequence windows assume the shapes of fragments randomly chosen from high-ranking fragments. After each fragment is inserted, the scoring functions of the entire protein are evaluated, and using the Monte Carlo method this insertion is kept or discarded. At the beginning of the fragment assembly fewer scoring functions are strategically evaluated to encourage local folding of the molecule and toward the end of the process all scoring functions are evaluated to encourage compact packing.

After the decoys are generated, a refinement protocol perturbs each decoy configuration with several small moves and performs a minimization after each perturbation using an analytical optimization algorithm. Here, the QNA-XEL, the QNA-XEL (*Hessian only*), the BFGS-XEL, and the BFGS-XEL (*Hessian only*) along with the bracketing and Brent line search are implemented in the centroid refinement protocol and compared to the QNA and the BFGS algorithm. After each perturbation and each minimization, the Monte Carlo method is used to determine if the new configuration should be accepted or rejected. If accepted, the next set of the perturbation and minimization is done on the new configuration. Otherwise, the next set is done on the previous configuration. The centroid refinement protocol which uses score functions describing scores between residues is chosen to show proof-of-concept because it is less computational expensive than the full atom refinement protocol which is used in [11].

Shown as a flowchart in Figure 5.43, the Rosetta centroid refinement protocol repeats the following two steps six times: 1) minimize the energy on the entire chain, 2) perturb and minimize the energy on portions of the chain several times depending on the chain length. The minimization in both 1) and 2) uses the QNA-XEL, the



Figure 5.43: Rosetta centroid refinement protocol. The QNA-XEL, the QNA-XEL (*Hessian only*), the BFGS-XEL, and the BFGS-XEL (*Hessian only*) are implemented during the minimization process.

QNA-XEL (*Hessian only*), the BFGS-XEL, and the BFGS-XEL (*Hessian only*) to locate the configuration with the minimum score, which is the weighted sum of the scoring functions and implies an energy function E. Each minimization converges and stops when the change in the score is less than 0.5% or when iterations reach a maximum of 200.

Simulations are performed on several proteins, each analyzed with several values of n including n = 1 which corresponds to the unmodified energy landscape. Proteins used in the simulation are those provided in the Rosetta software package and the Robetta server [37], which contains fragment files and other necessary files to generate decoys, some of which are not in the protein database. Five small proteins analyzed here are 1ubq, 1d3z, dom6, hetr, and mdmi, which have 76, 82, 126, 149, and 304 residues, respectively. Various sizes are chosen to determine if the size of the protein effects the results and the range is limited to the size of an average protein because of the available computational resources.

For each protein 50 different decoys are generated and used as starting configurations in the refinement for each n value. Because of the probabilistic nature of the methods used in decoy generation and the perturbation in the refinement protocol, the number of decoys is chosen so that the sample size is sufficient to produce results that represent the performance of the optimization algorithm while small enough for the available computational resources. Simulations on 50 decoys for each n case take 3:40 to 32:18 hours depending on the protein on an AMD AthlonTM XP 3200+ (2.1 GHz) or AMD AthlonTM 64 Processor 3500+ (2.2 GHz) computers. Typical runtime for an accurate prediction of the conformation of a protein is 150 days.

The results are evaluated in terms of quality, speed, and efficiency. For quality an average score improvement, a lowest score, and similarity to the native conformation are determined from output scores and configurations. These quantities are defined and discussed in Sections 5.2.2, 5.2.3, and 5.2.6 respectively. The similarity of output

configurations to the native conformation are evaluated in terms of TM score [88], GDT score [79, 80], MaxSub score [62], and RMSD [32, 33]. These quantities are discussed in Section 5.2.6. For speed an average iteration is determined from the number of total iterations. This quantity is defined and discussed in Section 5.2.4. Efficiency is determined from the average score improvement and the average iteration. This quantity is defined and discussed in Section 5.2.7 discusses results from a further analysis studying the effect of XEL when its simulations are ran with the same number of iterations as that of the unmodified case.

5.2.2 Score Improvement

This section gives the evaluation of the QNA-XEL, the QNA-XEL (*Hessian only*), the BFGS-XEL, and the BFGS-XEL (*Hessian only*) on quality by determining the score improvement quantities described below.

After the refinement process, the scores E of the obtained configurations are evaluated. The average of score improvement AveSI is defined as the average of the score improvements SI which are the differences between scores of the resulting configurations (outputs) and those of the decoys (inputs):

$$AveSI = \frac{1}{N} \sum_{i=1}^{N} SI_i$$

$$SI_i = E_{m,i} - E_{0,i}$$
(5.2)

where N is the number of decoys (N = 50), SI_i is the score improvement of the i^{th} decoy, $E_{m,i}$ is the output score or the score at the m^{th} or last iteration for the i^{th} decoy, and $E_{0,i}$ is the input score or the score at the 0^{th} iteration for the i^{th} decoy. Since the refinement protocol is a minimization process, the score of a resulting configuration is considered "improved" when its value is less than that of the input decoy. Therefore, the lower SI and AveSI values imply better results.

The results from different cases of n are compared to those from unmodified cases

(n = 1) by subtracting the AveSI values of unmodified cases from those of the XEL cases:

$$\% \Delta AveSI_n = \frac{AveSI_n - AveSI_{n=1}}{|AveSI_{n=1}|} \times 100$$
(5.3)

where $\% \Delta AveSI_n$ is the percent difference of the average score improvement compared to the unmodified case, $AveSI_n$ is the AveSI value of the XEL case, and $AveSI_{n=1}$ is the AveSI value of unmodified case. The lower $\% \Delta AveSI_n$ indicates that the XEL case yields greater improvement and therefore a better result.

5.2.2.1 QNA-XEL and QNA-XEL (Hessian only)

Figures 5.44 and 5.45 display AveSI versus various values of n for each protein for the QNA-XEL and the QNA-XEL (*Hessian only*), respectively. The black dashed horizontal lines indicate the AveSI from the n = 1 cases and the red dotted lines above and below the black dashed lines indicate $\pm 5\%$ deviations. Some red dotted lines may not be visible if all data lie inside the upper or lower boundary. Some black dashed lines may not be visible if the data points for n = 1 are at the lower ends of the data ranges. Note that lines connecting data points in all figures presented here by no means imply any relationships nor trends, but they are only used for clarity. The values of n are shown on the x-axes which are neither a linear nor logarithmic scale. The range of n values is selectively investigated because it sufficiently represents the performance of the algorithms. Also, as n becomes lower than 2^{-9} the performance in terms of the score improvement reduces and as n becomes higher than 10, the optimization algorithm diverges.

For the QNA-XEL Figure 5.44 shows that all test proteins have the lowest AveSIwhen n = 1, 2, or 6 and that all n > 1 cases yield better results than those of the n < 1 cases. While the unmodified case (n = 1) yields the best or the second best results in all proteins, most n > 1 cases are within the $\pm 5\%$ deviations which implies similar performance. For the QNA-XEL (*Hessian only*) Figure 5.45 shows that the



Figure 5.44: QNA-XEL: The average of the score improvement (AveSI) versus various values of n in a) 1d3z, b) 1ubq, c) dom6, d) hetr, and e) mdmi proteins. Lower values indicate greater improvements. The red dotted lines above and below black dashed horizontal lines indicate 5% deviations above and below the AveSI of the n = 1 cases.



Figure 5.45: QNA-XEL (*Hessian only*): The average of the score improvement (*AveSI*) versus various values of n in a) 1d3z, b) 1ubq, c) dom6, d) hetr, and e) mdmi proteins. Lower values indicate greater improvements. The red dotted lines above and below black dashed horizontal lines indicate 5% deviations above and below the AveSI of the n = 1 cases.



Figure 5.46: QNA-XEL: The percent difference between the average score improvement of the various values of n and those of unmodified energy landscape (n = 1)case, $\Delta AveSI_n$. Negative values indicate that the XEL yields better results than the unmodified case. Lower values indicate greater improvements.

test proteins have the best results when n = 4, 6, or 10 and all but one n > 1 cases yield better results than those of n < 1 and n = 1 cases. In fact, proteins 1ubq, 1d3z, dom6, and hetr have more than 5% lower AveSI than n = 1.

Note that when n = 1 both the QNA-XEL and the QNA-XEL (*Hessian only*) become the QNA and therefore they should yield the same results. However, results may be slightly different since the results of n = 1 shown in each algorithm are from different runs and there is the probabilistic nature of the methods used in the perturbation in the refinement protocol. This also applies to the BFGS-XEL and the BFGS-XEL (*Hessian only*).

Figures 5.46 and 5.47 display the percent difference of the average score improvement compared to the unmodified case $\Delta AveSI_n$ for the QNA-XEL and the QNA-XEL (*Hessian only*), respectively. Negative values indicate that the XEL yields better



Figure 5.47: QNA-XEL (*Hessian only*): The percent difference between the average score improvement of the various values of n and those of unmodified energy landscape (n = 1) case, $\Delta AveSI_n$ versus various values of n. Negative values indicate that the XEL yields better results than the unmodified case. Lower values indicate greater improvements.

results than the unmodified case. Lower values indicate greater score improvements. For the QNA-XEL Figure 5.46 illustrates consistent results throughout all test proteins as n < 1 yields worse results than $n \ge 1$ and n > 1 yields results within $\pm 5\%$ of those from n = 1. No correlation between the protein size and the value of $\% \Delta AveSI_n$ can be established. For the QNA-XEL (*Hessian only*) Figure 5.47 illustrates consistent results throughout all test proteins which are similar to those from the QNA-XEL but n > 1 generally yields lower results than n = 1 and in some cases more than 10% lower. For n > 2 the value of $\% \Delta AveSI_n$ seems dependent on the protein size as larger proteins yield lower $\% \Delta AveSI_n$ except for the protein mdmi.



Figure 5.48: BFGS-XEL: The average of the score improvement (AveSI) versus various values of n in a) 1d3z, b) 1ubq, c) dom6, d) hetr, and e) mdmi proteins. Lower values indicate greater improvements. The red dotted lines above and below black dashed horizontal lines indicate 5% deviations above and below the AveSI of the n = 1 cases.



Figure 5.49: BFGS-XEL (*Hessian only*): The average of the score improvement (*AveSI*) versus various values of n in a) 1d3z, b) 1ubq, c) dom6, d) hetr, and e) mdmi proteins. Lower values indicate greater improvements. The red dotted lines above and below black dashed horizontal lines indicate 5% deviations above and below the AveSI of the n = 1 cases.

5.2.2.2 BFGS-XEL and BFGS-XEL (Hessian only)

Figures 5.48 and 5.49 display AveSI versus various values of n for each protein for the BFGS-XEL and the BFGS-XEL (*Hessian only*), respectively. The black dashed horizontal lines indicate the AveSI from the n = 1 cases and the red dotted lines above and below the black dashed lines indicate $\pm 5\%$ deviations. Some red dotted lines may not be visible if all data lie inside the upper or lower boundary. Lower values indicate greater improvement.

For the BFGS-XEL Figure 5.48 shows that all test proteins have the lowest AveSI when n = 2, 4, or 6 and that most n > 1 cases yield better results than those of the n < 1 cases and lie within $\pm 5\%$ boundaries of n = 1 which implies similar performance. For the BFGS-XEL (*Hessian only*) Figure 5.49 shows that the test proteins have the best results when n = 2 or 4 and most n < 1 cases yield worse results than those of $n \ge 1$. In fact, all but one n < 1 case yields worse results than those of the n = 1 cases.

Figures 5.50 and 5.51 display the percent difference of the average score improvement compared to the unmodified case $\% \Delta AveSI_n$ for the BFGS-XEL and the BFGS-XEL (*Hessian only*), respectively. For the BFGS-XEL Figure 5.50 illustrates consistent results throughout all test proteins as n < 1 yields worse results than $n \ge 1$ and most n > 1 yields results within $\pm 5\%$ or lower than those from n = 1. For n > 2the value of $\% \Delta AveSI_n$ tends to be dependent on the protein size as larger proteins yield lower $\% \Delta AveSI_n$ except for protein mdmi. For the BFGS-XEL (*Hessian only*) Figure 5.51 illustrates consistent results throughout all test proteins which are similar to those from the BFGS-XEL but all n > 1 cases yields results within $\pm 5\%$ of those from n = 1. No correlation between the protein size and the value of $\% \Delta AveSI_n$ can be established.



Figure 5.50: BFGS-XEL: The percent difference between the average score improvement of the various values of n and those of unmodified energy landscape (n = 1)case, $\Delta AveSI_n$. Negative values indicate that the XEL yields better results than the unmodified case. Lower values indicate greater improvements.

5.2.2.3 Conclusion of score improvement

The AveSI from configurations generated by the QNA-XEL, the QNA-XEL (Hessian only), the BFGS-XEL, and the BFGS-XEL (Hessian only) with different values of n in proteins 1ubq, 1d3z, dom6, hetr, and mdmi are compared. Results show that for all algorithms n > 1 yields better quality than n < 1 and in many cases yield better quality than n = 1. The same conclusion can also be drawn from results in Figures A.2 through A.8 a) through c), g), and h) in Appendix A which present the percentage of decoys with lower, equal, or higher score than those of n = 1 for all test algorithms. This is also in agreement with a conclusion from Section 5.1 and Guideline *iv.* in Table 4.1.

Note that although results in this section show that n < 1 yields worse quality than n = 1, the worst case has only 15% higher AveSI.



Figure 5.51: BFGS-XEL (*Hessian only*): The percent difference between the average score improvement of the various values of n and those of unmodified energy landscape (n = 1) case, $\Delta AveSI_n$. Negative values indicate that the XEL yields better results than the unmodified case. Lower values indicate greater improvements.

5.2.3 Lowest Score

In addition to performance in terms of quality, the lowest score indicates performance in terms of global search performance or ability to find the global solution. Since results can not be directly compared to the native state or the global solution, the lowest score E_{lowest} among the final scores of all decoys are evaluated and compared. Additionally, the percent difference of the lowest score $\%\Delta E_{\text{lowest}}$ compared to n = 1is determined which is defined as

$$\%\Delta E_{\text{lowest}} = \frac{(E_{\text{lowest}})_n - (E_{\text{lowest}})_{n=1}}{|(E_{\text{lowest}})_{n=1}|} \times 100$$
(5.4)

where $(E_{\text{lowest}})_n$ is the lowest score of the XEL case and $(E_{\text{lowest}})_{n=1}$ is the lowest score of the unmodified case. Lower values of E_{lowest} and $\%\Delta E_{\text{lowest}}$ imply better quality and global search performance.

5.2.3.1 QNA-XEL and QNA-XEL (Hessian only)

The lowest scores from different values of n in proteins 1ubq, 1d3z, dom6, hetr, and mdmi are compared in Figures 5.52 and 5.53 for the QNA-XEL and the QNA-XEL (*Hessian only*), respectively. The red dotted lines indicate $\pm 5\%$ deviations of the lowest scores from the n = 1 cases indicated by black dashed horizontal lines. Some red dotted lines may not be visible if all data lie inside the upper or lower boundary. Some black dashed lines may not be visible if the data points for n = 1 are at the lower ends of the data ranges.

For the QNA-XEL Figure 5.52 shows that three of five test proteins have the lowest score when n = 10 and the others have the lowest score when $n = 2^{-3}$ or 10. The deviation of the lowest score between different values of n is higher than that of the AveSI but on average in most proteins n > 1 has lower lowest score than n < 1 which agrees with results presented in the previous section. For the QNA-XEL (Hessian only) Figure 5.53 shows that two test proteins have the best results when n = 1 and the others have the lowest score when $n = 2^{-7}$, 6, or 10. Again, the deviation of the lowest score between different values of n is higher than that of the AveSI but on average in most protein n > 1 has lower lowest score than n < 1.

Figures 5.54 and 5.55 display the percent difference of the lowest score $\%\Delta E_{\text{lowest}}$ compared to n = 1 for the QNA-XEL and the QNA-XEL (*Hessian only*), respectively. Negative values indicate that the XEL yields better results than the unmodified case. Lower values indicate better performance. For the QNA-XEL Figure 5.54 illustrates that in all test proteins $n \neq 1$ yields worse results than n = 1 on average but three cases yield significantly better results than others. In protein hetr when n = 6, 8, or 10 the lowest score can be more than 20% lower than that of n = 1. The range of $\%\Delta E_{\text{lowest}}$ is between -31% to 28%, most values fall within ±15% range. No correlation between the protein size and the value of $\%\Delta AvesI_n$ can be established. For the QNA-XEL (*Hessian only*) Figure 5.55 illustrates that in all test proteins



Figure 5.52: QNA-XEL: The lowest scores (E_{lowest}) versus various values of n in a) 1ubq, b) 1d3z, c) dom6, d) hetr, and e) mdmi. Lower values indicate better performance. The red dotted lines above and below black dashed horizontal lines indicate 5% deviations above and below the lowest scores from the n = 1 cases.



Figure 5.53: QNA-XEL (*Hessian only*): The lowest scores (E_{lowest}) versus various values of n in a) 1ubq, b) 1d3z, c) dom6, d) hetr, and e) mdmi. Lower values indicate better performance. The red dotted lines above and below black dashed horizontal lines indicate 5% deviations above and below the lowest scores from the n = 1 cases.



Figure 5.54: QNA-XEL: The percent difference of the lowest scores ($\%\Delta E_{\text{lowest}}$) compared to n = 1 in a) 1ubq, b) 1d3z, c) dom6, d) hetr, and e) mdmi. Lower values indicate better performance.

 $n \neq 1$ yields worse results than n = 1 on average. The worst cases of the QNA-XEL (*Hessian only*) have worse results than those in the QNA-XEL. The range of $\% \Delta E_{\text{lowest}}$ is between -31% to 88%, most values fall within ±20% range. No correlation between the protein size and the value of $\% \Delta E_{\text{lowest}}$ can be established.

5.2.3.2 BFGS-XEL and BFGS-XEL (Hessian only)

The lowest scores from different values of n in proteins 1ubq, 1d3z, dom6, hetr, and mdmi are compared in Figures 5.56 and 5.57 for the BFGS-XEL and the BFGS-XEL (*Hessian only*), respectively. The red dotted lines indicate $\pm 5\%$ deviations of the lowest scores from the n = 1 cases indicated by black dashed horizontal lines. Some red dotted lines may not be visible if all data lie inside the upper or lower boundary.

For the BFGS-XEL Figure 5.56 shows that three of five test proteins have the lowest score when n = 4 and the others have the lowest score when n = 2 or 8. The



Figure 5.55: QNA-XEL (*Hessian only*): The percent difference of the lowest scores $(\%\Delta E_{\text{lowest}})$ compared to n = 1 in a) 1ubq, b) 1d3z, c) dom6, d) hetr, and e) mdmi. Lower values indicate better performance.

deviation of the lowest score between different values of n is higher than that of the AveSI but on average in most proteins n > 1 yield slightly lower lowest score than n < 1 which agrees with results presented in the previous section. For the BFGS-XEL (*Hessian only*) Figure 5.57 shows that each test protein has the lowest score when $n \neq 1$. Again, the deviation of the lowest score between different values of n is higher than that of the AveSI but for proteins 1d3z, hetr, and mdmi, most lowest scores are lower or inside the boundaries.

Figures 5.58 and 5.59 display the percent difference of the lowest score $\%\Delta E_{\text{lowest}}$ compared to n = 1 for the BFGS-XEL and the BFGS-XEL (*Hessian only*), respectively. Negative values indicate that the XEL yields better results than the unmodified case. Lower values indicate better performance. For the BFGS-XEL Figure 5.58 illustrates that in all test proteins $n \neq 1$ yields worse results than n = 1 on average but one case yields significantly better results than others. In protein hetr when



Figure 5.56: BFGS-XEL: The lowest scores (E_{lowest}) versus various values of n in a) 1ubq, b) 1d3z, c) dom6, d) hetr, and e) mdmi. Lower values indicate better performance. The red dotted lines indicate $\pm 5\%$ deviations of the lowest scores from the n = 1 cases.



Figure 5.57: BFGS-XEL (*Hessian only*): The lowest scores (E_{lowest}) versus various values of n in a) 1ubq, b) 1d3z, c) dom6, d) hetr, and e) mdmi. Lower values indicate better performance. The red dotted lines indicate $\pm 5\%$ deviations of the lowest scores from the n = 1 cases.



Figure 5.58: BFGS-XEL: The percent difference of the lowest scores ($\%\Delta E_{\text{lowest}}$) compared to n = 1 in a) 1ubq, b) 1d3z, c) dom6, d) hetr, and e) mdmi. Lower values indicate better performance.

n = 8 the lowest score can be almost 30% lower than that of n = 1. The range of $\%\Delta E_{\text{lowest}}$ is between -30% to 31%, most values fall within -10% to 20% range. No correlation between the protein size and the value of $\%\Delta AveSI_n$ can be established. For the BFGS-XEL (*Hessian only*) Figure 5.59 illustrates that in all test proteins $n \neq 1$ yields about the same results than n = 1 on average. The worst cases of the BFGS-XEL (*Hessian only*) have better results than those in the BFGS-XEL. The range of $\%\Delta E_{\text{lowest}}$ is between -26% to 25%, most values fall within ±15% range. No correlation between the protein size and the value of $\%\Delta E_{\text{lowest}}$ can be established.

5.2.3.3 Conclusion of lowest score

The lowest score from configurations generated by the QNA-XEL, the QNA-XEL (*Hessian only*), the BFGS-XEL, and the BFGS-XEL (*Hessian only*) with different values of n in proteins 1ubq, 1d3z, dom6, hetr, and mdmi are compared. On average



Figure 5.59: BFGS-XEL (*Hessian only*): The percent difference of the lowest scores $(\%\Delta E_{\text{lowest}})$ compared to n = 1 in a) 1ubq, b) 1d3z, c) dom6, d) hetr, and e) mdmi. Lower values indicate better performance.

 $n \neq 1$ yields higher lowest scores than n = 1 which could be up to 38% higher and n < 1 yields higher lowest score than n > 1.

Table 5.4 summarizes the lowest score for each protein along with the corresponding n value for all test algorithms. All but two cases are the XEL cases, twelve out of all twenty cases are n > 1, and five cases are n = 10. This indicates a positive effect of the XEL especially with n > 1 or n = 10 on finding global solutions. Comparing all algorithms, the BFGS-XEL and the BFGS-XEL (*Hessian only*) are the best in finding global solutions as each gives configurations with the lowest score in two proteins.

5.2.4 Average Iterations

For the evaluation of speed, the number of total iterations during the minimization process is determined instead of the actual time of the iterations. This is because Rosetta outputs the time for the entire process while only the time for the analytical

	Quantities	1ubq (76)	1d3z (82)	dom6 (126)	hetr (149)	mdvi (304)
QNA-XEL	Lowest Score	-38.17	-28.58	-38.32	-25.10	-62.68
	n	10	2^{-3}	10	10	2^{-7}
QNA-XEL	Lowest Score	-39.76	-30.62	-36.34	-29.15	-63.12
(Hessian only)		1	6	2^{-7}	10	1
BFGS-XEL	Lowest Score	-39.12	-30.82	-37.86	-30.26	-59.64
	n	2	4	4	8	4
BFGS-XEL	Lowest Score	-37.39	-28.61	-39.69	-26.10	-66.48
(Hessian only)	n	2^{-1}	2	10	2^{-5}	2^{-3}

Table 5.4: The *n* case in which the lowest score of each protein is found.

minimization is of the interest and it was not possible to separate this out. In addition, the numbers of total iterations are independent of the CPU speed which varies by computer.

Average iteration AveIt is the average of these numbers over the number of decoys N,

$$AveIt = \frac{1}{N} \sum_{i=1}^{N} m_i \tag{5.5}$$

where m_i is the total iterations for the i^{th} decoy. The results from XEL cases are compared to those from unmodified cases by calculating differences between the *AveIt* values of unmodified cases and those of the XEL cases relative to the *AveIt* values of unmodified cases. The percent difference of the average iteration compared to unmodified case $\% \Delta AveIt$ is defined as

$$\% \Delta AveIt_n = \frac{AveIt_n - AveIt_{n=1}}{|AveIt_{n=1}|} \times 100$$
(5.6)

where $AveIt_n$ is the AveIt value of the XEL case and $AveIt_{n=1}$ is the AveIt value of unmodified case. A negative $\% \Delta AveIt_n$ indicates that the XEL case has a smaller AveIt value than the unmodified case and therefore a better result. This means simulations in the XEL case run on average with fewer total iterations.

5.2.4.1 QNA-XEL and QNA-XEL (Hessian only)

The AveIt from different values of n in proteins 1ubq, 1d3z, dom6, hetr, and mdmi are compared in Figures 5.60 and 5.61 for the QNA-XEL and the QNA-XEL (Hessian only), respectively. A lower value implies higher speed. The red dotted lines above and below black dashed horizontal lines indicate 5% deviations above and below the average iteration of the n = 1 case. Some red dotted lines may not be visible if all data lie inside the upper or lower boundary. Some black dashed lines may not be visible if the data points for n = 1 are at the higher ends of the data ranges.

For the QNA-XEL Figure 5.60 shows that in most cases the XEL requires fewer iterations than the unmodified case. On average n < 1 cases require slightly fewer iterations than n > 1 cases. When n < 1 AveIt decreases as n decreases but when $n < 2^{-3}$ it decreases at a lower rate. When n > 1 AveIt decreases as n increases. Almost all proteins have the maximum AveIt when n = 1 but protein hetr has the maximum AveIt when n = 2. For the QNA-XEL (Hessian only) Figure 5.61 shows results similar to the QNA-XEL but when n > 1 AveIt decreases at a slower rate as n increases. For almost all proteins, the maximum AveIt is at n = 2, but for protein 1ubq the maximum AveIt is at n = 1.

The $\%\Delta AveIt_n$ from different values of n in proteins 1ubq, 1d3z, dom6, hetr, and mdmi are compared in Figures 5.62 and 5.63 for the QNA-XEL and the QNA-XEL (*Hessian only*), respectively. A lower value implies higher speed. For the QNA-XEL Figure 5.62 demonstrates no correlation between the size of proteins and the $\%\Delta AveIt_n$ values. The results also show that the XEL can reduce the number of iterations by up to 40% when n < 1 and up to 30% when n > 1. For the QNA-XEL (*Hessian only*) Figure 5.63 shows results similar to the QNA-XEL but when n > 1the $\%\Delta AveIt_n$ is higher than that in the QNA-XEL on average. The results also show that the XEL can reduce the number of iterations by up to 47% when n < 1and up to 22% when n > 1.



Figure 5.60: QNA-XEL: Average iterations (AveIt) versus various values of n in a) 1d3z, b) 1ubq, c) dom6, d) hetr, and e) mdmi. Lower numbers mean simulations run with fewer iterations. The red dotted lines above and below black dashed horizontal lines indicate 5% deviations above and below the average iteration of the n = 1 case.



Figure 5.61: QNA-XEL (*Hessian only*): Average iterations (*AveIt*) versus various values of n in a) 1d3z, b) 1ubq, c) dom6, d) hetr, and e) mdmi. Lower numbers mean simulations run with fewer iterations. The red dotted lines above and below black dashed horizontal lines indicate 5% deviations above and below the average iteration of the n = 1 case.



Figure 5.62: QNA-XEL: The percent difference of average iterations ($\% \Delta AveIt_n$) compared to the unmodified case (n = 1) versus various values of n. Lower values mean simulations run with fewer average iterations.

5.2.4.2 BFGS-XEL and BFGS-XEL (Hessian only)

The AveIt from different values of n in proteins 1ubq, 1d3z, dom6, hetr, and mdmi are compared in Figures 5.64 and 5.65 for the BFGS-XEL and the BFGS-XEL (Hessian only), respectively. A lower value implies higher speed. The red dotted lines above and below black dashed horizontal lines indicate 5% deviations above and below the average iteration of the n = 1 case. Some red dotted lines may not be visible if all data lie inside the upper or lower boundary. Some black dashed lines may not be visible if the data points for n = 1 are at the higher ends of the data ranges.

For the BFGS-XEL Figure 5.64 shows that in most cases the XEL requires fewer iterations than the unmodified case. On average n < 1 cases require slightly fewer iterations than n > 1 cases. When n < 1 AveIt decreases as n decreases, but when n > 1 AveIt decreases as n increases. Almost all proteins have the maximum AveIt



Figure 5.63: QNA-XEL (*Hessian only*): The percent difference of average iterations $(\% \Delta AveIt_n)$ compared to the unmodified case (n = 1) versus various values of n. Lower values mean simulations run with fewer average iterations.

when n = 2 but protein 1ubq has a maximum AveIt when n = 1. For the BFGS-XEL (Hessian only) Figure 5.65 shows results similar to the BFGS-XEL but when n > 1AveIt decays at a slower rate as n increases. For smaller proteins with the number of residues less than 100, 1ubq and 1d3z, the maximum AveIt's are at n = 1, but for the larger proteins the maximum are at n = 2. For protein mdmi when n > 1 AveIt is higher than n = 1.

The $\%\Delta AveIt_n$ from different values of n in proteins 1ubq, 1d3z, dom6, hetr, and mdmi are compared in Figures 5.66 and 5.67 for the BFGS-XEL and the BFGS-XEL (*Hessian only*), respectively. A lower value implies higher speed. For the BFGS-XEL Figure 5.66 demonstrates some correlation between the size of proteins and the $\%\Delta AveIt_n$ values. In most n cases, the graphs of 1d3z and 1ubq, which have the smallest numbers of residues, frequently exhibit the most improvement, and the graph of mdmi, which has the largest number of residues, frequently exhibits the



Figure 5.64: BFGS-XEL: Average iterations (*AveIt*) versus various values of n in a) 1d3z, b) 1ubq, c) dom6, d) hetr, and e) mdmi. Lower numbers mean simulations run with fewer iterations. The red dotted lines above and below black dashed horizontal lines indicate 5% deviations above and below the average iteration of the n = 1 case.



Figure 5.65: BFGS-XEL (*Hessian only*): Average iterations (*AveIt*) versus various values of n in a) 1d3z, b) 1ubq, c) dom6, d) hetr, and e) mdmi. Lower numbers mean simulations run with fewer iterations. The red dotted lines above and below black dashed horizontal lines indicate 5% deviations above and below the average iteration of the n = 1 case.



Figure 5.66: BFGS-XEL: The percent difference of average iterations ($\% \Delta AveIt_n$) compared to the unmodified case (n = 1) versus various values of n. Lower values mean simulations run with fewer average iterations.

least improvement. This suggests $\% \Delta AveIt_n$ values may increase as the protein sizes increase. The results also show that the XEL can reduce the number of iterations by up to 40% when n < 1 and up to 34% when n > 1. For the BFGS-XEL (*Hessian* only) Figure 5.67 shows results similar to the BFGS-XEL but when n = 8 and 10 the $\% \Delta AveIt_n$ are higher than those in the BFGS-XEL. The results also show that the XEL can reduce the number of iterations by up to 40% when n < 1 and up to 20% when n > 1.

5.2.4.3 Conclusion of average iterations

Results from all algorithms are quite similar as most cases have a lower number of iterations than n = 1 and n < 1 has a lower number of iterations than n > 1. This implies that the XEL results in higher speed with n < 1 yielding higher speed than n > 1 and that the XEL has consistent effects on the number of iterations



Figure 5.67: BFGS-XEL (*Hessian only*): The percent difference of average iterations $(\% \Delta AveIt_n)$ compared to the unmodified case (n = 1) versus various values of n. Lower values mean simulations run with fewer average iterations.

in the QNA-XEL, the QNA-XEL (*Hessian only*), the BFGS-XEL, and the BFGS-XEL (*Hessian only*). However, this finding is not consistent with that in Section 5.1.1 which shows that with the bracketing and Brent line search the QNA-XEL, the QNA-XEL (*Hessian only*), the BFGS-XEL, and the BFGS-XEL (*Hessian only*) have slightly higher speed when n > 1. This inconsistency may be because the simulated energy landscapes do not represent the score function of a protein molecule.

Reducing the number of iterations by up to 40% roughly means reducing the runtime by 40% because the runtime for the BFGS and the BFGS-XEL with n = 1 are roughly the same. This is very beneficial since the protein conformation prediction process can take several months even for a small protein. For example, a high-resolution structure prediction algorithm [11] refining 20,000 to 30,000 models of several 49 to 88 residue long molecules takes about 150 CPU days per molecule. 40% reduction means the runtime reduces by 60 CPU days.

5.2.5 Efficiency

To consider both score improvement and total iterations in the performance evaluation, efficiency, e_n , is defined as the value of average score improvement per thousand iterations,

$$e_n = \frac{AveSI_n}{AveIt_n/1000} \tag{5.7}$$

Since negative $AveSI_n$'s mean results are improved from input decoys, the negative efficiency indicates the improvement rate and lower values mean higher rates. In all simulations presented here the AveSIs are negative so all values for e are negative.

The percent differences between the values of efficiency in the various values of nand those in the unmodified cases, $\%\Delta e_n$, are displayed in Figure 5.75,

$$\%\Delta e_n = \frac{e_n - e_{n=1}}{|e_{n=1}|} \times 100 \tag{5.8}$$

where $e_{n=1}$ is the efficiency of the unmodified case. Lower values indicate that the XEL cases have better performance than the unmodified case. This means the same number of iterations in XEL-implemented simulations yields more score improvement.

5.2.5.1 QNA-XEL and QNA-XEL (Hessian only)

The efficiency e_n from different values of n in proteins 1ubq, 1d3z, dom6, hetr, and mdmi are compared in Figures 5.68 and 5.69 for the QNA-XEL and the QNA-XEL (*Hessian only*), respectively. A lower value implies better efficiency. The red dotted lines indicate $\pm 5\%$ deviations of the efficiency from the n = 1 cases indicated by black dashed horizontal lines. Some red dotted lines may not be visible if all data lie inside the upper or lower boundary. Some black dashed lines may not be visible if the data points for n = 1 are at the higher ends of the data ranges.

For the QNA-XEL Figure 5.68 shows that efficiency is dependent on the size of the protein as it decreases when the protein chain length increases. This means a larger protein results in better efficiency. The better efficiency in a larger protein is because



Figure 5.68: QNA-XEL: The efficiency (e_n) versus various values of n in a) 1ubq, b) 1d3z, c) dom6, d) hetr, and e) mdmi. Lower values mean better performance. The red dotted lines above and below black dashed horizontal lines indicate 5% deviations above and below the efficiency of the n = 1 case.


Figure 5.69: QNA-XEL (*Hessian only*): The efficiency (e_n) versus various values of n in a) 1ubq, b) 1d3z, c) dom6, d) hetr, and e) mdmi. Lower values mean better performance. The red dotted lines above and below black dashed horizontal lines indicate 5% deviations above and below the efficiency of the n = 1 case.



Figure 5.70: QNA-XEL: The percent difference of the efficiency $(\% \Delta e_n)$ compared to those in the unmodified cases (n = 1) versus various values of n. Lower values mean better performance.

a larger gap of scores between an input decoy and a semi-folded configuration. An input decoy of a larger protein is more likely to be a bad approximation because of a larger conformational space and thus a refinement protocol can significantly improve the score which yields better efficiency. On average n < 1 cases have better efficiency than n > 1. When n < 1 efficiency is improved as n decreases but not at $n = 2^{-9}$. When n > 1 efficiency is also improved as n increases but at a lower rate. Proteins 1ubq, 1d3z, and dom6 have the worst efficiency when n = 1 and proteins hetr and mdmi have the worst efficiency when n = 2. For the QNA-XEL (*Hessian only*) Figure 5.69 show results similar to those in the QNA-XEL but with better efficiency.

The $\%\Delta e_n$ from different values of n in proteins 1ubq, 1d3z, dom6, hetr, and mdmi are compared in Figures 5.70 and 5.71 for the QNA-XEL and the QNA-XEL (*Hessian only*), respectively. A lower value implies better efficiency.

For the QNA-XEL Figures 5.70 shows that the efficiency can be up to 52% better



Figure 5.71: QNA-XEL (*Hessian only*): The percent difference of the efficiency $(\% \Delta e_n)$ compared to those in the unmodified cases (n = 1) versus various values of n. Lower values mean better performance.

than that of n = 1 which means with the same number of iterations as n = 1 the score improvement can be lower by 50%. When n > 1 the efficiency can be up to 40% better than that of n = 1. No correlation between the size of proteins and the values of $\%\Delta e_n$ can be found but a smaller protein tends to have higher improvement in efficiency than a larger protein. For the QNA-XEL (*Hessian only*) Figures 5.71 shows results similar to those of the QNA-XEL but efficiency can be up to 75% better than that of n = 1. When n > 1 efficiency can be up to 30% better than that of n = 1. Some correlation between the size of proteins and the values of $\%\Delta e_n$ can be seen as a smaller protein has higher improvement in efficiency than a larger protein when $n < 2^{-1}$ or n > 6 except for protein hetr.

5.2.5.2 BFGS-XEL and BFGS-XEL (Hessian only)

The e_n from different values of n in proteins 1ubq, 1d3z, dom6, hetr, and mdmi are compared in Figures 5.72 and 5.73 for the BFGS-XEL and the BFGS-XEL (*Hessian* only), respectively. A lower value implies better efficiency. The red dotted lines indicate $\pm 5\%$ deviations of efficiency from the n = 1 cases indicated by black dashed horizontal lines. Some red dotted lines may not be visible if all data lie inside the upper or lower boundary. Some black dashed lines may not be visible if the data points for n = 1 are at the higher ends of the data ranges.

For the BFGS-XEL Figure 5.72 shows that the value of efficiency is dependent on the size of the protein as the value decreases when the protein chain length increases. On average the n < 1 case has better efficiency than the n > 1 case in proteins 1ubq, 1d3z, and dom6 but both cases have about the same efficiency in proteins hetr and mdmi. When n < 1 efficiency is improved as n decreases but when n > 1efficiency is improved as n increases. The worst efficiency is when n = 1 or 2. For the BFGS-XEL (*Hessian only*) Figure 5.73 show results slightly different from those of the BFGS-XEL. While in the BFGS-XEL (*Hessian only*) efficiency is also improved as the protein chain length increases, n < 1 case has much better efficiency than n > 1 case.

The $\%\Delta e_n$ from different values of n in proteins 1ubq, 1d3z, dom6, hetr, and mdmi are compared in Figures 5.74 and 5.75 for the BFGS-XEL and the BFGS-XEL (*Hessian only*), respectively. A lower value implies better efficiency.

For the BFGS-XEL Figure 5.74 shows that efficiency can be up to 50% better than that of n = 1 and when n > 1 efficiency can be up to 35% better than that of n = 1. No correlation between the size of proteins and the values of $\%\Delta e_n$ can be found but a smaller protein tends to have better efficiency than a larger protein. For the BFGS-XEL (*Hessian only*) Figure 5.75 shows results similar to those of the BFGS-XEL but efficiency can be up to 52% better than that of n = 1 and when



Figure 5.72: BFGS-XEL: efficiency (e_n) versus various values of n in a) 1ubq, b) 1d3z, c) dom6, d) hetr, and e) mdmi. Lower values mean better performance. The red dotted lines above and below black dashed horizontal lines indicate 5% deviations above and below the efficiency of the n = 1 case.



Figure 5.73: BFGS-XEL (*Hessian only*): efficiency (e_n) versus various values of n in a) 1ubq, b) 1d3z, c) dom6, d) hetr, and e) mdmi. Lower values mean better performance. The red dotted lines above and below black dashed horizontal lines indicate 5% deviations above and below the efficiency of the n = 1 case.



Figure 5.74: BFGS-XEL: The percent difference of the efficiency $(\%\Delta e_n)$ compared to those in the unmodified cases (n = 1) versus various values of n. Lower values mean better performance.

n > 1 efficiency can be up to 20% better than that of n = 1. No correlation between the size of proteins and the values of $\% \Delta e_n$ can be seen but a smaller protein tends to have better efficiency than a larger protein.

5.2.5.3 Conclusion of efficiency

Results show that the XEL can improve the efficiency in the QNA-XEL by up to 52%, in the QNA-XEL (*Hessian only*) by up to 75%, in the BFGS-XEL by up to 50%, and in the BFGS-XEL (*Hessian only*) by up to 52%. In all algorithms efficiency is dependent on the size of the protein as it is improved when the protein chain length increases. On average when n < 1 the efficiency is better than when $n \ge 1$ which implies that better efficiency can be achieved by using n < 1.



Figure 5.75: BFGS-XEL (*Hessian only*): The percent difference of the efficiency $(\% \Delta e_n)$ compared to those in the unmodified cases (n = 1) versus various values of n. Lower values mean better performance.

5.2.6 Similarity to the native configuration

In addition to quality, speed, and efficiency, the similarity of output configurations to the native conformation are evaluated in terms of TM score [88], GDT score [79, 80], MaxSub score [62], and RMSD [32, 33] using TM-score software package [86]. The RMSD (Relative Mean Square Deviation) value commonly used to evaluate the similarity is usually not sufficient [88] because the value depends on the protein length and percent alignment which is the compared portions of the output configuration and the native conformation. The TM, the GDT, and the MaxSub scores are developed to improve a correlation between high similarity and a high score. These scores are range between 0 and 1 and a value closer to 1 implies a configuration more similar to the native state. The TM score has a better correlation than the GDT and the MaxSub scores [88].



Figure 5.76: BFGS-XEL (*Hessian only*): The scores versus TM scores of resulting configurations with various values of n for protein 1ubq. Lower score means better quality. TM score closer to 1 means a configuration more similar to the native conformation.



Figure 5.77: BFGS-XEL (*Hessian only*): The scores versus GTD scores of resulting configurations with various values of n for protein 1ubq. Lower score means better quality. GTD score closer to 1 means a configuration more similar to the native conformation.



Figure 5.78: BFGS-XEL (*Hessian only*): The scores versus MaxSub scores of resulting configurations with various values of n for protein 1ubq. Lower score means better quality. MaxSub score closer to 1 means a configuration more similar to the native conformation.



Figure 5.79: BFGS-XEL (*Hessian only*): The scores versus RMSD values of resulting configurations with various values of n for protein 1ubq. Lower score means better quality. RMSD value closer to 0 means a configuration more similar to the native conformation.

Figures 5.76 through 5.79 present the TM scores, the GDT scores, the MaxSub scores, and the RMSD values of the resulting configurations of protein 1ubq with BFGS-XEL (*Hessian only*) and various values of n. Results from each value of n(shown as stars) are compared to those from n = 1 (shown as circles). A TM, a GDT, or a MaxSub score closer to 1 means a configuration more similar to the native conformation. A TM score greater than 0.4 implies a meaningful prediction. An RMSD value closer to 0 means a configuration more similar to the native conformation. Note that only protein 1ubq is evaluated because other proteins do not have the structural information of native conformations available.

Results in Figures 5.76 through 5.79 show that the TM scores, the GDT scores, the MaxSub scores, and the RMSD values from $n \neq 1$ and n = 1 are similar since the circles (n = 1) and the stars $(n \neq 1)$ cover similar areas. This means $n \neq 1$ yields configurations similar to the native conformation as those of n = 1. Figures 5.76 through 5.78 show that as n increases the stars move slightly downward and to the right. This implies that the quality of resulting configurations is improved as nincreases.

5.2.6.1 Conclusion of similarity to the native configuration

Overall results show that XEL has little effect on the similarity to the native configuration. Since XEL yields better speed, this means XEL would be expected to yield configurations with the same quality as those of the unmodified case in terms of similarity with fewer iterations.

5.2.7 Further analysis: Same-iteration comparison

The XEL algorithms yield lower quality than the unmodified cases as their average score improvement AveSI and lowest score E_{lowest} are typically higher than those for n = 1. However, these quantities could possibly be improved if XEL simulations are allowed to run with the same number of total iterations from all decoys as the unmodified cases but with more perturbations or decoys. For example, in protein 1ubq running simulations with 29 additional decoys yields the same number of total iterations as the n = 1 case. This claim is investigated on the BFGS-XEL (*Hessian* only) with $n = 2^{-9}$ and two proteins, 1ubq and dom6. As additional simulations are run, more decoys (Decoys+) or more perturbations (Perturb+) are added so that the resulting number of total iterations are nearly equal to the n = 1 case (less than 1% difference). The results from the runs with more decoys and perturbations are compared to the results from the BFGS-XEL (*Hessian only*) with $n = 2^{-9}$ presented in previous sections and referred to below as the $n = 2^{-9}$ run without additional iterations or $n = 2^{-9}$ (50) in figures.

Figures 5.80 presents a) the percent difference of the average score improvement $\%\Delta AveSI$, b) the percent difference of the lowest score $\%\Delta E_{lowest}$, c) the percent difference of the average iteration $\%\Delta AveIt$, and d) the percent difference of the efficiency $\%\Delta e_n$ from special runs with more decoys (Decoys+) and more perturbations (Perturb+) and the BFGS-XEL (*Hessian only*) without additional iterations. All runs use $n = 2^{-9}$ and are performed for proteins 1ubq and dom6. Lower percentage implies better performance.

Figures 5.80 a) shows that more decoys does not improve the values of $\% \Delta AveSI_n$ in the case of either protein but more perturbations improves the values of $\% \Delta AveSI_n$ in the case of both proteins by 9% and 6% from those in the $n = 2^{-9}$ run without additional iterations. This shows that the values of $\% \Delta AveSI_n$ is sensitive to the average number of total iterations per decoy (AveIt). Since adding more decoys does not change the AveIt value (Figure 5.80 c) and the values of $\% \Delta AveSI_n$ in the $n = 2^{-9}$ run without additional iterations is possibly a good representative value for that AveIt, more decoys has almost no effect on the values of $\% \Delta AveSI_n$. With more perturbations the values of $\% \Delta AveSI_n$ are improved since adding more perturbations increases the AveIt.



Figure 5.80: Same-Iteration: a) the percent difference of average score improvement $\% \Delta AveSI$, b) the percent difference of lowest score $\% \Delta E_{lowest}$, c) the percent difference of average iteration $\% \Delta AveIt$, and d) the percent difference of the efficiency $\% \Delta e_n$ from special runs with $n = 2^{-9}$ and either more decoys (Decoys+) or more perturbations (Perturb+) and the BFGS-XEL (*Hessian only*) without additional iterations. All runs use $n = 2^{-9}$ and are performed for proteins 1ubq and dom6. Lower percentage means better results.

Figure 5.80 b) shows that more decoys does not improve or worsen the values of $\%\Delta E_{\text{lowest}}$ in the case of either protein which suggests that adding more decoys would not change the lowest score. This result agrees with an additional test that runs 1000 decoys with $n = 2^{-9}$ and 614 decoys of n = 1. Results also show that more perturbations improves the value of $\%\Delta E_{\text{lowest}}$ in protein dom6 by 21% but worsens the value of $\%\Delta E_{\text{lowest}}$ in protein 1ubq by 5% from those in the $n = 2^{-9}$ run without additional iterations. Increasing the number of perturbations can yield a lower or a higher lowest score than those of the $n = 2^{-9}$ run without additional iterations because of the probabilistic nature of the perturbations. Unlike the more perturbation case with new simulations on all original 50 decoys, the more decoy case adds new simulations on additional 29 decoys to the $n = 2^{-9}$ run with original 50 decoys so the lowest score of the more decoy case can only be lower or equal to that of the original $n = 2^{-9}$ run.

Figure 5.80 c) shows that the values of $\% \Delta AveIt$ in the more decoy case do not change but in the more perturbation case the values of $\% \Delta AveIt$ increases to 0% which means AveIt of the more perturbation case is equal to that of n = 1. The change in the more perturbation case is because the number of total iterations is now nearly equal to that of n = 1.

Lastly, Figure 5.80 d) shows that more decoys do not significantly improve or worsen the values of $\%\Delta e_n$ from those in the $n = 2^{-9}$ run without additional iterations but more perturbations worsen the values of $\%\Delta e_n$ such that they are about equal to those for n = 1. While adding more decoys does not change efficiency, increasing perturbations does yield worse efficiency. As the simulation progresses efficiency becomes worsened because the values of the score improvement between consecutive iterations are most likely larger at the beginning of the simulation than those toward the end of the simulation. Thus, this suggests that more perturbations should not be added to a simulation for better efficiency.



Figure 5.81: BFGS-XEL (*Hessian only*): The scores versus the TM, the GTD, the MaxSub scores and the RMSD values of resulting configurations for protein 1ubq. These quantities are from special runs with $n = 2^{-9}$ and either more decoys (Decoys+) or more perturbations (Perturb+). They are compared to those of n = 1 without varying the numbers of original decoys or perturbations. Lower score means better quality. The TM, the GTD, or the MaxSub score closer to 1 means a configuration more similar to the native conformation. RMSD value closer to 0 means a configuration more similar to the native conformation.



Figure 5.82: BFGS-XEL (*Hessian only*): The scores versus the TM, the GTD, the MaxSub scores and the RMSD values of resulting configurations for protein 1ubq. These quantities are from special runs with $n = 2^{-9}$ and either more decoys (Decoys+) or more perturbations (Perturb+). They are compared to those of $n = 2^{-9}$ without varying the numbers of original decoys or perturbations. Lower score means better quality. The TM, the GTD, or the MaxSub score closer to 1 means a configuration more similar to the native conformation. RMSD value closer to 0 means a configuration more similar to the native conformation.

For similarity evaluation the TM, the GTD, the MaxSub scores and the RMSD values of the resulting configurations from special runs with more decoys and more perturbations for protein 1ubq are evaluated. Figures 5.81 and 5.82 compare these quantities from special runs with $n = 2^{-9}$ and either more decoys (Decoys+) or more perturbations (Perturb+) to those of the n = 1 case and the $n = 2^{-9}$ respectively for which the original numbers of decoys and perturbations are not altered. Results from special runs with more decoys (Decoys+) and more perturbations (Perturb+) are shown as triangles, those from n = 1 are shown as circles, and those from $n = 2^{-9}$ are shown as stars. A TM, a GDT, or a MaxSub score closer to 1 means a configuration more similar to the native conformation.

Figure 5.81 shows that with more decoys (Decoys+) the results are similar to those of n = 1 but cover slightly larger areas. With more perturbations (Perturb+) the results are also similar to those of n = 1. Figure 5.82 shows that the additional decoys do not have a high similarity and more perturbations do not improve or worsen the similarity in resulting configurations.

5.2.7.1 Conclusion of further analysis: same-iteration comparison

Overall adding more decoys does not change the quality, speed, or the efficiency of the BFGS-XEL (*Hessian only*) but increasing perturbations can improve quality in terms of the average score improvement and the lowest score while worsening speed and efficiency. Note that only one value of n and two proteins are used in this analysis. Running similar tests with other values of n and more proteins will increase the confidence of this conclusion.

5.2.8 Conclusion of XEL Algorithms with Probabilistic Search

The QNA-XEL, the QNA-XEL (*Hessian only*), the BFGS-XEL, and the BFGS-XEL (*Hessian only*) are implemented with the bracketing and Brent line search in an

optimization with a probabilistic search under the Rosetta platform. Simulations are run on 50 decoys from five proteins with 11 values of n including n = 1 which is the unmodified case. The results are evaluated in terms of quality, speed, and efficiency.

In terms of quality the average score improvement AveSI and the lowest scores E_{lowest} are determined. Results show that n > 1 yields a lower AveSI, which means higher quality, than n < 1 and sometimes than n = 1. This is in agreement with a conclusion from Section 5.1 and Guideline *iv.* in Table 4.1. A larger n has smaller steps so the matrix estimation becomes more accurate which results in better quality. However, when n is too large, the quality becomes unexpectedly worse. This is because with steps too small a relative change in the score becomes so small that it satisfies a stopping criterion before converging to a solution and results in a worse quality.

Results also show that on average $n \neq 1$ yields a higher E_{lowest} than n = 1 but some cases of $n \neq 1$ can yield a lower E_{lowest} than n = 1. A lower E_{lowest} implies higher quality and global search performance. The results are in agreement with those from Section 5.1. This suggests that a lower E_{lowest} can be achieved, but simply having a constant exponent n for the duration of the optimization may not allow a lower value of E_{lowest} to be achieved. The adaptive exponential energy landscaping (AXEL) that changes n during the course of simulation is presented in Section 5.3.

For speed and efficiency evaluation the average iteration AveIt and the efficiency e_n are determined. Results show that n < 1 yields a lower AveIt and a lower e_n than n > 1 which implies better speed and efficiency. This however does not agree with the results seen in Section 5.1 which show that with the bracketing and Brent line search n > 1 has slightly better speed than n < 1. This inconsistency may be because the simulated simpler energy landscapes do not represent the score function of a protein molecule.

n < 1 yields better speed because the steps become larger so n < 1 paths travel

Table 5.5: The range of the percent difference of the average iteration $\% \Delta AveIt$ yielded by different algorithms with different values of n. Lower values imply better speed. The ranges are displayed from worse to better speed. Negative value means better speed than n = 1 and positive value means worse speed.

n	QNA-XEL	QNA-XEL (<i>Hessian only</i>)	BFGS-XEL	BFGS-XEL (Hessian only)	Overall
	(%)	(%)	(%)	(%)	(%)
2^{-9}	-29 to -42	-34 to -47	-24 to -41	-24 to -40	-24 to -47
2^{-7}	-28 to -41	-34 to -44	-20 to -37	-21 to -36	-20 to -44
2^{-5}	-25 to -36	-32 to -42	-18 to -33	-15 to -28	-15 to -42
2^{-3}	-21 to -28	-27 to -34	-8 to -27	-12 to -22	-8 to -34
2^{-1}	-8 to -15	-14 to -19	-2 to -11	-2 to -11	-2 to -19
2	3 to -3	8 to -1	9 to -3	7 to -2	9 to -3
4	0 to -10	7 to -6	6 to -10	3 to -10	7 to -10
6	-3 to -14	3 to -8	1 to -17	2 to -18	3 to -18
8	-12 to -21	-7 to -15	-6 to -24	1 to -17	1 to -24
10	-18 to -28	-9 to -22	-15 to -32	1 to -20	1 to -32

to the minima faster. However, this reasoning does not explain why n > 1 yields better speed than n = 1. Keep in mind that n > 1 steps may be shorter but it does not imply that the number of iterations will increase. Since the matrix estimation is more accurate with n > 1, the path converges to the closest solution which means the path travels a much shorter distance so it does not require as many steps. Also with steps too small a relative change in the score becomes so small that a stopping criterion is satisfied before converging to a solution. Therefore, n > 1 yields better speed than n = 1.

Comparing results from the XEL and the XEL (*Hessian only*) shows that different magnitudes of the steps yield similar trends but some deviations can be seen. Although the term β^{-1} in the XEL (*Hessian only*) gives significantly different magnitudes, the results are similar because the bracketing and Brent line search reduces their effect.

Table 5.6: The range of the percent difference of the average score improvement $\% \Delta AveSI$ yielded by different algorithms with different values of n. Lower values imply higher quality. The ranges are displayed from lower to higher quality. Negative values mean higher quality than n = 1 and positive values mean lower quality.

n	QNA-XEL	QNA-XEL (Hessian only)	BFGS-XEL	BFGS-XEL (Hessian only)	Overall
	(%)	(%)	(%)	(%)	(%)
2^{-9}	15 to 7	11 to 7	13 to 6	13 to 4	15 to 4
2^{-7}	11 to 7	12 to 7	11 to 7	9 to 4	12 to 4
2^{-5}	13 to 6	10 to 6	13 to 4	12 to 4	13 to 4
2^{-3}	13 to 4	12 to 5	11 to 2	7 to 1	13 to 1
2^{-1}	10 to 5	8 to 3	10 to 1	6 to -1	10 to -1
2	4 to -3	1 to -4	-1 to -10	-1 to -4	4 to -10
4	4 to 0	-2 to -12	4 to -10	2 to -4	4 to -12
6	6 to -5	-5 to -12	1 to -9	4 to -1	6 to -12
8	5 to 0	-3 to -9	3 to -5	5 to -1	5 to -9
10	8 to 0	-1 to -10	8 to 1	4 to -2	8 to -10

Results shows that when the XEL is implemented a trade-off between quality and speed must be considered. Tables 5.5 and 5.6 are look-up tables for the range of the percent difference of the average iteration and the average score improvement yielded by different algorithms with different values of n. Table 5.5 shows that when the QNA-XEL is implemented $n = 2^{-9}$ yields the best speed which is 29% to 42% higher than the unmodified case; however, Table 5.6 shows that although an improvement in speed is achieved quality may be reduced by 7% to 15%. If the BFGS-XEL is implemented and higher quality is desired, Table 5.6 suggests that n = 2 should be used because it gives 1% to 10% higher quality, but the trade-off in speed may be up to 9% as shown in Table 5.5.

Results show that XEL significantly improves the efficiency of the analytical minimization, especially with the values of n in the $2^{-9}-2^{-5}$ range. Figures 5.70 and 5.71 show that efficiency in the QNA-XEL and the QNA-XEL (*Hessian only*) can be

Table 5.7: The average TM, GDT, MaxSub scores, and RMSD values for XEL with various values of n and two special runs with more decoys or perturbations. All results are for protein 1ubq and from BFGS-XEL (*Hessian only*). The two special runs use $n = 2^{-9}$. A TM, a GDT, or a MaxSub score closer to 1 means an average configuration more similar to the native conformation. An RMSD value closer to 0 means an average configuration more similar to the native conformation. The values corresponding to the best results are in boldface.

n	Score	TM score	GTD score	MaxSub score	RMSD
2^{-9}	-14.3	0.482	0.530	0.441	6.60
2^{-7}	-15.5	0.487	0.535	0.448	6.51
2^{-5}	-14.7	0.482	0.532	0.443	6.60
2^{-3}	-16.9	0.479	0.526	0.439	6.57
2^{-1}	-18.5	0.490	0.535	0.454	6.49
1	-18.5	0.482	0.533	0.437	6.41
2	-18.7	0.478	0.528	0.439	6.58
4	-18.4	0.481	0.532	0.440	6.56
6	-17.1	0.476	0.527	0.435	6.54
8	-16.9	0.482	0.532	0.441	6.52
10	-17.2	0.490	0.536	0.454	6.49
More decoys	-14.1	0.464	0.514	0.423	7.14
More perturbations	-17.3	0.472	0.523	0.431	6.57

improved by 25% to 51% and 32% to 75%. Figures 5.74 and 5.75 show that efficiency in the BFGS-XEL and the BFGS-XEL (*Hessian only*) can be improved by 13% to 49% and 13% to 52%.

Another measure of quality is the similarity between resulting configurations and the native conformation. The TM, the GTD, the MaxSub scores and the RMSD values of resulting configurations for protein 1ubq are evaluated. Results show that XEL does not improve or worsen the similarity to the native conformation in terms of the TM, the GDT, the MaxSub scores, and the RMSD values. Table 5.7 presents average TM, GDT, MaxSub scores, and RMSD values for XEL with various values

Table 5.8: The percent difference of average TM, GDT, MaxSub scores, and RMSD values compared to n = 1 for XEL with various values of n and two special runs with more decoys or perturbations. All results are for protein 1ubq and from BFGS-XEL (*Hessian only*). The two special runs use $n = 2^{-9}$. A negative value means better performance than n = 1.

n	Score (%)	TM score (%)	GTD score (%)	MaxSub score (%)	RMSD (%)
2^{-9}	22.7	0	0.56	-0.92	2.96
2^{-7}	16.2	-1.04	-0.38	-2.52	1.56
2^{-5}	20.5	0	0.19	-1.37	2.96
2^{-3}	8.65	0.62	1.31	-0.46	2.50
2^{-1}	0	-1.66	-0.38	-3.89	1.25
2	-1.08	0.83	0.94	-0.46	2.65
4	0.54	0.21	0.19	-0.69	2.34
6	7.57	1.24	1.13	0.46	2.03
8	8.65	0	0.19	-0.92	1.72
10	7.03	-1.66	-0.56	-3.89	1.25
More decoys	23.78	3.73	3.56	3.20	11.39
More perturbations	6.49	2.07	1.88	1.37	2.50

of n and two special runs with more decoys or perturbations as described in Section 5.2.7. Table 5.8 presents the percent difference of these values compared to n = 1. These values are calculated such that a negative value means better performance than n = 1.

Table 5.7 shows that n = 1 yields a better average score and an average RMSD value than the other runs, but it yields similar and sometimes lower average TM, GDT, and MaxSub scores than other runs. Table 5.8 shows that average TM, GDT, and MaxSub scores for $n \neq 1$, including the special runs, are within 4% of those for n = 1 and 90% of these quantities are within 2%. This means for $n \neq 1$ the configurations are similar to those for n = 1 with fewer iterations. Note that although these results give an insight about the similarity of the resulting configurations, results

from many more decoys will increase the confidence in this conclusion because of the probabilistic nature of the process.

While results show that n < 1 can yield up to 47% better speed than n = 1, it can yield up to 15% lower quality in terms of the average score improvement. The benefits from better speed however may exceed this drawback and may be more desirable. In a practical protein prediction, running refinements on more decoys is more preferable than running longer refinements the same decoys because it means the higher chance of locating the global minimum energy conformation. Moreover, since the scoring functions are far from perfect [57, 76], most of the successful resulting conformations are merely close to the native conformations, and sometimes the lowest energy results from simulations are discarded because they contain inaccurate or incorrect folds [11]. Since XEL has shown almost no effect on the similarity of resulting configurations, XEL can yield configurations as similar to the native conformation as the unmodified case with less conformation cost. This means that XEL can significantly improve the protein prediction simulation even though it yields less average score improvement.

Results from same-iteration comparison show that when $n = 2^{-9}$ increasing the number of decoys does not lower the average score improvement or the lowest score and suggest that increasing the perturbations is not desirable. While this implies that simply increasing the number of decoys does not improve the quality, the confidence in this conclusion is low because only one case of n and two proteins are investigated. A study on more proteins and with more cases of n is needed for a more definite conclusion.

To improve quality without sacrificing speed, the AXEL presented in Section 5.3 is investigated.

5.3 BFGS with AXEL

Since when n < 1 the landscape is flattened and when n > 1 the global minimum becomes more prominent, different values of the exponent n yield different effects on the performance of the XEL. This is reflected on results from Section 5.2 which show that n > 1 yields better quality than n < 1 but worse speed and efficiency. Therefore, the energy landscape should be dynamically modified as the optimization search progresses.

To improve the overall performance the AXEL is composed of two parts, the AXEL schemes that describe how n changes during the course of conformation prediction, and the XEL or the varying-n XEL algorithms that are used in minimization. The AXEL schemes are categorized by when the exponent n changes as described in Section 4.3.2. The AXEL is investigated with several AXEL schemes on the BFGS-XEL (*Hessian only*) and the varying-n BFGS-XEL algorithm. Sections 5.3.1 and 5.3.2 present AXEL results from global and local AXEL schemes, respectively. Lastly, Section 5.3.3 concludes the BFGS with AXEL section.

5.3.1 Global schemes

Global AXEL schemes change the exponent n depending on the global location of the probabilistic search but n's are constant during each minimization portion of the simulation. Referring to Figure 5.43 that presents the refinement protocol of Rosetta, the global AXEL schemes change n before the minimization that allows all dihedral angles on the entire chain to change, or the minimization that allows only perturbed dihedral angles to change, or both. Since n is constant during the minimization, XEL algorithms are used with the global AXEL schemes. In this section the following global AXEL schemes are investigated with the BFGS-XEL (*Hessian only*):

Scheme A) This scheme is used as the control case for the global scheme. Two cases of Scheme A are implemented. First, n is randomly modified six times at points

before the minimization that allows all dihedral angles on the entire chain to change. Second, n is randomly modified at points before the minimization that allows only perturbed dihedral angles to change. The results are presented in Section 5.3.1.1.

- Scheme B) The values of n to be used during the beginning, the middle, and the end of the simulation are determined from the linear regressions of the negative natural log of the absolute running average efficiency and the natural log of *unnormalized* iterations. The results are presented in Section 5.3.1.2.
- Scheme C) The values of n to be used during the beginning, the middle, and the end of the simulation are determined from the linear regressions of the negative natural log of the absolute running average efficiency and the natural log of *normalized* iterations. Results presented in Section 5.3.1.2.
- Scheme D) $n \neq 1$ is used during the first half of the simulation. Then n = 1 is used during the second half of the simulation and the convergence criterion is reduced by half. The results are presented in Section 5.3.1.3.
 - 5.3.1.1 Scheme A: Random n

In this section two cases of Scheme A that randomly changes n are implemented. In Case (i) n is randomly modified six times at points before the minimization that allows all dihedral angles on the entire chain to change. In Case (ii) n is randomly modified at points before the minimization that allows only perturbed dihedral angles to change. Figure 5.83 presents a) the percent difference of average score improvement ($\% \Delta AveSI$), b) the percent difference of the lowest score ($\% \Delta E_{lowest}$), c) the percent difference of the average iteration ($\% \Delta AveIt$), and d) the percent difference of the efficiency ($\% \Delta e_n$) from the AXEL with Scheme A. Lower values mean better performance and negative values mean better performance than n = 1.



Figure 5.83: Scheme A: a) the percent difference of average score improvement $\%\Delta AveSI$, b) the percent difference of lowest score $\%\Delta E_{lowest}$, c) the percent difference of average iteration $\%\Delta AveIt$, and d) the percent difference of the efficiency $\%\Delta e_n$. Case (i) has *n* randomly modified at points before the minimization that allows all dihedral angles on the entire chain to change and Case (ii) has *n* randomly modified at points before the minimization that allows only perturbed dihedral angles to change. Lower values mean better performance and negative values mean better performance than n = 1.

Results (Figure 5.83 a) show that there is no relationship between the rate of the varying n and the improvement or worsening of the values of $\%\Delta AveSI$ compared those previously seen in Section 5.2. The values of $\%\Delta E_{lowest}$ (Figure 5.83 b) are also within the same range as those previously seen but in protein 1d3z Scheme A is able to yield 10% lower $\%\Delta E_{lowest}$ by changing n only a few times. The values of $\%\Delta AveIt$ (Figure 5.83 c) are also within the same range as those previously seen but higher than the best case of those previously seen. The values of $\%\Delta e_n$ (Figure 5.83 d) are also within the same range as those previously seen but generally have $\%\Delta e_n$ values higher. Since the results yield about the same values of $\%\Delta AveSI$ and $\%\Delta E_{lowest}$ but lower values of $\%\Delta AveIt$ and $\%\Delta e_n$ compared to those previously seen, this suggests that changing n blindly can worsen speed and efficiency without any improvement in quality.

5.3.1.2 Schemes B and C: Running average efficiency

To investigate if efficiency can be further improved, Schemes B and C are developed. The values of n to be used in these schemes are the values that give the best running average efficiency, defined below, during the beginning, the middle, and the end of the simulation. Schemes B and C determine n from the linear regressions of the negative natural log of the absolute running average efficiency $(-\ln |e_{\text{running}}|$ defined below) versus the natural log of *unnormalized* iterations and versus the natural log of *normalized* iterations, respectively.

The efficiency e_n in Equation (5.7) is the overall efficiency computed at the end of the simulation. To understand the effect of n on the performance during the different periods of simulations, the running average efficiency $e_{running}$ is evaluated.

$$(e_{\text{running}})_{j,i} = \frac{(\min\left([E_{1,i}\dots E_{j,i}]\right) - E_{0,i})}{m_{j,i}/1000}; \quad i = 1\dots 50.$$
(5.9)

where $(e_{\text{running}})_{j,i}$ is the running average efficiency at the j^{th} configuration of the i^{th} decoy, $E_{1,i}$ is the score of the first configuration (input) of the i^{th} decoy, $E_{j,i}$ is the



Figure 5.84: The running average efficiency, $e_{running}$, versus iterations for protein 1d3z with various values of n. Lower values mean better running average efficiency.

score of the j^{th} configuration of the i^{th} decoy, and $m_{j,i}$ is the number of iterations of the j^{th} configuration of the i^{th} decoy. Then the average set of running average efficiency e_{running} for all 50 decoys is found. Since the set of running average efficiency for each decoy is not available for every iteration and it corresponds to a different set of iterations for different decoy, cubic spline data interpolation is used to obtain the running average efficiency from the same iteration set so that an average can be calculated.

Illustrating a relationship between e_{running} and iteration for various values of n of a representative example, 1d3z protein, Figure 5.84 shows that all n cases give almost identical logarithmic relationships. When these relationships are plotted in log scales, they become linear as shown in Figure 5.85 which illustrates that n that yields the best values of e_{running} has its graph lower than the others. Identifying the best values of n at the beginning, the middle, or the end of the simulation directly from these graph can be difficult because they are not perfect lines. Therefore, the



Figure 5.85: The running average efficiency, $e_{running}$, (log scale) versus iterations (log scale) with various values of n for protein 1d3z. Lower values mean better running average efficiency.

best values of n are determined from the linear regressions of these graphs. Two sets of the regressions which look similar to Figure 5.85 give n values for Schemes B and C.

Scheme B determines the values of n that give the best e_{running} from the linear regressions between $-\ln |e_{\text{running}}|$ and the natural log of iterations during the beginning, the middle, and the end of simulations for protein 1ubq, 1d3z, dom6, hetr, and mdmi. Scheme C determines values of n that give the best e_{running} from the linear regressions between $-\ln |e_{\text{running}}|$ and the natural log of normalized iterations. With unnormalized iterations in Scheme B the beginning, the middle, and the end periods of each simulation are defined by certain numbers of iterations, but with the normalized iterations in Scheme C the periods are defined by the length of the simulation itself. A chosen value of n yields the lowest regression line during each period which means the best performance.

Period	Scheme	1ubq	1d3z	dom6	hetr	mdvi
Beginning	B C	$2 \\ 2^{-9}$	$1 2^{-9}$	$2 \\ 2^{-9}$	$2 \\ 2^{-5}$	$2 \\ 2^{-9}$
Middle	B C	$2 \\ 2^{-9}$	$2 \\ 2^{-7}$	$1 2^{-9}$	$2 \\ 2^{-9}$	$2 \\ 2^{-9}$
End	B C			$1 2^{-9}$	2^{-3} 2^{-9}	2^{-5} 2^{-9}

Table 5.9: The values of n for Schemes B and C which give the best efficiency during the beginning, the middle, and the end of simulation periods for each protein from unnormalized and normalized iteration plots.

The values of n of Schemes B and C that give the best e_{running} during the beginning, the middle, and the end of the simulations for each protein are displayed in Table 5.9. n = 2 is often chosen for Scheme B and $n = 2^{-9}$ is often chosen for Scheme C. These values of n are used during their corresponding periods in the BFGS-XEL (*Hessian only*). Figure 5.86 gives a) the percent difference of average score improvement ($\% \Delta AveSI$), b) the percent difference of the lowest score ($\% \Delta E_{lowest}$), c) the percent difference of the average iteration ($\% \Delta AveIt$), d) and the percent difference of the efficiency ($\% \Delta e_n$) from the AXEL with Schemes B and C, respectively.

Figures 5.86 a) and b) show that Scheme B yields similar performance as n = 1 cases in terms of quality but Scheme C yields lower quality. Figure 5.86 a) shows that AveSI's from Scheme B are comparable to those from n = 1 cases (within 0 to -3%), but those from Scheme C are 4 to 15% higher. This means obtaining the values of n from the normalized iteration data does not improve the performance of the algorithm regarding to the AveSI values. Figure 5.86 b) shows that most lowest scores from Scheme B are comparable to those from n = 1 cases (within 0 to 5%) but for protein hetr Scheme B gives 15% lower lowest score. Scheme C gives a higher lowest score than Scheme B in small proteins but up to a 12% lower lowest score for larger proteins.



Figure 5.86: Schemes B and C: a) the percent difference of average score improvement $\% \Delta AveSI$, b) the percent difference of lowest score $\% \Delta E_{lowest}$, c) the percent difference of average iteration $\% \Delta AveIt$, and d) the percent difference of the efficiency $\% \Delta e_n$. Lower values mean better performance and negative values mean better performance than n = 1.

Figures 5.86 c) and d) demonstrate that Scheme B yields the same performance as n = 1 case in terms of speed and efficiency but Scheme C yields better speed and efficiency. In Scheme C, obtaining n's from the normalized iteration data yields the best speed and efficiency because most obtained n's are equal to 2^{-9} , which is the case that yields the best speed and efficiency in BFGS-XEL. In fact, three proteins, 1ubq, dom6, and mdmi have $n = 2^{-9}$ during the entire simulations. In Scheme B, obtaining n's from the unnormalized iteration data does not yield as good speed and efficiency because most obtained n's are 2, which yields about the same speed and efficiency as the n = 1 case.

The results from Schemes B and C show that determining the best n for each simulation period from the running average efficiency may not be suitable for the AXEL. Since the running average efficiency are cumulative, the n that gives the best result at the current simulation period is most likely the same as the best n at the last period. Having repeated n's yields almost the same results as the BFGS-XEL, in which n is constant.

5.3.1.3 Scheme D: Stricter convergence criterion

Scheme D changes both n and the convergence criterion. During the first half of the simulation $n \neq 1$ is used, and during the second half of the simulation n changes to 1 and the convergence criterion is reduced by half. Two values of $n \neq 1$ are used: $n = 2^{-9}$ and n = 10. Since the XEL algorithm yields higher efficiency than the unmodified case, this scheme is investigated to determine if quality can be improved by reducing speed with a stricter convergence criterion.

Figure 5.87 presents a) the percent difference of the average score improvement $(\%\Delta AveSI)$, b) the percent difference of the lowest score $(\%\Delta E_{lowest})$, c) the percent difference of the average iteration $(\%\Delta AveIt)$, and d) the percent difference of the efficiency $(\%\Delta e_n)$, respectively. Lower values mean better performance and negative



Figure 5.87: Scheme D: a) the percent difference of average score improvement $\%\Delta AveSI$, b) the percent difference of lowest score $\%\Delta E_{lowest}$, c) the percent difference of average iteration $\%\Delta AveIt$, and d) the percent difference of the efficiency $\%\Delta e_n$ from two values of starting n. Lower values mean better performance and negative values mean better performance than n = 1.

values mean better performance than n = 1.

Figures 5.87 a) and b) indicate that Scheme D performs as well as the n = 1 case in terms of quality. Figure 5.87 a) shows that almost all cases of Scheme D gives AveSI within $\pm 5\%$ of $AveSI_{n=1}$ which is a significant improvement for the $n = 2^{-9}$ case compared to the constant $n = 2^{-9}$ case. Most cases with the starting n = 10 give lower AveSI than cases with $n = 2^{-9}$, which yield up to 6% reduction in $\% \Delta AveSI$. For the $\% \Delta E_{lowest}$, Figure 5.87 b) shows that while only a few cases of Scheme D give slightly over 5% higher E_{lowest} than those of n = 1, most cases have E_{lowest} within $\pm 5\%$ of $(E_{lowest})_{n=1}$ and other two cases have E_{lowest} 12.5% and 26% lower than those of n = 1.

Figures 5.87 c) and d) indicate that Scheme D performs better than the n = 1 case in terms of speed and efficiency. Figure 5.87 c) shows that most cases of Scheme D have lower numbers of iterations than n = 1. All cases with $n = 2^{-9}$ have lower $\% \Delta AveIt$ than cases with n = 10, but these numbers are higher than those of BFGS-XEL cases (Figure 5.67). This is due to the reduction of the convergence criterion during the second half of the simulation, which increases the number of iterations. Similar to $\% \Delta AveIt$, all $\% \Delta e_n$ for $n = 2^{-9}$ cases are lower than those of n = 10 cases, but these numbers are higher than those of of BFGS-XEL cases (Figure 5.75). The results of Scheme D demonstrate that the improvement of quality can be achieved by sacrificing speed and that the value of n at the beginning of the simulation have significant effect on the performance.

5.3.2 Local schemes

In local AXEL schemes, n changes depending on the information derived from the energy landscape which conditionally happens during the minimization. Since n varies during the minimization, the varying-n XEL algorithms are used with the local AXEL schemes. This section presents the results from simulations in which the varying-n BFGS-XEL is implemented. Several local AXEL schemes are investigated but only the following representative examples are presented in this section:

- Scheme E) n changes such that the error residue is bounded. The results are presented in Section 5.3.2.1.
- Scheme F) n changes when one of the criteria is met. Figure 5.89 gives the pseudo code for Scheme F). The results are presented in Section 5.3.2.2.

Note that a scheme that randomly changes n is not used as a control case for the local AXEL schemes because the changes can be too large and cause the minimization
to diverge.

5.3.2.1 Scheme E: Bounded error residue

Following Guideline *iii*. in Table 4.1 several schemes that change n such that the scalar multiplier λ (Equation 4.21) is bounded have been investigated, but they yield unsuccessful results and often diverge. Scheme E follows a slightly different approach by changing n such that the magnitude of the term $\lambda(n-1)E^{-1}$ is less than 1. As this term multiplies the first-order term of the XEL error residue ρ^* (Equation 4.27), a bounded $\lambda(n-1)E^{-1}$ implies a bounded $|\rho^*|$. Although ρ^* is not derived for the varying-n BFGS-XEL, by substituting \hat{H} for K^{-1} in λ the modified $\lambda(n-1)E^{-1}$ are used. This is because the behaviors of the modified $\lambda(n-1)E^{-1}$ are assumed to be similar to the $\lambda(n-1)E^{-1}$ of the Newton-XEL.

Figure 5.88 presents a) the percent difference of the average score improvement $(\%\Delta AveSI)$, b) the percent difference of the lowest score $(\%\Delta E_{lowest})$, c) the percent difference of the average iteration $(\%\Delta AveIt)$, and d) the percent difference of the efficiency $(\%\Delta e_n)$. In Case (i) there is no limit for the value of n and in Case (ii) only the value of n between 2^{-9} and 10 is allowed. Lower values mean better performance and negative values mean better performance than n = 1.

Figure 5.88 a) and b) shows that Scheme E slightly improves the quality and can yield higher quality in some proteins. As the values of $\%\Delta AveSI$ of hetr can be lower than -7%, the values of $\%\Delta E_{lowest}$ of hetr and 1d3z are -15% and 25%. On average the values of $\%\Delta AveSI$ and $\%\Delta E_{lowest}$ are slightly lower than those yielded by a constant n.

Figure 5.88 c) and d) shows that Scheme E only slightly worsens speed and efficiency compared to n = 1 as almost all values of $\% \Delta AveIt$ and $\% \Delta e_n$ are within the $\pm 5\%$ from zero. While the values of $\% \Delta AveIt$ (Figure 5.88 c) are about the same as those seen in the worst case of the constant n (Figure 5.66), the values of $\% \Delta e_n$ are



Figure 5.88: Schemes E: a) the percent difference of average score improvement $\%\Delta AveSI$, b) the percent difference of lowest score $\%\Delta E_{lowest}$, c) the percent difference of average iteration $\%\Delta AveIt$, and d) the percent difference of the efficiency $\%\Delta e_n$ from Schemes E cases. Case (i) has no limit for n and Case (ii) only allows the values of n to be between 2^{-9} and 10. Lower values mean better performance and negative values mean better performance than n = 1.

slightly lower than those seen in the worst case (Figure 5.74) because of lower values of $\% \Delta AveIt$.

The results show that on average Scheme E yields slightly higher quality for all proteins and worsens speed and efficiency no more than the worst case of the constant n cases.

5.3.2.2 Scheme F: Path and landscape characteristic criteria

Pseudo-code: Varying-*n* BFGS-XEL with Scheme F and a line search

Initialize ϵ , n_1 , θ_0 , θ_1 , \hat{H}_0^* , and μ_i where $i = 1, \ldots, 6$ **Calculate** E_0 and τ_0 for $k = 1, \dots, k_{max}$ do Calculate E_k , and τ_k if $\left|\frac{E_k - E_{k-1}}{E_{k-1}}\right| < \epsilon$ then break // Convergence criterion met end if **Determine** n_{k+1} and \hat{h}_{nk}^* if $k > \mu_1$ then // Change n only after μ_1^{th} iteration if $\frac{\tau_k \cdot \tau_{k-1}}{|\tau_k| |\tau_{k-1}|} > \mu_2$ then // An angle between the consecutive gradient is too large **Increase** n // Take a smaller step else if $\frac{|E_k - E_{k-1}|}{|E_{k-1}|} < \mu_3 \& |\tau_k| > \mu_4$ then // Change in energy is small but the gradient is large **Decrease** n // Take a larger step else if $\frac{h_{\theta k-1}^* \cdot \tau_k}{\left|\hat{h}_{\theta k-1}^*\right| |\tau_k|} < \mu_5$ then // The step is almost perpendicular to the gradient **Increase** *n* else if $\frac{k}{\mu_6} = 0$ then **Decrease** n // Increase step size every μ_6^{th} iteration end if end if Calculate \hat{H}_k^* $\tau_k^* = \beta_k^k \tau_k$ $\hat{h}_{\theta k}^{*} = \hat{H}_{k}^{*} \left(1 + \left(\frac{1}{n_{\nu}} + \ln |E_{k}| \right) \hat{h}_{n k}^{*} \right) \tau_{k}^{*} / / \text{ Varying-} n \text{ BFGS-XEL step}$ $\alpha_k^* = \texttt{line_search}(\hat{h}_{\theta k}^*) / / \text{Determine the gain with a line search}$ $\theta_{k+1} = \theta_k + \alpha_k^* \hat{h}_{\theta k}^* / / (k+1)^{\text{th}}$ solution end for

Figure 5.89: A pseudo-code for Scheme F with varying n BFGS-XEL

Scheme F changes n in a systematic manner which is based on the observations from Section 5.1. Following Guideline iv. n is increased when the estimated modified Hessian inverse \hat{H}^* is inaccurate. Therefore, n > 1 is used at the beginning of the simulation and n is changed only after the certain number of iterations. Three criteria are used to determine how and when to change n. (1) If the energy landscape changes too much, which occurs when the size of an angle between the current and the previous gradient exceeds a certain criterion, the step size is reduced by increasing n. (2) If the step size becomes too small, which occurs when a change in score is smaller than a criterion but a magnitude of the gradient is larger than a criterion parameter, the step size is increased by decreasing n. (3) When the \hat{H} is not accurate, the step size is decreased by increasing n. \hat{H} is considered inaccurate if the the step is almost perpendicular to the current gradient, therefore n is increased when an angle between the gradient and the step exceeds a criterion parameter. When none of the criteria are met, a larger step size is preferred for better speed, so n is decreased every certain number of iterations. Figure 5.89 gives a pseudo code for Scheme F). Two sets of criterion parameters i) and ii) shown in Table 5.10 are implemented and their results are presented in Figure 5.90. These parameter sets are those that yields the best results in preliminary tests implemented without the probabilistic search. Most parameters are kept the same as they do not significantly affect the results. Although Set ii) has some stricter criterion parameters, their results are similar.

 Table 5.10: Two sets of criterion parameters for Scheme F.

Set	μ_1	μ_2	μ_3	μ_4	μ_5	μ_6
i)	3	$0.707~(45^{\circ})$	0.1 (10%)	0.5	$0.259~(75^{\circ})$	5
ii)	3	$0.866~(30^{\circ})$	0.1~(10%)	0.5	$0.5~(60^{\circ})$	5

Figure 5.90 presents a) the percent difference of the average score improvement $(\% \Delta AveSI)$, b) the percent difference of the lowest score $(\% \Delta E_{lowest})$, c) the percent



Figure 5.90: Scheme F: a) the percent difference of average score improvement $\%\Delta AveSI$, b) the percent difference of lowest score $\%\Delta E_{lowest}$, c) the percent difference of average iteration $\%\Delta AveIt$, and d) the percent difference of the efficiency $\%\Delta e_n$ from two sets of criterion paremeters. Lower values mean better performance and negative values mean better performance than n = 1.

difference of the average iteration ($\% \Delta AveIt$), and d) the percent difference of the efficiency ($\% \Delta e_n$). Lower values mean better performance and negative values mean better performance than n = 1.

Figure 5.90 a) and b) indicates that on average Scheme F yields lower quality than the n = 1 case. As proteins 1ubq and 1d3z yield more than 5% higher $AveSI_{n=1}$, almost all proteins yield 5% higher $\%\Delta E_{lowest}$. The achieved quality is similar to that seen in the constant n < 1 case (Figures 5.50 and 5.58).

Figure 5.90 c) and d) indicates that Scheme F yields significantly better speed

and efficiency than the n = 1 case. Both $\% \Delta AveIt$ and $\% \Delta e_n$ are dependent on the size of protein chains. That is they decrease as the size of protein chains increases. The achieved speed and efficiency are similar to that seen in the constant n < 1 case (Figures 5.66 and 5.74).

The results of Scheme F demonstrate that systematically changing n yields about the same results as the constant n < 1 case and two sets of the criterion parameters yields almost identical results. This is most likely because n is decreased more often than increased, stricter parameters μ_3 and μ_4 and larger μ_6 may yield different results. These parameters do not seem to affect the results in the preliminary test because the simulated energy landscapes most likely do not represent the score function of the protein molecule.

5.3.3 Conclusion of BFGS with AXEL

This section presents results of the AXEL applying to the BFGS algorithm. Several global and local AXEL schemes are implemented with the BFGS-XEL (*Hessian only*) or the varying-*n* BFGS-XEL and results are presented.

While the results from the global and local AXEL schemes do not give any insight into the global and the local effects of the AXEL on the prediction simulation, they show that the AXEL can improve quality or speed but rarely both at the same time. With Scheme A, results show that randomly changing n can worsen speed without improving quality. With Schemes B and C results show that choosing n for different period of the simulation from the running average efficiency is not suitable for the AXEL. Results from Scheme D show that higher quality can be achieved by sacrificing speed and that the value of n at the beginning of the simulation has a significant effect on the performance. The results show that Scheme E yields the best results in terms of quality than the other schemes as a lower AveSI than that of the constant n case can be achieved in all proteins while the AveIt is not more than 5% of the n = 1 case. Lastly, the results from Scheme F demonstrate that systematically changing n yields about the same results as the constant n < 1 case.

While most schemes yield the performance that can be achieved by constant n, Scheme E yields at least 4% lower AveSI for all proteins which are not achieved by any constant n. Although only the varying-n BFGS-XEL is implemented with Scheme E, the better improvement seen here is expected to be found in other algorithms because of the consistency seen in results from cases with the constant n.

5.4 Conclusion of Simulation Results and Discussions

The effects of the XEL on the Newton's method, the QNA, and the BFGS method are investigated with two implementations. First, the investigated algorithms are implemented without a probabilistic search on a simulated two-dimensional energy landscape and second they are implemented with a probabilistic search on a multidimensional energy landscape. The AXEL is implemented with a probabilistic search on a multi-dimensional energy landscape.

Results from both implementations show that better quality or speed can be achieved by the XEL, but not both at the same time. Both results show that the quality can be achieved with n > 1 which is in agreement with Guideline *iv*. presented in Table 4.1. In terms of speed the results from the two implementations do not agree since n > 1 yields better speed in the first implementation but n < 1 yields better speed in the second implementation. In terms of the global search performance or ability to locate the global solution, on average the XEL yields higher lowest score than that of n = 1 case which implies worse global search performance.

The results from the XEL implementation with a probabilistic search show that a trade-off between the quality and speed must be considered when the XEL is implemented. Tables 5.5 and 5.6 on pages 213–214 are provided to be used as look-up tables. The results also show that XEL has no effect on the similarity of the resulting configurations to the native conformation.

The results from the AXEL shows that quality or speed can be improved but rarely both at the same time. Scheme E yields at least 4% lower AveSI for all proteins compared to the XEL with constant n; however, the rest of AXEL yields roughly the same results. Results lead to three recommendations for better quality in terms of average score improvement, better speed in terms of the average iteration, and better efficiency.

To improve speed by 15% to 47%, the XEL with n within $2^{-9}-2^{-5}$ should be used. This range is suggested because of the consistent improvement of speed in all algorithms. Table 5.5 shows that for the QNA-XEL and the QNA-XEL (*Hessian* only) speed can be 25% to 42% and 32% to 47% better than the unmodified case and for the BFGS-XEL and the BFGS-XEL (*Hessian only*) speed can be 18% to 41% and 15% to 40% better. The XEL with n within $2^{-9}-2^{-5}$ is suggested for the first round of the refinement to evaluate a large set of decoys so that a better subset can be obtained for further refinement.

To improve quality by 4% to 7%, the AXEL with Scheme E should be used. Although only the varying-n BFGS-XEL were implemented, Scheme E is expected to yield similar improvement with the varying-n QNA-XEL. The varying-n XEL with Scheme E is suggested for the further round of the refinement for improved quality.

To improve efficiency by 13% to 75%, the XEL with n within $2^{-9}-2^{-5}$ should be used. Figures 5.70 and 5.71 show that the efficiency in the QNA-XEL and the QNA-XEL (*Hessian only*) cases can be 25% to 51% and 32% to 75% better than the unmodified case. Figures 5.74 and 5.75 show that the efficiency in the BFGS-XEL and the BFGS-XEL (*Hessian only*) cases can be 13% to 49% and 13% to 52% better than the unmodified case. The same range of n is recommended for improving speed.

CHAPTER VI

CONCLUDING REMARKS

This thesis develops and demonstrates the exponential energy landscaping (XEL) and the adaptive exponential energy landscaping (AXEL) for an analytical optimization algorithm in protein conformation prediction. The effects of XEL are investigated on Newton's method, the quasi-Newton algorithm (QNA), and the Broyden-Fletcher-Goldfarb-Shanno (BFGS) algorithm. The AXEL is a combination of a scheme for adaptively modifying energy landscape and the XEL algorithms.

XEL shares traits with hypersurface deformation [4, 5, 50] and the energy landscape flattening [87] in terms of modifying the energy landscape. However, the XEL has the following attributes that set it apart from the aforementioned techniques:

- XEL is applicable to any energy function.
- XEL is only applicable for an analytical optimization algorithm.
- XEL does not require remapping to the original surface.

This work addresses the bottleneck in protein conformation prediction which is the analytical optimization algorithm and provides a solution for improving speed and efficiency. The work also provides a potential solution for improving quality.

6.1 Contributions

This work makes the following contributions that are different from previous work:

- 1. A theoretical basis for applying a method of the XEL to analytical optimization algorithms.
 - The XEL that does not change the minima of the landscape and is applicable to any energy function is developed.

- The Newton's method, the QNA, and the BFGS algorithm with XEL are derived.
- Two methods of adaptive energy landscaping are derived: adaptive-n XEL algorithms and varying-n XEL algorithms with AXEL schemes.

2. Analyses for theoretical validation and characterization of the XEL.

- The positive definite property of the Newton's method with XEL (Newton-XEL) is studied and conditions that give a positive definite Hessian matrix of the Newton-XEL are found.
- The XEL is found to affect only the magnitude of the Newton step but not the direction. The scalar multiplier is found to be eigenvalue of a matrix and its properties are used to derive the error residue of the Newton-XEL.
- The error residue of the Newton-XEL is found to be on the same order of that of Newton's method which implies that they have similar convergence property.
- The scalar multiplier λ is also found to be inversely dependent of the exponent n.
- The exponent n is found to weight the energy landscape information in relation to the unweighted Hessian matrix.
- Guidelines for changing the exponent n in AXEL are developed.

3. Simulation validation of the application and the effect of the XEL.

- XEL is found to be significantly affected by different line search algorithms.
- XEL is found to significantly improve efficiency in the analytical portion of the protein prediction.
- When XEL is implemented a trade-off between speed and quality must be considered.

- XEL is found to have little effect on the similarity of resulting configurations to the native conformation.
- AXEL is mostly found to yield similar performance as XEL.
- Recommendations for improving speed, quality, or efficiency are presented.

This work is complementary to [87]; both are applicable to any energy function and do not require remapping but [87] applies landscaping to the Monte Carlo method, a probabilistic optimization algorithm. Comparing this work to [87] can be done only by comparing the effects of landscaping on the optimization algorithm which was not studied in [87].

To put the above contributions into context, a brief summary is given here. Chapter 1 discusses that the bottleneck in protein conformation prediction is the analytical optimization algorithm and it should be addressed. Chapter 2 presents the literature review of several aspects in the protein conformation prediction problem. This leads to the objective of this research which is to develop a method of energy landscaping that enhances the efficiency of the analytical algorithms for the protein prediction problem. Chapter 3 presents the derivation and the application to the protein folding problem of the investigated algorithms which are Newton's method, the QNA, and the BFGS algorithm.

Chapter 4 presents the XEL, the AXEL, and their derivation on investigated optimization algorithms. The XEL modifies the energy landscape such that the minima stay at the same locations. The Newton's method, the QNA, and the BFGS algorithm are applied to the modified energy function and yield Newton-XEL, QNA-XEL, and BFGS-XEL algorithms. The Newton-XEL direction which determines the direction of the optimization step is proved to be parallel to the Newton direction. The convergence property of the Newton-XEL method is found to be similar to that of Newton's method as their error residues are on the same order. Several properties of the Newton-XEL yield guidelines for changing the exponent n of the XEL to achieve certain operational characteristics such as improved stability or larger optimization step.

In the AXEL, the investigated algorithms are applied to the adaptively modified energy function, which is also of a function of n, and they yield adaptive-n XEL and varying-n XEL algorithms. Adaptive-n XEL algorithms simultaneously calculate the changes in n and the dihedral angles resulting in difficulties in reaching convergence. Varying-n XEL algorithms only calculate the change in dihedral angles based on the changes of n determined by the AXEL schemes. The AXEL schemes are categorized by when n changes as either global or local schemes.

Chapter 5 first presents and discusses the effects of the XEL on the Newton's method, the QNA, and the BFGS method on two implementations, with and without a probabilistic search. Then the AXEL implemented on the BFGS with a probabilistic search is presented and discussed. The results are evaluated in terms of quality, speed, and efficiency. While quality in terms of similarity are not affected by XEL, XEL can achieve better quality in terms of average score improvement or better speed in terms of average iteration. A better quality compared to unmodified case can be achieved with n > 1 and a better speed can be achieved with n < 1. However, the AXEL results show that both quality or speed can rarely be improved at the same time. This indicates that the trade-off between quality and speed must be considered when the XEL is implemented. The results lead to the following recommendations: (Note that quality is in terms of average score improvement and speed is in terms of the average iterations.)

- To improve *speed* by 15% to 47%, the XEL with n within $2^{-9}-2^{-5}$ should be used. Quality may be worsened by 4% to 15% on that range of n.
- To improve quality by 4% to 7%, the AXEL with Scheme E should be used. Speed may be worsened up to 6%.
- To improve efficiency by 13% to 75%, the XEL with n within $2^{-9}-2^{-5}$ should be

used.

Since in protein folding the conformation space is very large, sampling must be extensive to increase the possibility of covering the area that contains the global minimum. As a result, the process usually starts with several thousand decoys even for a small molecule. The first few rounds of the refinement are used to differentiate the good from the bad decoys. It is usually low resolution and does not required high quality. After the first or the second round of refinement, the best decoys are picked for the further round of refinement that usually requires higher quality. The XEL with n within $2^{-9}-2^{-5}$ is suggested for the first few rounds of the refinement and the AXEL with Scheme E is suggested for the higher refinement round.

While several investigated algorithms are derived and implemented, this work is not intended to compare the performance across different algorithms, but to compare the XEL algorithm to the unmodified case.

6.2 Future Work

Since the XEL is much more successful in increasing speed than quality, future work should be intermediately focused on further improving the quality and later focused on improving both simultaneously by developing an AXEL scheme. One approach can be the improvement or the variation of Scheme E. Another approach can be an improved global AXEL scheme as the current schemes yields the same performance as the constant n cases. Another study can combine the XEL to landscaping on the probabilistic optimization algorithm, such as energy landscape flattening, while another study can revisit the adaptive-n XEL approach by integrating a robust weighting algorithm.

With more computational resources at hand, a study of the global effect of the XEL can be done by implementing the XEL to a full-scale protein prediction process

which includes several rounds of full-atom refinement. Implementing XEL on the fullatom refinement will allow the effect of different energy functions to be studied. Since the improvement of the speed tends to be reduced for a larger protein, more proteins with higher residues will allow the effect of the protein size on the performance of XEL to be studied. Also the effect of different secondary structural contents in proteins on the performance of XEL can be studied. For example, proteins with large and small numbers of α -helix contents can be compared with those with large and small numbers of β -strand contents and with each other. While this work gives an insight about the similarity of the resulting configurations, future work should evaluate the similarity from the results of many more decoys and proteins. This will allow a conclusion to be drawn with more confidence because of the probabilistic nature of the process.

This work investigates the effects of the XEL on a few Newtonian algorithms. A study can be done to determine if other algorithms such as the Davidon-Fletcher-Powell (DFP) or the Broyden family can be improved by the XEL. Another study can compare the performance across these algorithms including the QNA and BFGS algorithm investigated here. Another study can investigate gradient-based optimization algorithms, such as the conjugate gradient method, that do not need Hessian calculation or estimation.

The energy landscape can be lowered or raised by a constant such that all energy values become negative or positive. A study can investigate the effects of lowering and raising the landscape on the performance of XEL in different algorithms.

A study of the application of the XEL on probabilistic methods can be investigated. An application of the XEL to the Monte Carlo method can possibly be done by modifying the energy term in the Boltzmann factor. With the exponent of n < 1the Boltzmann factor becomes smaller which gives the same effect as flattening the energy landscape.

APPENDIX A

SUPPLEMENTARY DOCUMENTS

Appendix A provides supplementary data to Chapter 5, Section A.1.

A.1 QNA-XEL and BFGS-XEL Combining with Probabilistic Search

A.1.1 QNA-XEL and QNA-XEL (Hessian only)

Figures A.1 and A.2 present the percentage of decoys found by the QNA-XEL with lower, equal, or higher score or number of iterations than those of n = 1 for protein 1ubq, 1d3z, dom6, hetr, and mdmi. Figures A.3 and A.4 present results found by the QNA-XEL (*Hessian only*).

A.1.2 BFGS-XEL and BFGS-XEL (*Hessian only*)

Figures A.5 and A.6 present the percentage of decoys found by BFGS-XEL with lower, equal, or higher score or number of iterations compared to n = 1 for protein 1ubq, 1d3z, dom6, hetr, and mdmi. Figures A.7 and A.8 present results found by BFGS-XEL (*Hessian only*).



Figure A.1: QNA-XEL: The percentage of decoys with less, equal, higher scores or number of iterations compared to n =for proteins 1ubq (a and d), 1d3z (b and e), and dom6 (c and f).



Figure A.2: QNA-XEL: The percentage of decoys with less, equal, higher scores or number of iterations compared to n = 1for proteins hetr (g and i) and mdmi (h and j).



QNA-XEL (Hessian only): The percentage of decoys with less, equal, higher scores or number of iterations compared to n = 1 for proteins 1ubq (a and d), 1d3z (b and e), and dom6 (c and f). Figure A.3:



Figure A.4: QNA-XEL (Hessian only): The percentage of decoys with less, equal, higher scores or number of iterations compared to n = 1 for proteins hetr (g and i) and mdmi (h and j).



Figure A.5: BFGS-XEL: The percentage of decoys with less, equal, higher scores or number of iterations compared to n = 1for proteins 1ubq (a and d), 1d3z (b and e), and dom6 (c and f).







Figure A.7: BFGS-XEL (Hessian only): The percentage of decoys with less, equal, higher scores or number of iterations compared to n = 1 for proteins 1ubq (a and d), 1d3z (b and e), and dom6 (c and f).



Figure A.8: BFGS-XEL (Hessian only): The percentage of decoys with less, equal, higher scores or number of iterations compared to n = 1 for proteins hetr (g and i) and mdmi (h and j).

REFERENCES

- "Residue count histogram." RCSB Protein Data Bank http://www.pdb.org, November 2009.
- [2] ALEXANDROV, V., LEHNERT, U., ECHOLS, N., MILBURN, D., ENGELMAN, D., and GERSTEIN, M., "Normal modes for predicting protein motions: A comprehensive database assessment and associated web tool," *Protein Science*, vol. 14, pp. 633–643, 2005.
- [3] ALVARADO DÍAZ, C. A., Computationally Efficient Kinematic Analysis Methods in Biological Structures. Doctor of philosophy dissertation, Mechanical Engineering, University of Connecticut, Storrs, CT, 2005.
- [4] AZMI, A. M., BYRD, R. H., ESKOW, E., SCHNABEL, R. B., CRIVELLI, S., PHILIP, T. M., and HEAD-GORDON, T., "Predicting protein tertiary structure using a global optimization algorithm with smoothing," in *Optimization in Computational Chemistry and Molecular Biology: Local and Global Approaches* (FLOUDAS, C. A. and PARDALOS, P. M., eds.), pp. 1–18, Dordrecht/Boston/ London: Kluwer Academic, 2000.
- [5] AZMI, A. M., BYRD, R. H., ESKOW, E., and SCHNABEL, R. B., "A new smoothing-based global optimization algorithm for protein conformation problems," in *Global Optimization: Scientific and Engineering Case Studies* (PINTÉR, J. D., ed.), vol. 85 of *Nonconvex Optimization and Its Applications*, pp. 73–102, New York: Springer, 2006.
- [6] BELEGUNDU, A. D. and CHANDRUPATLA, T. R., *Optimization Concepts and Applications in Engineering*. Upper Saddle River, N.J.: Prentice Hall, 1999.
- [7] BERENDSEN, H. J. C., "Molecular dynamics simulations: The limits and beyond," in 2nd International Symposium on Algorithms for Macro-Molecular Modelling (DEUFLHARD, P., HERMANS, J., LEIMKUHLER, B., MARK, A. E., RE-ICH, S., and SKEEL, R. D., eds.), (Berlin, Germany), pp. 3–36, Springer, 1997.
- [8] BERTSCH, R. A., VAIDEHI, N., CHAN, S. I., and GODDARD, W. A., "Kinetic steps for α-helix formation," *Proteins: Structure, Function, and Genetics*, vol. 33, no. 3, pp. 343–357, 1998.
- [9] BOHNENKAMP, P., KAZEROUNIAN, K., and ILIES, H. T., "Structural prediction of peptide based nano systems via progressive landscape evolution," in 12th IFToMM World Congress, (Besançon, France), 2007.

- [10] BOWERS, P. M., STRAUSS, C. E. M., and BAKER, D., "De novo protein structure determination using sparse NMR data.," *Journal of Biomolecular NMR*, vol. 18, no. 4, pp. 311–318, 2000.
- [11] BRADLEY, P., MISURA, K. M. S., and BAKER, D., "Toward high-resolution de novo structure prediction for small proteins," *Science*, vol. 309, pp. 1868–1871, 2005.
- [12] BROOKS, B. R., BRUCCOLERI, R. E., OLAFSON, B. D., STATES, D. J., SWAMINATHAN, S., and KARPLUS, M., "Charmm - a program for macromolecular energy, minimization, and dynamics calculations," *Journal of Computational Chemistry*, vol. 4, no. 2, pp. 187–217, 1983.
- [13] BURKE, M. G. and YALIRAKI, S. N., "Exploring model energy and geometry surfaces using sum of squares decompositions," *Journal of Chemical Theory and Computation*, vol. 2, no. 3, pp. 575–587, 2006.
- [14] CAFLISCH, A. and KARPLUS, M., "Molecular dynamics studies of protein and peptide folding and unfolding," in *The Protein Folding Problem and Tertiary Structure Prediction* (MERZ, K. M. and GRAND, S. M. L., eds.), Boston, MA: Birkhäuser, 1994.
- [15] CANUTESCU, A. A. and DUNBRACK, R. L., "Cyclic coordinate descent: A robotics algorithm for protein loop closure," *Protein Science*, vol. 12, pp. 963– 972, 2003.
- [16] CORNELL, W., CIEPLAK, P., BAYLY, C., GOULD, I., MERZ, K., FERGUSON, D., SPELLMEYER, D., FOX, T., CALDWELL, J., and KOLLMAN, P., "A second generation force field for the simulation of proteins, nucleic acids, and organic molecules," *Journal of the American Chemical Society*, vol. 117, pp. 5179–5197, 1995.
- [17] DENNIS, J. E. and SCHNABEL., R. B., Numerical Methods for Unconstrained Optimization and Nonlinear Equations. Classics in applied mathematics, Philadelphia: Society for Industrial and Applied Mathematics, 1996.
- [18] DEUFLHARD, P., Newton Methods for Nonlinear Problems: Affine Invariance and Adaptive Algorithms. Springer Series in Computational Mathematics, Berlin, Germany: Springer, 2006.
- [19] DJURDJEVIC, D. P. and BIGGS, M. J., "Ab initio protein fold prediction using evolutionary algorithms: Influence of design and control parameters on performance," *Journal of Computational Chemistry*, vol. 27, no. 11, pp. 1177–1195, 2006.
- [20] DUNFIELD, L. G., BURGESS, A. W., and SCHERAGA, H. A., "Energy parameters in polypeptides. 8. Empirical potential energy algorithm for the conformational analysis of large molecules," *The Journal of Physical Chemistry*, vol. 82, no. 24, pp. 2609–2616, 1978.

- [21] FEATHERSTONE, R., Robot Dynamics Algorithms. The Kluwer International Series in Engineering and Computer Science; Robotics: Vision, Manipulation and Sensors, Boston/Dordrecht/Lancaster: Kluwer Academic Publishers, 1987.
- [22] FEATHERSTONE, R., "An empirical study of the joint space inertia matrix," International Journal Robotics Research, vol. 23, no. 9, pp. 859–871, 2004.
- [23] FEATHERSTONE, R., "Efficient factorization of the joint space inertia matrix for branched kinematic trees.," *International Journal Robotics Research*, vol. 24, no. 6, pp. 487–500, 2005.
- [24] FEATHERSTONE, R. and ORIN, D. E., "Robot dynamics: Equations and algorithms," in *IEEE International Conference Robotics & Automation*, (San Francisco, CA), pp. 826–834, 2000.
- [25] GUPTA, K., "Kinematic solutions of robots with continuous three-roll wrists using the zero reference position method," *IEEE*, 1987.
- [26] GUPTA, K. and CARLSON, G. J., "On certain aspects of the zero reference position method and its applications to an industrial manipulator," *IEEE*, 1986.
- [27] HANSMANN, U. H. E., "Parallel tempering algorithm for conformational studies of biological molecules," *Chemical Physics Letters*, vol. 281, pp. 140–150, 1997.
- [28] JACOB, M., SCHINDLER, T., BALBACH, J., and SCHMID, F. X., "Diffusion control in an elementary protein folding reaction," *Proceedings of The National Academy of Sciences of the United States of America*, vol. 94, pp. 5622–5627, 1997.
- [29] JACOBS, D. J., KUHN, L. A., and THORPE, M. F., "Flexible and rigid regions in proteins," in *Rigidity Theory and Applications* (THORPE, M. F. and DUXBURY, P., eds.), New York: Kluwer Academic/Plenum, 1999.
- [30] JACOBS, D. J., RADER, A., KUHN, L. A., and THORPE, M., "Protein flexibility predictions using graph theory," *Proteins: Structure, Function, and Genetics*, vol. 44, pp. 150–165, 2001.
- [31] JAIN, A., VAIDEHI, N., and RODRIGUEZ, G., "A fast recursive algorithm for molecular dynamics simulation," *Journal of Computational Physics*, vol. 106, pp. 258–268, 1993.
- [32] KABSCH, W., "A solution for the best rotation to relate two sets of vectors," Acta crystallographica. Section A, Crystal physics, diffraction, theoretical and general crystallography, vol. 32, pp. 922–923, 1976.
- [33] KABSCH, W., "A discussion of the solution for the best rotation to relate two sets of vectors," Acta crystallographica. Section A, Crystal physics, diffraction, theoretical and general crystallography, vol. 34, pp. 827–828, 1978.

- [34] KAZEROUNIAN, K., LATIF, K., and ALVARADO, C., "Protofold (Part II): A successive kineto-static compliance method for protein conformation prediction," in ASME 2004 Design Engineering Technical Conferences and Computers and Information in Engineering Conference, (Salt Lake City, Utah, USA), 2004.
- [35] KAZEROUNIAN, K., "From mechanisms and robotics to protein conformation and drug design," *Journal of Mechanical Design*, vol. 126, pp. 40–45, 2004.
- [36] KAZEROUNIAN, K., LATIF, K., RODRIGUEZ, K., and ALVARADO, C., "Nanokinematics for analysis of protein molecules," *Journal of Mechanical Design*, vol. 127, pp. 699–711, 2005.
- [37] KIM, D. E., CHIVIAN, D., and BAKER, D., "Protein structure prediction and analysis using the Robetta server," *Nucleic Acids Research*, vol. 32, no. Suppl 2:W526–31, 2004.
- [38] KIM, M. K., CHIRIKJIAN, G. S., and JERNIGAN, R. L., "Elastic models of conformational transitions in macromolecules," *Journal of Molecular Graphics* and Modelling, vol. 21, pp. 151–160, 2002.
- [39] KIM, M. K., JERNIGAN, R. L., and CHIRIKJIAN, G. S., "Rigid-cluster models of conformational transitions in macromolecular machines and assemblies," *Biophysical Journal*, vol. 89, no. 1, pp. 43–55, 2005.
- [40] KIRKPATRICK, S., GELATT, C. D., and VECCHI, M. P., "Optimization by simulated annealing," *Science*, vol. 220, no. 4598, pp. 671–680, 1983.
- [41] KOH, S. K., ANANTHASURESH, G. K., and CROKE, C., "A quadratic programming formulation for the design of reduced protein models in continuous sequence space," *Journal of Mechanical Design*, vol. 127, no. 4, pp. 728–735, 2005.
- [42] KOH, S. K., ANANTHASURESH, G. K., and VISHVESHWARA, S., "A deterministic optimization approach to protein sequence design using continuous models," *The International Journal of Robotics Research*, vol. 24, no. 2–3, pp. 109–130, 2005.
- [43] LAU, K. F. and DILL, K. A., "A lattice statistical mechanics model of the conformational and sequence spaces of proteins," *Macromolecules*, vol. 22, pp. 3986– 3997, 1989.
- [44] LEACH, A. R., Molecular Modelling: Principles and Applications. Prentice Hall, 2 ed., 2001.
- [45] LEVITT, M., "Protein folding by restrained energy minimization and molecular dynamics," *Journal of Molecular Biology*, vol. 170, pp. 723–764, 1983.

- [46] LI, Z. and SCHERAGA, H. A., "Monte Carlo-minimization approach to the multiple-minima problem in protein folding," *Proceedings of the National Academy of Sciences of the United States of America*, vol. 84, no. 19, pp. 6611– 6615, 1987.
- [47] MATHIOWETZ, A. M., JAIN, A., KARASAWA, N., and GODDARD, W. A., "Protein simulations using techniques suitable for very large systems: the cell multipole method for nonbond interactions and the Newton-Euler inverse mass operator method for internal coordinate dynamics," *Proteins: Structure, Function, and Genetics*, vol. 20, pp. 227–247, 1994.
- [48] METROPOLIS, N., ROSENBLUTH, A. W., ROSENBLUTH, M. N., TELLER, A. H., and TELLER, E., "Equation of state calculations by fast computing machines," *The Journal of Chemical Physics*, vol. 21, no. 6, pp. 1087–1092, 1953.
- [49] NESTEROV, Y., Introductory Lectures on Convex Optimization: A Basic Course. Applied optimization, Boston: Kluwer Academic Publishers, 2004.
- [50] PHILLIPS, A. T., ROSEN, J. B., and DILL, K. A., "Energy landscape projections of molecular potential functions," in *Optimization in Computational Chemistry and Molecular Biology: Local and Global Approaches* (FLOUDAS, C. A. and PARDALOS, P. M., eds.), pp. 47–55, Dordrecht/Boston/London: Kluwer Academic, 2000.
- [51] PIELA, L., KOSTROWICKI, J., and SCHERAGA, H. A., "On the multiple-minima problem in the conformational analysis of molecules: deformation of the potential energy hypersurface by the diffusion equation method," *The Journal of Physical Chemistry*, vol. 93, no. 8, pp. 3339–3346, 1989.
- [52] PRESS, W. H., FLANNERY, B. P., TEUKOLSKY, S. A., and VETTERLING, W. T., Numerical Recipes: The art of scientific computing. Cambridge, UK; New York, NY: Cambridge University Press, 1986.
- [53] RAPAPORT, D. C., "Molecular dynamics simulation of polymer helix formation using rigid-link methods," *Physical Review E*, vol. 66, no. 1, p. 011906, 2002.
- [54] RAPAPORT, D. C., "Dynamics of polymer chain collapse into compact states," *Physical Review E*, vol. 68, no. 4, p. 041801, 2003.
- [55] RAPAPORT, D. C., The art of molecular dynamics simulation. Cambridge, UK; New York, NY: Cambridge University Press, 2 ed., 2004.
- [56] ROHL, C. A., STRAUSS, C. E. M., MISURA, K. M. S., and BAKER, D., "Protein structure prediction using rosetta," *Methods in Enzymology*, vol. 383, pp. 66–93, 2004.

- [57] SAMUDRALA, R. and LEVITT, M., "Decoys 'R' Us: A database of incorrect conformations to improve protein structure prediction," *Protein Science*, vol. 9, no. 7, pp. 1399–1401, 2000.
- [58] SAPSAMAN, T. and LIPKIN, H., "Improving the efficiency of protein conformation prediction with energy landscape modification," in ASME 2009 International Design Engineering Technical Conferences and Computers and Information in Engineering Conference, (San Diego, California, USA), 2009.
- [59] SCHELSTRAETE, S. and VERSCHELDE, H., "Finding minimum-energy configurations of Lennard-Jones clusters using an effective potential," *The Journal of Physical Chemistry A*, vol. 101, no. 3, pp. 310–315, 1997.
- [60] SCHLICK, T., Molecular Modeling and Simulation. Springer, 2002.
- [61] SHARMA, G., BADESCU, M., DUBEY, A., MAVROIDIS, C., TOMASSONE, S. M., and YARMUSH, M. L., "Kinematics and workspace analysis of protein based nano-actuators," *Journal of Mechanical Design*, vol. 127, pp. 718–727, 2005.
- [62] SIEW, N., ELOFSSON, A., RYCHLEWSKI, L., and FISCHER, D., "MaxSub: an automated measure for the assessment of protein structure prediction quality," *Bioinformatics*, vol. 16, no. 9, pp. 776–785, 2000.
- [63] SIMONS, K. T., KOOPERBERG, C., HUANG, E., and BAKER, D., "Assembly of protein tertiary structures from fragments with similar local sequences using simulated annealing and Bayesian scoring functions," *Journal of Molecular Biology*, vol. 268, no. 1, pp. 209–225, 1997.
- [64] SIMONS, K. T., RUCZINSKI, I., KOOPERBERG, C., FOX, B. A., BYSTROFF, C., and BAKER, D., "Improved recognition of native-like protein structures using a combination of sequence-dependent and sequence-independent features of proteins," *Proteins: Structure, Function, and Genetics*, vol. 34, no. 1, pp. 82– 95, 1999.
- [65] SKOLNICK, J., "In quest of an empirical potential for protein structure prediction," Current Opinion in Structural Biology, vol. 16, no. 2, pp. 166–171, 2006.
- [66] STEPHENSON, D. S. and BINSCH, G., "Automated analysis of high-resolution NMR spectra. I. Principles and computational strategy," *Journal of Magnetic Resonance*, vol. 37, no. 3, pp. 395–407, 1980.
- [67] STRANG, G., Linear Algebra and Its Applications. Brooks/Cole, third ed., 1988.
- [68] SUBRAMANIAN, R. and KAZEROUNIAN, K., "Residue level inverse kinematics of peptide chains in the presence of observation inaccuracies and bond length changes," in ASME 2004 Design Engineering Technical Conferences and Computers and Information in Engineering Conference, (Long Beach, California, USA), 2005.

- [69] SUBRAMANIAN, R. and KAZEROUNIAN, K., "Improved molecular model of a peptide unit for proteins," in ASME 2006 International Design Engineering Technical Conferences and Computers and Information in Engineering Conference, (Philadelphia, Pennsylvania, USA), 2006.
- [70] SWENDSEN, R. H. and WANG, J.-S., "Replica Monte Carlo simulation of spinglasses," *Physical Review Letters*, vol. 57, no. 21, pp. 2607–2609, 1986.
- [71] THORPE, M., LEI, M., RADER, A., JACOBS, D. J., and KUHN, L. A., "Protein flexibility and dynamics using constraint theory," *Journal of Molecular Graphics* and Modelling, vol. 19, pp. 60–69, 2001.
- [72] VAIDEHI, N., JAIN, A., and GODDARD, W. A., "Constant temperature constrained molecular dynamics: The Newton-Euler inverse mass operator method," *J. Phys. Chem.*, vol. 100, no. 25, pp. 10508–10517, 1996.
- [73] WALES, D. J. and SCHERAGA, H. A., "Global optimization of clusters, crystals, and biomolecules," *Science*, vol. 285, pp. 1368–1372, 1999.
- [74] WAWAK, R. J., WIMMER, M. M., and SCHERAGA, H. A., "Application of the diffusion equation method of global optimization to water clusters," *The Journal* of *Physical Chemistry*, vol. 96, no. 12, pp. 5138–5145, 1992.
- [75] WELLS, S., MENOR, S., HESPENHEIDE, B., and THORPE, M. F., "Constrained geometric simulation of diffusive motion in proteins," *Physical Biology*, vol. 2, pp. S127–S136, 2005.
- [76] WROBLEWSKA, L. and SKOLNICK, J., "Can a physics-based, all-atom potential find a protein's native structure among misfolded structures? I. Large scale AMBER benchmarking," *Journal of Computational Chemistry*, vol. 28, no. 12, pp. 2059–2066, 2007.
- [77] YANG, L., TAN, C., HSIEH, M. J., WANG, J., DUAN, Y., CIEPLAK, P., CALD-WELL, J., KOLLMAN, P. A., and LUO, R., "New-generation amber united-atom force field," *The Journal of Physical Chemistry B*, vol. 110, no. 26, pp. 13166–13176, 2006.
- [78] YOON, H.-S., Optimization Approaches to Protein Folding. Ph.d. dissertation, Industrial and Systems Engineering, Georgia Institute of Technology, Atlanta, GA, 2006.
- [79] ZEMLA, A., "LGA: a method for finding 3D similarities in protein structures," Nucleic Acids Research, vol. 31, no. 13, pp. 3370–3374, 2003.
- [80] ZEMLA, A., VENCLOVAS, C., MOULT, J., and FIDELIS, K., "Processing and analysis of CASP3 protein structure predictions," *Proteins: Structure, Function*, and Genetics, vol. 37, no. S3, pp. 22–29, 1999.

- [81] ZHANG, M. and KAVRAKI, L. E., "A new method for fast and accurate derivation of molecular conformations," *Journal of Chemical Information and Computing Sciences*, vol. 42, pp. 64–70, 2002.
- [82] ZHANG, M., WHITE, R. A., WANG, L., GOLDMAN, R., KAVRAKI, L., and HASSETT, B., "Improving conformational searches by geometric screening," *Bioinformatics*, vol. 21, no. 5, pp. 624–630, 2005.
- [83] ZHANG, Y. and SKOLNICK, J., "Automated structure prediction of weakly homologous proteins on a genomic scale," *Proceedings of The National Academy of Sciences of the United States of America*, vol. 101, no. 20, pp. 7594–7599, 2004.
- [84] ZHANG, Y. and SKOLNICK, J., "Tertiary structure predictions on a comprehensive benchmark of medium to large size proteins," *Biophysical Journal*, vol. 87, no. 4, pp. 2647–2655, 2004.
- [85] ZHANG, Y. and SKOLNICK, J., "The protein structure prediction problem could be solved using the current PDB library," *Proceedings of The National Academy* of Sciences of the United States of America, vol. 102, no. 4, pp. 1029–1034, 2005.
- [86] ZHANG, Y. and SKOLNICK, J., "TM-score: Assessment of quality of protein structure templates and full-length model predictions." Skolnick Research Group: Online Resources http://cssb.biology.gatech.edu/skolnick/webservice/ TM-score/index.shtml, November 2009.
- [87] ZHANG, Y., KIHARA, D., and SKOLNICK, J., "Local energy landscape flattening: Parallel hyperbolic Monte Carlo sampling of protein folding," *Proteins: Structure, Function, and Genetics*, vol. 48, pp. 192–201, 2002.
- [88] ZHANG, Y. and SKOLNICK, J., "Scoring function for automated assessment of protein structure template quality," *Proteins: Structure, Function, and Bioinformatics*, vol. 57, no. 4, pp. 702–710, 2004.

VITA

Temsiri Sapsaman was born in Nakhonsawan, a province in the middle of Thailand. She grew up in a teachers' housing at a small rural elementary school where her parents worked and she went to school. After fourth grade, she moved to a bigger town in Nakhonsawan to live with her grandparents and became very close to her grandmother. She went to a high school in Bangkok, Triam Udom Suksa school, and stayed with relatives. There she learned about an opportunity to study in the United States. and in 1996, she came to the US and after spending one year in a private prep school, Westover, in Middlebury, CT, she pursued a Bachelor's degree in Mechanical Engineering at Carnegie Mellon University, Pittsburgh, PA. After graduating in May 2001, she moved to Atlanta, GA to pursue her graduate degrees at the Georgia Institute of Technology. There she developed an interest in teaching as she has had several opportunities to work as a graduate teaching assistant and to supervise undergraduate students on their undergraduate research. Also at Georgia Tech, she served on a committee of Woodruff School Graduate Women, a social group for female graduate students in the Woodruff School of Mechanical Engineering, for two years.